Massachusetts Electric

6 2 3

A National Grid Company

SCANNED

27 June 2003

Massachusetts Department of Environmental Protection Northeast Regional Office One Winter Street Boston, Massachusetts 02108



Attention: Site Management Branch

Subject: Phase III - Remedial Action Plan Former Malden Manufactured Gas Plant (MGP) Site – Upland Portion Malden, Massachusetts RTN 3-0362 and Linked RTNs 3-3757, 3-11581, 3-12448, 3-13310, 3-13345, 3-13753, and 3-13754 Tier IB Permit Number 7378

Dear Ladies and Gentlemen:

Massachusetts Electric Company (MEC) is pleased to submit this report entitled "Phase III – Remedial Action Plan, Former Manufactured Gas Plant (MGP) Site - Upland Portion, Malden, Massachusetts, RTN 3-0362, Tier IB Permit 7378," prepared by Haley & Aldrich, Inc. This report is designed to meet the Phase III requirements under the Massachusetts Contingency Plan (310 CMR 40.0000) for the upland portions of the former Malden MGP Site. The Malden River portion of the Site will be addressed separately, as discussed below.

As MEC discussed in a meeting with Massachusetts Department of Environmental Protection (MADEP) officials on April 3, 2003, MEC intends to address the sediments in the Malden River within the Site boundary as a separate operable unit. MGP-related impacts attributable to the former MGP Site were identified during the Phase II assessment in Malden River sediments from the Malden River culvert outfall to a point approximately 1,400 feet downstream (just north of the Medford Street bridge). The Mystic Valley Development Commission (MVDC), through the TeleCom City partnership, has formed a group of parties with interest in Malden River sediment remediation from the culvert outfall to the Amelia Earhart Dam. The Telecom City partnership formed due to the development of a telecommunications research and development park on 200 acres of land situated in Malden, Medford and Everett. This area is located along the Malden River downstream of the Site boundary. The MVDC has partnered with the Army Corps of Engineers to conduct a study of the nature and extent of sediment impacts and to identify potential remedial measures that may be undertaken in the area. It is also our understanding that MADEP is participating in this work. MEC is contributing technical and financial support to this project and intends to participate in this study, which has been designated the Malden River Ecosystem Restoration Study. MEC also intends to participate in discussions regarding the remediation and restoration of the Malden River. In light of these recent developments, remedial measures related to Malden River sediments associated with the MGP Site are best conducted in

> 55 Bearfoot Road Northborough, MA 01532-1555 508.421.7000

conjunction with the efforts along the larger portion of the Fiver to ensure consistency and coordination. Representatives of MADEP endorsed this approach at our April 3 meeting and during conversations thereafter.

5.

;

An original signed copy of transmittal form BWSC-108 is submitted, unbound, along with this report. A copy of the signed form and this cover letter are provided in Appendix A of this report along with copies of Notification of Availability letters to appropriate Malden municipal officials.

Please contact me at 508-421-7564 with any questions or comments regarding this matter.

Sincerely,

1 ....

Michel V. Leme Bow

Michele V. Leone Senior Environmental Engineer

cc: Gregg Hunt, DEP (without enclosure) Rick Standish, H&A File

	Massachusetts Department of Environmental Protection Bureau of Waste Site Cleanup COMPREHENSIVE RESPONSE ACTION TRANSMITTAL	BWSC-108 A. K. Release Tracking Number
DEP	FORM & F'HASE I COMPLETION STATEMENT Pursuant to 310 CMR 40.0484 (Subpart D) and 40.0800 (Subpart H)	3 - 362
A. SITE LOCATION: Site Name: (optional)	Former Malden MGP Site	
Street: 100 Comm	ercial Street: Location Aid: Commecial & Ch	arles_Streets
City/Town: Malden	ZIP Code: 02148-5510	
Related Release Tracking	ng Numbers that this Form Addresses:	, 3-13753, 3-13754
Tier Classification: (che	eck one of the following) 🗌 Tier IA 📝 Tier IB 🗌 Tier IC 🗌 Tier II	Not Tier Classified
If a Tier I Permit ha	s been issued, state he Permit Number: Permit No. 7378, effective date 12,	/28/1999
B. THIS FORM IS B	EING USED TO: (check all that apply)	
Submit a Phase I	Completion Statem ant, pursuant to 310 CMR 40.0484 (complete Sections A, B, C, G, H, I and J).	
Submit a Phase II	Scope of Work, pursuant to 310 CMR 40.0834 (complete Sections A, B, C, G, H, I and J).	
Submit a final Pha (complete Section:	se II Comprehensive Site Report and Completion Statement, pursuant to 310 CMR 40.0836 s A, B, C, D, G, H, I and J).	
Submit a Phase II	Remedial Action Plan and Completion Statement, pursuant to 310 CMR 40.0862 (complete Section	s A, B, C, G, H, I and J).
Submit a Phase IV	Remedy Implementation Plan, pursuant to 310 CMR 40.0874 (complete Sections A, B, C, G, H	
Submit an As-Bui	It Construction Report, pursuant to 310 CMR 40.0875 (complete Sections A, B, C, G, H, I and J)	CUEIVED
Submit a Phase IV (complete Section	/ Final Inspection Report and Completion Statement, pursuant to 310 CMR 40.0878 and 40.0879 s A, B, C, E, G, H, I ard J).	JUL 0 2 2003
Submit a periodic	Phase V Inspection & Monitoring Report, pursuant to 310 CMR 40.0892 (complete Sections A, B, C,	G, H, I and ()
Submit a final Pha (complete Section	se V Inspection & Monitoring Report and Completion Statement, pursuant to 310 CMB 40,0893 s A, B, C, F, G, H, I and J).	AST REGIONAL OFFI
	any Legil Notices and Notices to Public Officials required by 310 CMR 40.1400.	sopies di
C. RESPONSE ACT	IONS:	
Check here if any interested in using	response action(s) that serves as the basis for the Phase submittal(s) involves the use of Innovative Tech this information to create an Innovative Technologies Clearinghouse.)	nologies. (DEP is
Describe Technoli	ogies:	
D. PHASE II COMPL	LETION STATEMENT:	
Specify the outcome of	the Phase II Comprehensive Site Assessment:	
Additional Compre	thensive Response Actions are necessary at this Site, based on the results of the Phase II Comprehensiv	ve Site Assessment.
The requirements be submitted to D	of a Class A Response Action Outcome have been met and a completed Response Action Outcome Sta EP.	stement (BWSC-104) will
The requirements be submitted to D	of a Class B Response Action Outcome have been met and a completed Response Action Outcome Sta EP.	tement (BWSC-104) will
Rescuring of this ?	Site using the Numeric a Ranking System is necessary, based on the results of the final Phase II Report.	
E. PHASE IV COMP	PLETION STATEMIENT:	
Phase V operation	Phase in activities: n, maintenance or monitoring of the Comprehensive Resconse Action is necessary to achieve a Response	se Action Outcome.
(This site will be s	ubject to a Phase V Operation, Maintenance and Monitoring Annual Compliance Fee.)	
The requirements ensure the integri	of a Class A Response Action Outcome have been met. No additional operation, maintenance or monito ty of the Response Action Outcome. A completed Response Action Outcome Statement (BWSC-104) with	pring is necessary to Il be submitted to DEP.
The requirements ensure the integri	of a Class C Respons ≱ Action Outcome have been met. No additional operation, maintenance or monitor by of the Response Action Outcome. A completed Response Action Outcome Statement (BWSC-104) with SECTION E IS CONTINUED ON THE NEXT PAGE	oring is necessary to II be submitted to DEP.
Revised 3/30/95	Supersedes Forms BWSC-010 (in part) and 013 Do Not Alter This Form	Page 1 of 3

2. . 1

	lassachusetts Jureau of Waste	Department of Site Cleanup	Environm	ental Protection	BWSC-1
	OMPREHENS			TRANSMITTAL	Release Tracking Numb
DEP	ursuant to 310 CMR	40.0484 (Subpart D) a	and 40.0800 (S	ubpart H)	3 - 362
PHASE IV COMPLETI	ON STATEMENT:	(continued)			
The requirements of a C necessary to ensure that Outcome Statement (BV	lass C Response Action t conditions are maintaine VSC-104) will be submitt	Outcome have been met. ed and that further progre ed to DEP.	Further operations is made toward	on, maintenance or monitoring d a Permanent Solution. A cor	of the remedial action is mpleted Response Action
Indicate whether the ope	tration and maintenance	will be Active or Passive.	(Active Operatio	n and Maintenance is defined	at 310 CMR 40.0006.):
Active Operation a	nd Maintenancu		O Passive	Operation and Maintenance	
(Asthus Opportunities and M	laistenance makes the C	its subject to a Deel PAC	Class C Asthus (	Operation and Maintenance An	musi Compliance Eng \
PLACE V COMPLETI	A CTATEMENT.	ite subject to a Posi-role	Ciass C Active (	operation and Maintenance An	idal compliance rec.)
Social the outcome of Phone	Viostiulias:				
Specify the outcome of Phase	s v acuviuss.	22 2 2			
The requirements of a C be submitted to DEP.	lass A Response Action	Outcome have been met	and a completed	Response Action Outcome St	atement (BWSC-104) will
The requirements of a C the integrity of the Resp	lass C Response Action onse Action Outcome. A	Outcome have been met completed Response Ac	No additional o tion Outcome Sta	peration, maintenance or monit atement (BWSC-104) will be su	loring is necessary to ensure ubmitted to DEP.
The requirements of a C necessary to ensure tha Outcome Statement (B)	lass C Response Action t conditions are maintain NSC-104) will be submit	Outcome have been met ed and that further progre ted to DEP.	Further operations is made towar	on, maintenance or monitoring d a Permanent Solution. A con	of the remedial action is mpleted Response Action
Indicate whether the ope	eration and mair tenance	will be Active or Passive.	(Active Operatio	on and Maintenance is defined	at 310 CMR 40.0006.):
Active Operation a	and Maintenance		O Passive	Operation and Maintenance	
(Active Operation and )	taintanance makes the S	tio subject to a Post PA	Class C Active	Operation and Maintenance An	nual Compliance Fee )
(Active Operation and w	taimenance makes ure o	site subject to a Post-ton	J Glass C Active	operation and Maintenance A	indai compilance r co.y
nformation and belief, if Section B indicates that that is (are) the subject of this 310 CMR 40.0000, (ii) is (are M.G.L. c. 21E and 310 CMR if Section B indicates that (are) the subject of this subm (are) appropriate and reason CMR 40.0000, and (iii) comp if Section B indicates that action(s) that is (are) the sub 40.0000, (ii) is (are) appropri 21E and 310 CMR 40.0000,	a Phase I, Phase II, Ph. submittal (i) has (have) t ) appropriate and reason 40.0000, and (iii) complet a Phase II Scope of Wo ittal (i) has (have) been d able to accomplish the p lies(y) with the identified an As-Built Constructi ject of this submittal (i) is ate and reasonable to ar and (iii) complies (y) with	ase III, Phase IV or Pha been developed and impli- table to accomplish the p es(y) with the identified pro- pork or a Phase IV Remei- teveloped in accordance of urposes of such responsi- provisions of all orders, p for Report or a Phase V (are) being implemented complish the purposes of the identified provisions	se V Completion emented in accor- urposes of such r rovisions of all or dy Implementati with the applicable e action(s) as set eermits, and appro- Inspection and in accordance w of such response of all orders, per	n Statement is being submitter dance with the applicable provi response action(s) as set forth ders, permits, and approvals ide ion Plan is being submitted, the provisions of M.G.L. c. 21E a forth in the applicable provision ovals identified in this submittal Monitoring Report is being so ith the applicable provisions of action(s) as set forth in the app nits, and approvals identified in	ed, the response action(s) isions of M.G.L. c. 21E and in the applicable provisions of entified in this submittal; the response action(s) that is and 310 CMR 40.0000, (ii) is ns of M.G.L. c. 21E and 310 ; ubmitted, the response M.G.L. c. 21E and 310 CMR licable provisions of M.G.L. of this submittal.
I am aware that significant pe false, inaccurate or materially	nalties may result, includ v incomplete.	ling, but not limited to, pos	ssible fines and in	nprisonment, if I submit informa	tion which I know to be
Check here if the Response	is checked, you MUST a	inis opinion is based, if an attach a statement identify	iy, are (were) sub ing the applicable	e provisions where of the OF March	poror approval(s) issued by
DEP OF EPA. It the box	. Standish	LSP #: 2242	Stamp:	3	191E
LSP Name: <u>Richard</u> F				ALS RICHARD	SIL
LSP Name: <u>Richard</u> F	400	Ext.: 3150			15 P
LSP Name: <u>Richard F</u> Telephone: <u>860-282-9</u> FAX: (optional) <u>860-28</u>	2-950	Ext.: <u>3150</u>		STANDISH No. 2242	)=]K
LSP Name: <u>Richard</u> F Telephone: <u>860-282-9</u> FAX: (optional) <u>860-28</u>	2-9501_A-	Ext.: <u>3150</u>	-	STANDISH No. 2242	)]
LSP Name: <u>Richard</u> F Telephone: <u>860-282-9</u> FAX: (optional) <u>860-28</u> Signature:	2-95 Ca Jul	Ext.: 3150	-	STANDISH No. 2242	
LSP Name: <u>Richard</u> F Telephone: <u>860-282-9</u> FAX: (optional) <u>860-28</u> Signature: <u><u>Richard</u> F</u>	2-950 timb	Ext.: 3150		STANDISH No. 2242 PEGISTERES SITE PROFE	

	Massachusetts Bureau of Waste	Department of E	nviration	tal Protection	BWSC-108
D E P	COMPREHENS FORM & PHAS Pursuant to 310 CMR	IVE RESPONSE E I COMPLETION 40.0484 (Subpart D) and	ACTION TRA STATEMEN and 40.0800 (Subp	ANSMITTAL NT part H)	Release Tracking Number           3         -         362
H. PERSON UNDER	TAKING RESPONSE A	CTION(S):			
Name of Organization:	Massachusetts El	ectric Company			
Name of Contact: Mic	chele V. Leone		Title: Seni	or Environment	al Engineer
Street: 55 Bearfo	ot_Road		<u></u>		
City/Town: Northb	orough		State: MA	ZIP Code: _0.3	532-0000
Telephone: 508-42	1-7564	Ext.:	FAX: (optional	) 508-890-4706	0.001
Check here if there	e has been a change in the p	person undertaking the Resp	oonse Action.		
I. RELATIONSHIP T	O SITE OF PERSON UN	DERTAKING RESPON	SE ACTION(S):	(check one)	
RP or PRP Spe	cify: () Owner () Op	perator () Generator (	) Transporter Oti	her RP or PRP: Part	y of Interest
Fiduciary, Secured	Lender or Municipality with	Exempt Status (as defined	by M.G.L. c. 21E. s	(2)	
Agency of Public I	Utility on a Right of Way (as	defined by M.G.L. c. 21F. s	5(ii)	-	
Any Other Person	Undertaking Response Activ	nn Specify Relationshin			
		AKING RESPONSE AC			
of those individuals imm krowledge and belief, to this submittal. I/the per possible fines and impr By:	nediately responsible 'or obta rue, accurate and complete, son or entity on whose beha isonment, for willfully submit	alning the information, the m and (iii) that I am fully autho If this submittal is made am ting false, inaccurate, or inc	aterial information c rized to make this at /is aware that there omplete information Title: Seni	contained in this submitta ttestation on behalf of th are significant penalties, or Environment	a is, to the best of my e entity legally responsible for including, but not limited to,
(signature)			N	125103	
For: Massachuset (print name of pers	tts Electric Comp son or entity recorded in Sec	pany tion H)	Date;0	010010-2	tu or
Enter address of the pe	erson providing certification, i	if different from address rec	orded in Section H:		
City/Town:			State:	ZIP Code:	
Telephone:		Ext.:	FAX: (optiona	0	
YOU MUST ( INCO	COMPLETE ALL RELEV MPLETE, IF YOU SUBI	VANT SECTIONS OF T MIT AN INCOMPLETE I A REQUIRED	HIS FORM OR D FORM, YOU MAY DEADLINE.	EP MAY RETURN T	THE DOCUMENT AS
Revised 3/30/95	St	upersedes Forms BWS0	C-010 (in part) an	d 013	Page 3 of 3

.

# REPORT ON PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE – UPLAND PORTION MALDEN, MASSACHUSETTS RTN 3-0362 TIER IB PERMIT 7378

1

VOLUME I OF III

by

Haley & Aldrich, Inc. Boston, Massachusetts

RECEIVED Commonwealth of Massachusetts JUL 0 2 2003

> DEP/BOSTON 2nd floor Reception

1

for

Massachusetts Electric Company Northborough, Massachusetts

File No. 06558-634 June 2003



TABLE OF CONTENTS

# VOLUME I OF III

LIST	OF TA	BLES	v
LIST	OF FIG	URES	vii
1.	INTR	ODUCTION	1
	1.01	Site Background Information	1
	1.02	Site Description and Location	1
	1.03	Phase III Purpose and RAP Overview	3
	1.04	Summary of Disposal Site History and Regulatory Status	4
		A. MGP Processes Used at the Site	4
		B. Land Use and History	4
		C. Historic Re-routing of Surface Water Bodies on the Site	5
		D. Current Regulatory Status	6
п.	DISP	OSAL SITE CONDITIONS	8
	2.01	Summary of Site Geology and Hydrogeology	8
		A. Subsurface Stratigraphy	8
		B. Hydrogeology	9
	2.02	Site Physical Conditions and Current Use	10
		A. Structures and Current Use	11
		B. Utilities	11
		C. Below-grade Structures	12
	2.03	New Disposal Site Information	12
		A. Results of Soil Sampling Conducted at 51 Commercial Street	13
		B. Results of Supplemental Investigations at 129 Commercial Street	13
	2.04	Nature and Extent of Contamination: Conceptual Site Model Summary	14
		A. TSM	15
		B. Shallow DNAPL	15
		C. Deep DNAPL	16
		D. LNAPL	17
		E. BTEXSN in Soil and Groundwater	18
		F. Petroleum-impacted Soil	19
	2.05	Risk Characterization Summary	20
		A. General	20
		B. Human Health Risk Characterization	21
		C. Risk to Public Welfare	24
		D. Environmental Risk	24
		E. Substantial Hazard Evaluation	24
		F. Conclusions Based on Risk Characterization Results	26
	2.06	Remedial Action Alternative Areas	26
		A. Area 1, Northern Portion of Parcel E	28

1

1

.

۰.



1

I

0

i

Page

TABLE OF CONTENTS

(Continued)

1

Í

Í

Í

Ĩ

Í

-

		В.	Area 2, Southern Po	ortion of Parcel E	30
		C.	Area 3, Northern Po	ortion of Parcel A	32
		D.	Area 4, Southern Po	ortion of Parcel A	33
		Ε.	Area 5, Parcel B		35
ш.	REM	EDIAL	<b>OBJECTIVES</b>		38
				1	
	3.01	Phase	e III Approach		38
	3.02	Rem	edial Objectives to Ach	ieve a Permanent Solution	39
		Α.	Elimination of Poter	ntial Exposure to Contaminated Soil	39
		В.	Elimination of Poter	ntial Exposure to Indoor Air	40
		C.	Elimination of soil 1	UCL Exceedences.	40
		D.	Elimination of the p	resence of NAPL	41
		E.	Groundwater Qualit	y	41
	3.03	Rem	edial Objectives for a T	emporary Solution	41
	3.04	Rem	edial Objectives for Are	ea l	42
	3.05	Rem	edial Objectives for Are	ea 2	42
	3.06	Rem	edial Objectives for Are	ea 3	42
	3.07	Rem	edial Objectives for Are	ea 4	43
	3.08	Rem	edial Objectives for Are	ea 5 ,	43
IV.	REM	EDIAI	ACTION ALTERNA	TIVE EVALUATION PROCESS	45
	1 01	Initia	Screening Process		45
		A	Initial Screening		45
		B	Elimination of Rem	edial Technologies Based on Site Char	acteristics45
	4 02	Deve	lopment of Remedial A	Action Alternatives	46
	4.02	Deta	iled Evaluation Process	celon Anernatives	46
	4.05	Δ	Evaluation Criteria		46
		B.	Weighting of the Ex	valuation Criteria	40
	4.04	Selec	ted Remedial Action A	Iternatives for the Site	48
270					
v.	EVAI	LUATI	ON OF REMEDIAL	ALTERNATIVES FOR AREA 1	49
	5.01	Intro	duction	2	49
	5.02	Sum	mary of Area I Conditi	ions	49
	5.03	Sum	mary of Area 1 Remed	ial Objectives	49
	5.04	Initia	al Screening of Remedi	al Technologies for Area 1	50
		Α.	Shallow DNAPL		50
		B.	Soil	-	51
		C.	Elimination of Rem	edial Technologies Based on Site Cha	racteristics 52
	5.05	Dev	elopment of Remedial	Action Alternatives for Area 1	52
	5.06	Deta	iled Evaluation of Rem	edial Action Alternatives for Area 1	56
	5.07	Sele	cted Remedial Action A	Iternative for Area 1	61
	5.07	Sele	ette Remoulai Action /	memative for Area 1	01

÷

....

00

1

.

MANDA

ii

Page

# TABLE OF CONTENTS

# (Continued)

1

1

		B. Remedial Action Objectives	89
	9.03	Identification, Initial Screening, and Development of Remedial Action	80
		Alternatives	89
		A. Identification and Screening of Technologies	89
	0.04	B. Development of Remedial Alternatives	90
	9.04	Selection of a Demodial Action Alternatives	91
	9.05	Selection of a Remedial Action Alternative	92
		A. Comparison of Alternatives	92
		B. Feasibility of Implementing a Permanent Solution	92
		C. Feasibility of Achieving of Approaching Background	93
		E. Selection of Alternatives	93 93
x.	SUMN	MARY OF SELECTED REMEDIAL ACTION ALTERNATIVES	95
	10.01	Phase III RAP Overview	95
	10.02	Feasibility of Achieving a Permanent Solution or Background Concentrati	ons96
	10.03	Selected Remedial Action Alternative	98
	10.05	A Selected Remedial Solution Components	98
		B. Definitive and Enterprising Steps	99
		C. Anticipated Remedial Implementation Approach	100
		D. Estimated Net Present Value of the Selected Remedial Alternative	: 101
	10.04	Justification of Temporary Solution	101
REF	ERENCI	ES	102
TAB	LES		
FIGI	JRES		
APP	ENDIX A	A - Copy of Transmittal Form BWSC-108 and Public Notification of Availability Letters	
<b>APP</b>	ENDIX I	B – Revised AMEC Risk Characterization	
VOL	UME II	OF III	
APP	ENDIX (	C – Boring Logs, Monitoring Well Installation Reports, and Soil, Groundw and Indoor Air Data Sheets	ater
VOL	UME III	OF III	
APP	ENDIX I ENDIX I	<ul> <li>D - Cost Estimate Tables and Supporting Calculations</li> <li>E - Brown and Caldwell Focused Area 5 Feasibility Study</li> </ul>	



# LIST OF TABLES

I

I

-

	2	Page
Table No.	Title	
I	Site Structures and Current Property Use	11
п	Summary of Supplemental Soil Sampling Results, Commercial Street	
ш	Summary of Supplemental Soil Sampling Results, Commercial Street	
IV	Summary of Supplemental Groundwater Sampling Results, 129 Commercial Street	
v	Summary of Supplemental Indoor Air Sampling Results Commercial Street	i.
VI	Summary of Properties Included in Risk Characterization	21
VII	Summary of Human Health Risk Characterization, Futu Exposure Pathways	ire 23
VIII	Summary of Public Welfare Risk Characterization Results	24
IX	Summary of Remedial Action Alternative Areas and Associated Impacted Media	27
х	Summary of Soil Exposure Pathways Requiring Remediation	39
XI	Initial Screening of Remedial Technologies, Area 1	
ХШ	Remedial Technologies Eliminated Based on Site Characteristics, Area 1	51
ХШ	Detailed Evaluation of Remedial Action Alternatives, Area 1	
XIV	Initial Screening of Remedial Technologies, Area 2	
XV	Remedial Technologies Eliminated Based on Site Characteristics, Area 2	66

1

•

÷ F



# LIST OF TABLES (Continued)

ł

I

XVI	Detailed Evaluation of Remedial Action Alternatives, Area 2	
XVII	Initial Screening of Remedial Technologies, Area 3	
XIII	Remedial Technologies Eliminated Based on Site Characteristics, Area 3	75
XIX	Detailed Evaluation of Remedial Action Alternatives, Area 3	
XX	Initial Screening of Remedial Technologies, Area 4	
XXI	Remedial Technologies Eliminated Based on Site Characteristics, Area 4	83
XXII	Detailed Evaluation of Remedial Action Alternatives, Area 4	
XXIII	Summary of Remedial Action Alternative Areas and Associated Impacted Media	95
XXIV	Summary of Estimated Net Present Value for the Selected Remedial Alternative	

4

:



1

Page

LIST OF FIGURES

I

•

Figure No.	Title
1	Project Locus
2	Parcel Designations
3	Disposal Site Boundary
4	MADEP Site Natural Resources Map
5	Historic MGP Operational Features
6	Subsurface Profiles (Figures 6A through 6G)
7	Top of Organic Deposit Contour Plan
8	Subsurface Exploration Plan
9	Shallow Groundwater Elevation Contours, December 2000
10	Deep Groundwater Elevations, August 2001
11	Phase III Supplemental Soil Sample Locations, 51 Commercial Street
12	Phase III Supplemental Investigation, 129 Commercial Street
13	Nature and Extent of Observed Contamination
14	Maximum BTEXSN Concentrations Observed in Groundwater, 129 Commercial Street
15	Remedial Action Alternative Areas
16	Locations of Proposed Remedial Components, Area 1
17	Locations of Proposed Remedial Components, Area 2
18	Locations of Proposed Remedial Components, Area 3
19	Locations of Proposed Remedial Components, Area 4
20	Locations of Selected Remedial Alternative Components

ŧ.

-

à.

63



4

vii

# I. INTRODUCTION

This Remedial Action Plan (RAP) presents results of the Phase III – Identification, Evaluation and Selection of Comprehensive Remedial Action Alternatives (Phase III) for the upland portion of the Former Malden Manufactured Gas Plant (MGP) Site (the Site), in Malden, Massachusetts. The upland portion of the Site includes the terrestrial land parcels of the Site and excludes the Malden River downstream of the Malden River Culvert outfall to a point approximately 1,400 feet downstream (just north of the Medford Street Bridge). The Malden River portion of the Site will be addressed separately, as described below in Section 1.02. A Site Locus is presented on Figure 1. This report was prepared by Haley & Aldrich, Inc. (Haley & Aldrich), on behalf of Massachusetts Electric Company (MEC), in accordance with 310 CMR 40.0850.

# 1.01 Site Background Information

Portions of the former Malden MGP Site have been the subject of Massachusetts Contingency Plan (MCP) 310 CMR 40.0000 compliance activities since 1988 under various Massachusetts Department of Environmental Protection (MADEP) Release Tracking Numbers (RTNs). This Phase III report presents a Remedial Action Plan for those portions of the former Malden MGP which have not yet received a Response Action Outcome (RAO) or a Waiver Completion Statement, and are not subject to MCP investigations by others. The Phase II – Comprehensive Site Assessment (Phase II) for the Site identified Significant Risks to human health, public welfare and the environment. This Phase III Feasibility Study was completed to evaluate Remedial Action Alternatives to address the risks.

The former Malden and Melrose Gas Light Company (MMGLC) and its successor, the Mystic Valley Gas Company (MVGC), operated an MGP on approximately 16.4 acres of land in the vicinity of the intersection of Commercial and Charles Streets in Malden, Massachusetts from approximately the mid to late 1800s to the late 1960s/early 1970s. The former holdings occupied land currently referred to as Parcels A, B, C, D and E, as shown on Figure 2. Each of these properties was redeveloped following the decommissioning of the former MGP facilities in the 1970s, and is now owned and controlled by various parties. MEC does not own any of the Site parcels. As indicated above, this Phase III report addresses portions of the former MMGLC/MVGC holdings that have not been addressed by other parties, or that have not had a Waiver Completion Statement or RAO filed at MADEP. Therefore, Parcel C (RTN 3-2066), which has a Waiver Completion Statement dated 14 September 1990, and the portion of Parcel D (i.e., Callahan Park) subject to an existing Partial RAO (RTN -13310), are not included in this Phase III assessment.

## 1.02 Site Description and Location

The Former Malden MGP Site boundary and parcels of the MGP that are included in this Phase III are identified on Figure 3 and are described below:

Parcel A: Parcel A is a rectangular shape parcel of approximately 2.8 acres, occupied by six buildings (51 through 109 Commercial Street) on five separate properties. The



parcel is bounded by Commercial Street to the east, Charles Street to the south, a Massachusetts Bay Transit Authority (MBTA) Orange Line railroad right-of-way to the west, and Centre Street to the north. Current occupants of the buildings on Parcel A include: a dental office, a chiropractic office, a muffler shop, a retail liquor store, an automobile body shop, a rental car garage and office, a tanning/nail salon, and medical offices.

- Parcel B: Parcel B is a rectangular shape parcel of approximately 2 acres, bounded by Charles Street to the north, Commercial Street to the east, Adams Street to the south, and the MBTA Orange Line railroad right-of-way to the west. The single existing building on the parcel is located at 129 Commercial Street and is currently occupied by a commercial bakery.
- Parcel D: The portion of Parcel D that was not included in the Partial RAO (RTN-13310) includes an area associated with the former Governor House. The Governor House was an historic MGP facility that housed equipment used to regulate the flow of manufactured gas from gas holders on Parcel D into the gas distribution system located in the Charles Street right-of-way. Contamination associated with the former Governor House has been detected in soil in the eastern corner of the southern parking lot and in groundwater in monitoring wells located in the Charles Street right of way. This area includes approximately 3,670 square feet (sf), or 0.08 acres, of the southern Callahan Park parking lot, and a 267-ft length of Charles Street, covering an area of approximately 13,970 sf, or 0.32 acres. The Phase II Risk Characterization demonstrated a condition of No Significant Risk for this area. Furthermore, impacted soil in the Governor House area is not considered to be an ongoing source of contamination. Therefore, remedial alternatives are not developed or evaluated for the Governor House contamination in this Phase III RAP. The Governor House portion of the Site will be incorporated into the existing Grant of Environmental Restriction for Callahan Park, which specifies use restrictions (i.e., property use is maintained as a public park) and provides procedures and limitations for excavation of impacted soil. A Class B-1 RAO is anticipated for the Governor House portion of the Site.
- Parcel E: Parcel E is approximately 6.6 acres in size located at 100 Commercial Street, and is bounded by Commercial Street to the west, Charles Street to the south, Centre Street to the north, and the culverted Malden River to the east. The parcel is currently owned and occupied by the Boston Gas Company d/b/a KeySpan Energy Delivery New England (KeySpan) (formerly Boston Gas Company [BGC]), which actively uses the property as an Operations and Vehicle Maintenance Center.
- Malden River Sediments: Sediments in the reach of the Malden River between the Malden River Culvert outfall and a point approximately 1,400 ft downstream of the culvert outfall are also part of the Site, as outlined in the Phase II and is shown on Figure 3. The Malden River sediment portion of the Site will be excluded from this Phase III RAP for several reasons. As discussed with MADEP during a meeting held on 3 April 2003, MEC intends to address the sediments in the Malden River as a separate operable unit (OU). Additionally, the Mystic Valley Development



Commission (MVDC), through the TeleCom City partnership, has formed a group of parties with interest in Malden River sediment remediation. The Telecom City partnership formed due to the development of a state-of-the-art telecommunications research and development park on 200 acres of land situated in Malden, Medford and Everett. This area is located along the Malden River downstream of the Site boundary. The MVDC has partnered with the Army Corps of Engineers to conduct a study of the nature and extent of sediment impacts and to identify potential remedial measures that may be undertaken in the area. MEC is contributing technical and financial support to this project and intends to participate in this study, which has been designated the Malden River Ecosystem Restoration Study. MEC also intends to participate in discussions regarding the remediation and restoration of the Malden River. In light of these recent developments, remedial measures related to Malden River sediments associated with the former Malden MGP Site would be more efficient and cost-effective if conducted in conjunction with these efforts along the larger portion of the River.

The boundary of the Site was delineated in the Phase II Report, based on the areas where MGP residuals have come to be located and have not been addressed by other parties, or that have not had a Waiver Completion Statement or RAO filed at MADEP. The lateral limits of the disposal Site incorporate property lines where appropriate, and the extents of identified MGP-residuals where they extend beyond property lines. The boundaries of the former Malden MGP disposal Site are shown on Figure 3.

The Site is located within a designated Industrial Zone, and there are no institutions located within 500 ft. There are numerous residences within 0.5 miles of the Site and it is estimated that greater than 1,000 people live within 0.5 miles of the Site. The Site is not located within 3,000 ft of an Area of Critical Environmental Concern. Based on area groundwater use and recharge characteristics, the Site is not included within areas designated Zone I, Zone II, or Zone III. A MADEP Natural Resources Map for the Site is provided on Figure 4.

## 1.03 Phase III Purpose and RAP Overview

This Phase III evaluation assesses the feasibility of implementing various Remedial Action Alternatives for detected contamination at the Site. It considers those alternatives suited to the Site contaminants, affected media, and physical characteristics of the detected contaminants. The assessment is designed to select a Remedial Action Alternative that is a likely Permanent Solution, or to demonstrate that if a Permanent Solution is not feasible at this time, then implementation of a Temporary Solution is a more timely and cost-effective approach.

This RAP provides supporting information on the identification, evaluation, and selection of Remedial Action Alternatives for the Site. In order to develop and compare remedial alternatives for the Site, the Site is divided into Remedial Action Alternative Areas, based on property boundaries and types of contamination identified. For each Area, this Phase III provides a summary of Site conditions, remedial objectives, and presents results of an initial screening of remedial technologies, which is designed to select the most feasible technologies for further evaluation. Remedial alternatives, consisting of remedial technologies that were



retained in the initial screening, are then developed. Remedial alternatives are evaluated in the detailed evaluation for each Area, and Remedial Action Alternatives are selected. Selected remedial alternatives are presented in Section X, and a schedule for Phase IV of the project – Implementation of the Selected Remedial Action Alternative (Phase IV), is presented in Section XI.

## 1.04 Summary of Disposal Site History and Regulatory Status

Contamination present at the Site is the result of over 100 years of MGP operations, and has impacted soil, groundwater, indoor air, sediments and surface water to varying degrees. Currently, a Release Abatement Measure (RAM) is ongoing to address the presence of benzene, toluene, ethylbenzene, xylene (BTEX), styrene and naphthalene contaminants in indoor air at the 129 Commercial Street property. An Immediate Response Action (IRA) is ongoing to address the migration of dense, non-aqueous phase liquid (DNAPL) phase coal tar into subsurface culverts that cross the Site. The Phase II for the Site was submitted to MADEP on 28 December 2001.

## A. MGP Processes Used at the Site

The three known methods of manufactured gas production are coal gasification (CG), carbureted water gasification (CWG), and oil gasification (OG). Although available information indicates that the former Malden MGP used all three production methods over the duration of its operational history, the primary method of gas production was the CWG method, in which gas was manufactured from coal, steam and oil. Gas produced using this method involved production of raw gas by forcing steam through a bed of hot, coked coal. This raw gas was enriched in a carburetor, where nozzles injected oil onto the hot bricks of the carburetor, causing the oil to volatilize and further enrich the gas. The enriched gas was then sent through the condensing process, during which tars and oils were removed from the gas through several cooling steps. Purification was the final polishing step in the gas production process, in which gaseous impurities such as volatile organic compounds (VOCs), hydrogen cyanide, and hydrogen sulfide were removed from the gas in purification boxes. The product gas was then stored in large holders until it was sent into a system of distribution pipes to be distributed to customers for use. The process of manufactured gas production resulted in the generation of several waste products, such as tars, oils, lime wastes, spent iron oxides and cyanide.

# B. Land Use and History

Earliest available information indicates that the Malden and Melrose Gas Light Company (MMGLC) erected a gas manufacturing facility in 1855. The facility, which consisted of coal storage buildings, retort houses, a gas manufacturing building, a condenser house and limited purification facilities on Parcel A, reportedly began providing street lighting by gas on 1 November 1855. The locations of historic MGP facilities are shown on Figure 5. As the plant capacity was expanded, operations spread to Parcels B, D, and E. Parcel B was used primarily for gas purification operations, Parcel D was primarily used for storage and distribution of gas product,



and Parcel E was the location of the second condenser house, a series of aboveground storage tanks, and various tar handling facilities. By 1920, a tar refinery, the American Tar Company, was built on the northern portion of Parcel A.

The MGP continued to operate as MMGLC through 1953. At that time, the MMGLC reorganized and became the Mystic Valley Gas Company (MVGC). Manufactured gas production continued through the early 1960s, at which time natural gas became available and quickly became the primary gas source for MVGC. By 1963, the gas manufacturing building, the retort house, and the tar storage tanks associated with the American Tar Company on Parcel A were removed. Although pipeline natural gas had become the predominant form of gas supplied to customers by the 1960s, limited production of manufactured gas continued to supplement natural gas supplies during peak demand periods until the early 1970s. In the mid-1970s, remaining MGP affiliated structures were removed from Parcel A, and the Malden Redevelopment Authority (MRA) subdivided the parcel into four properties: 51 Commercial Street, 77 Commercial Street, 99-103 Commercial Street, and 105-109 Commercial Street. Development of Parcel A took place between 1973 and 1980. Gas purification operations on Parcel B ceased in the 1960s, and in the mid to late 1970s, previously existing structures were demolished and a two-story cinder block building, 129 Commercial Street, was constructed. Gas storage features remained on Parcel D until approximately 1975, when the gas holders, governor house and support facilities were dismantled. A partial RAO was submitted to MADEP on 10 January 1997 for the majority of Parcel D, and currently is the location of Callahan Park. The remaining portion of Parcel D (i.e., the governor house area) is addressed in this report. Parcel E remained as the primary operational property for MVGC. In the early 1960s gas manufacturing facilities on Parcel E were replaced by an office and storage building, which was constructed on Parcel E east of the control building. In the early 1970s, several structures associated with the gasification process were still present on the parcel. In the mid-1970s (approximately 1973-1975) remaining MGPaffiliated structures were razed except for the control building and the office and storage building. Following demolition of the former MGP structures in the mid-1970s, the office and storage building was expanded and extended north. This building is now known as the operations building. A maintenance garage was also constructed at this time. KeySpan Energy Delivery New England purchased the Boston Gas Company in 2000, and took ownership of Parcel E.

# C. Historic Re-routing of Surface Water Bodies on the Site

The former Malden MGP was located in a marshy area, partially underlain by an organic peat deposit, and was transected by two meandering water bodies. The historic courses of the two surface water bodies on the Site, the Malden River and the West End Brook, are shown on Figure 5. Historically, the Malden River meandered through the Site, flowing generally from the north to south side of the Site, along the eastern Site boundary. The West End Brook, a tributary to the Malden River, flowed across the center of the Site from west to east before emptying into the Malden River. The West End Brook was straightened sometime in the mid 1900s, and in approximately 1970 – 1971, the Metropolitan District Commission (MDC)

1



constructed a culvert to convey the West End Brook across the Parcel E portion of the Site. In 1977, the Malden River culvert was constructed. Both culverts are supported on wooden piles driven through the organic deposit to provide structural support, and are underlain by a layer of crushed stone, approximately 3 ft thick.

# D. Current Regulatory Status

Investigations began at the Site in 1988, when the presence of coal tar contamination was first confirmed through subsurface borings. The Site was initially designated a Tier II Site under the MCP and assigned Release Tracking Number (RTN) 3-0362. In 1997, as part of a Tier II extension submittal, a series of RTNs, assigned due to releases or possible releases of contaminants associated with the former Malden MGP facility, were linked with RTN 3-0362. These include the following: RTN 3-3757 (Parcel B, 129 Commercial Street), RTN 3-13310 (portion of Parcel D (Callahan Park) not subject to RAO), and RTN 3-13345 (Charles Street Immediate Response Action Completion, linked with RTN 3-13310). Therefore, RTN 3-0362 is being used to manage MCP compliance activities for the former holdings of the Malden MGP facility that have not been addressed by other parties, or that have not had a Waiver Completion Statement or RAO filed at MADEP. Annual Tier II Extension Submittals were made from 1996 to 1999. In August 1999, a Tier Re-Classification/Tier IA Permit Application was submitted to MADEP, and Tier IB Permit 7378, with an effective date of 28 December 1999, was subsequently issued by MADEP. A Phase II Comprehensive Site Assessment was submitted to MADEP on 28 December 2001. Notice of delay letters for the Phase III Report were submitted to MADEP on 27 December 2001, 31 October 2002 and 27 February 2003, indicating that a Phase III Report for the Site would be submitted on or before 1 November 2002, 28 February 2003 and 27 June 2003 respectively. These extensions were requested to allow MEC more time to evaluate remedial options for this complicated Site and to discuss the recommended remedial action alternatives with City officials and impacted property owners.

Currently, two response actions are ongoing at the Site. One is a RAM, which was initiated in November 1998 to address the elevated concentrations of benzene, toluene, ethylbenzene, xylenes, styrene and naphthalene contaminants (BTEXSN) detected in indoor air at 129 Commercial Street. The RAM has involved continued indoor air testing, evaluation of options to mitigate the migration of VOCs from beneath the building floor slab to indoor air, pilot testing of floor sealing activities and the installation of a pilot scale, sub-slab ventilation system at 129 Commercial Street. Recent updates to the risk characterization for 129 Commercial Street incorporating the 2002 and 2003 indoor air data indicate a condition of No Significant Risk associated with indoor air contaminants under current use.

Another ongoing response action is an IRA, which was initiated in May 1996 in response to observations of an intermittent sheen on the surface of water flowing in the MR Culvert, which borders the Site. Specifically, the IRA has consisted of work to prevent migration of coal tar into the culverts including the sealing of expansion joints and weep holes in the culverts, and re-lining of catch basins and storm drain



pipes that lead to the culverts. As part of the IRA, a pilot DNAPL extraction well designated RW-1 was installed adjacent to the WEB Culvert in the central portion of Parcel E in fall of 2001 to remove DNAPL from the Site subsurface and to reduce the potential for DNAPL migration into the culverts. Approximately 670 gallons of mobile DNAPL have been extracted from RW-1 between October 2001 and May 2003. In addition, joint sealing and drain line lining activities conducted between 1997 and 2002 have resulted in a significant reduction in observed sheens in the culvert. The 2003 annual culvert inspection activities conducted in June 2003 indicated no observed coal tar sources (i.e., leaking joints, drain line outfalls, etc.) to the culvert and no sheens on the water surface in the culvert. Based on these observations, IRA activities appear to have resulted in a IRA Completion Report will be prepared and submitted to MADEP in the near future closing out the IRA.

f

7

# II. DISPOSAL SITE CONDITIONS

The Phase II Comprehensive Site Assessment compiled and summarized data collected during Phase I and II investigations at the Site. This section presents a summary of the information presented in the Phase II, including a summary of the Site geology and hydrogeology, Site physical conditions, the nature and extent of contamination at the Site (including a summary of the conceptual Site model), and the extent of MCP Upper Concentration Limit (UCL) exceedences at the Site.

### 2.01 Summary of Site Geology and Hydrogeology

# A. Subsurface Stratigraphy

The fate and transport of contaminants at the former Malden MGP Site is significantly influenced by subsurface geologic conditions. The following stratigraphic units have been identified (listed from surface downward), and are indicated on subsurface profiles in Figures 6A through 6G.

- Miscellaneous Urban Industrial Fill: Miscellaneous urban industrial fill at the Site typically contains fine sand and gravel with ash, cinders, slag, wood, steel, glass, plaster, asphalt, concrete and other urban debris. Fill thickness varies across the Site from approximately 2.5 to 17.5 ft.
- ш Organic Deposits: A layer of organic silt and peat underlies the urban fill in most of the northern portion of the Site. Where encountered, fibrous peat generally overlies and grades into organic silts. The top of the organic deposit, where present, is located approximately 4 to 17.5 ft below ground surface. This peat/organic silt zone is not continuous across the Site. It was not encountered in borings and test pit explorations on the southern portion of the Site, and the surface of the organic deposit is irregular in areas where it is encountered. The approximate extent of the organic deposit and a top of organic contour plan is shown on Figure 7. As can be seen in this figure, the organic deposit contains depressed areas, which generally coincide with the locations of former meandering stream channels associated with the location of the former West End Brook and Malden River, and the current location of the WEB and MR culverts. Additionally, some depressed areas coincide with historic MGP structures, indicating that a portion of the organic deposit may have been removed during construction of foundations for these structures. The organic deposit is believed to have been excavated from beneath the MBTA right-of way for geotechnical reasons during the construction of the railroad tracks in the early 1800s.
- Upper Sands: Medium to fine sand, coarse sandy gravel and gravelly sands generally lie below the organic deposits, ranging in thickness from approximately 2 to 6 ft. In areas where the organic deposits are not present,



the sand deposits appear thicker, ranging from approximately 8 to 17 ft. Thin silty sand seams are sometimes found within the sand or sandy gravel zones.

- Silty Fine Sands: Silty fine sands typically underlie the upper sand unit across the Site, ranging in thickness from approximately 15 ft to greater than 50 ft. The silty sands appear to become finer with depth; the presence of silt and clay seams also increases with depth within the silty sand unit.
- Clay: Silty clay was observed sporadically in test borings and was not identified in a continuous layer at depth. The clay deposits were typically observed in seams at the bottom of the silty fine sand unit.
- Lower Sands: In isolated areas, a lower unit of coarse sandy material was observed beneath the silty fine sands and clay seams.

The alternation of sand units to silty fine sands to clay, then back to sands followed by iner gradations likely repeats and extends below the depth of penetration of explorations conducted by Haley & Aldrich to the top of bedrock. The stratigraphy is consistent with the interpretation of the Site location in the path of glacial meltwater streams emptying into an emergent shallow marine environment. Variations in meltwater stream energies likely account for silty seams within the sand units, sandy and clay seams within the silty sands, and sand seams within the clay units.

# B. Hydrogeology

The former Malden MGP Site is located in a level, industrial setting. Two culverted surface water bodies, the WEB Culvert and the MR Culvert, flow across and adjacent to the Site. Shallow hydrogeologic conditions are primarily controlled by these culverted surface water bodies and the organic peat and silt deposits, which are present in the northern portion of the Site. The organic deposit represents the surface of an historic marsh; where encountered, the top of the organic deposit is located approximately 4 to 17.5 ft below the Site ground surface. The interpreted limits and elevation contour of the top of the organic deposit layer are presented on Figure 7. Low areas in the organic deposit are observed along historic river channels, and in areas in which the organic deposit was partially removed during construction activities, such as the vicinity of the former holder on Parcel A and along the culverts. The depth to groundwater ranges from approximately 4.5 to 12 ft below ground surface at the Site.

Water level measurements have been collected from Site monitoring wells to estimate the direction of shallow groundwater flow on the Site. Locations of Site monitoring v/ells, soil borings and test pits are shown on Figure 8. The shallow groundwater flow direction is significantly influenced by the organic deposit and the river channels on the Site. December 2000 water level measurements, which are representative of t/pical groundwater observations, are presented on Figure 9. As shown on this figure, the shallow groundwater flow direction varies across the Site, but generally is toward the south, parallel to flow in the Malden River. Ultimately, shallow



groundwater is believed to flow into the Malden River, which drains the area in which the Site is located and discharges into the Mystic River. The water table is generally flat, although along the eastern boundary of the Site, hydraulic gradients are steeper, as groundwater drains into the channel of the Malden River. The WEB Culvert strongly influences groundwater flow near the center of the Site, as groundwater appears to drain into the WEB or the crushed stone backfill placed beneath the WEB culvert. Comparison of the groundwater levels in Figure 9 with the organic deposit elevations and contour on Figure 7 suggests that groundwater elevations are slightly elevated in areas of the Site that are underlain by high spots in the organic deposits. These areas are on Parcels E and A north of the WEB culvert (e.g., B1-OW, B6-OW, and B15-OW), and on Parcel E south of the WEB culvert (e.g., B112B-OW, E106-OW, and B104-OW). Shallow groundwater in these areas of the Site appears to be prevented from flowing downward into deeper formations. Rather, shallow groundwater appears to flow from high areas in the organic deposit to lower areas, such as the WEB and MR culverts.

Evep groundwater is separated to some degree from shallow groundwater in the northern portion of the Site by the organic deposit. Deep groundwater elevations based on August 2001 measurements are presented in Figure 10. As shown in Figure 10, deep groundwater flow at the Site is generally toward the south-southwest, parallel to the flow of the Malden River. Ultimately, deep groundwater most likely discharges into either the Malden or Mystic River, which drain the area in which the Site is located and discharge into Boston Harbor. However, the gradient is very slight, and the flow direction may vary with slight fluctuations in hydraulic head.

Measurements of shallow and deep groundwater elevations indicate that in some areas of the Site, strong downward gradients exist across the organic deposit. This may indicate a hydraulic separation in these areas between the upper and lower formations due to the presence of the organic deposit. Where it is present and has not been punctured (i.e., through construction, excavation, etc.), the organic deposit may prevent shallow recharge from flowing into the deeper formations. At three monitoring well locations along the WEB and MR culverts, either slightly upward or weak downward gradients are observed, indicating that the organic deposit may not be continuous near these wells. During construction for the culverts, the organic deposits may have been excavated and removed, or punctured by the wooden piles that were driven to support the culvert. As a result, shallow groundwater is allowed to flow into the deeper formations, relieving the vertical gradient at these locations.

# 2.02 Site Physical Conditions and Current Use

The Site is located in a developed, urban area, on land that has an industrial history that extends over the past 150 years. MGP facilities that previously existed on the Site were demolished between 1965 and 1980, and new buildings were built on the Site. Little is known about the demolition practices, however Site information presented in the Phase II suggests that contamination remains beneath some of the buildings on the Site.



### A. Structures and Current Use

As mentioned above, the Site is located in a heavily developed, commercial/industrial area. As a result, there are nine major structures present at the Site, located on eight different properties. Buildings on the Site are shown on Figure 8. Table I lists the significant structures on the Site by Parcel.

#### TABLE I

# SITE STRUCTURES AND CURRENT PROPERTY USE

Parcel Designation	Property Address	Structure Description	Current Property Use
Parcel A.	51 Commercial St.	1-story building	Dentist Office/ Chiropractor Office
	65 Commercial St.	1-story building	Muffler Service Station
	77 Commercial St.	1-story building	Liquor Store
	89 Commercial St.	1-story building	Auto Body Service Station
	99-109	99-103 Commercial Street; 2-story building	Auto Glass, Tanning Salon, and Print Shop
	Commercial St.	105-109 Commercial Street; 2-story building	Medical Offices
Parcel B	129 Commercial St.	2-story commercial building with office space	Commercial Bakery and Frozen Storage
Parcel D	Callahan Park	None; Asphalt pavement	Parking
Parcel E	100 Commercial St.	1-story building; office and mechanical space	KeySpan Operations Facility
	100 Commercial St.	1-story Maintenance Garage	KeySpan Vehicle Maintenance Facility

# B. Utilities

Utilities located on the Site include gas, water, electric, sewer, and drain pipes, located both beneath the city streets as well as on private property. Some of the more notable utilities include a 24-in. high-service Metropolitan Water Resources Authority (MWRA) water pipe that is located beneath Charles Street, as well as gas, electric, water and sewer utilities servicing buildings located on Parcels A, B and E. Additionally, the WEB Culvert transects the 100 Commercial Street property, and the MIR Culvert lies along the eastern border of the 100 Commercial Street property. It is also expected that some historic, abandoned gas lines remain from the operational facilities of the historic MGP.



Characterization is located in Appendix B and the findings of the Risk Characterization are summarized in Section 2.05.

#### A. Results of Soil Sampling Conducted at 51 Commercial Street

The additional investigations conducted on the 51 Commercial Street property included the collection of three soil samples, which were collected from 0 to 3 ft below ground surface. Locations of the additional samples are shown on Figure 11, and analytical results are presented in Table II. The sample results indicate slightly elevated concentrations of PAHs and extractable petroleum hydrocarbon (EPH) aromatic compounds; however UCL exceedences were not observed, and the concentrations are generally lower than those observed in MGP-impacted soils on the Site. VOC, metals, and cyanide concentrations were below applicable MADEP standards. The concentrations observed in the three new samples were averaged together with the one previous sample result from 0 to 3 ft below ground surface to re-calculate the risk to the current landscaper on this property. Analytical data reports for the new samples are included in Appendix C.

#### B. Eesults of Supplemental Investigations at 129 Commercial Street

As was outlined in the Phase II and below in Section 2.04D, a source of BTEXSN compounds likely exists in soil beneath the two-story commercial building located at 129 Commercial Street. Additional investigations were conducted at the 129 Commercial Street Property to more closely estimate the location of the source of contamination as well as more closely define the nature and extent of contamination at the property. The additional investigations conducted on the property consisted of the iristallation of soil borings, monitoring wells and a temporary soil vapor monitoring point through the floor of the 129 Commercial Street building. Because the 129 Commercial Street building is the location of an operating commercial business, the investigation has been conducted in stages organized around the operating schedule of the business. Seven borings and 5 monitoring wells were installed in March 2002, and one additional boring with a soil vapor sampling point was installed in July 2002. Groundwater samples were collected from the five monitoring wells inside the building in March, May, and July/August 2002. Locations of new and previously existing borings and monitoring wells on the 129 Commercial Street property are shown on Figure 12, and analytical results obtained from the recent soil and groundwater samples are presented in Tables III and IV, respectively. Indoor air sampling was also conducted in October 2001, January 2002, April 2002, June 2002, October 2002, and January 2003 as part of the ongoing RAM at the 129 Commercial Street property. These data are presented in Table V. Boring logs, monitoring well installation records, and analytical data reports for soil, groundwater and indoor air are included in Appendix C.

In summary, the additional investigations conducted at the 129 Commercial Street property did not encounter a discrete source of contamination (such as a tank, drip pot, or other container) beneath the building. Rather, the results indicate the presence of BTEXSN contamination in soil and groundwater at several exploration locations.



Stained soil was observed in soil boring samples in a zone approximately 2 to 6 ft in thickness, located at or near the water table elevation. Visual observations and analytical results indicate the highest concentrations of BTEXSN in soil were located in borings 02B-B922, 02B-B923, and 02B-B924, located in the central portion of the building. The highest BTEXSN concentrations in groundwater were observed at monitoring location 02B-B918-OW, located downgradient of these locations in the southern portion of the building. The results of these investigations support the conceptual model presented in the Phase II, which contends that contaminated soil is located beneath the 129 Commercial Street building. As groundwater flows beneath the building, BTEXSN compounds are dissolved from the contaminated soil into and flow with groundwater. The elevated BTEXSN concentrations in soil and groundwater beneath the building appear to result in the elevated indoor air concentrations of these compounds.

Section 2.04D contains additional information concerning the nature and extent of contamination on the 129 Commercial Street property.

# 2.04 Nature and Extent of Contamination: Conceptual Site Model Summary

The former Malden MGP operated from the mid-1800s until the early 1970s. The types and levels of contaminants detected in soil, groundwater and sediment are consistent with this long industrial history. As described in Section 1.03, MGP operations at the former Malden MGP facility used coal, coke, and oils as raw feedstock for combustion in retorts and produced a number of residuals while processing the generated gas and separating impurities prior to gas distribution. MGP residuals included solid residue from the retorts, hydrocarbon/aqueous condensate from gas separator units (e.g., condensers, tar separators), and solid wastes from the purification process.

In general, the predominant classes of chemical compounds identified as the primary residuals from the processes used at the former Malden MGP include PAH, VOCs, and cyanide. The presence and ultimate form of the process residuals and associated chemicals currently present at the Site are a function of such factors as the extent and management of materials on-site during former MGP operations, the disposition of residuals, and the local subsurface or environmental conditions.

The former Malden MGP Site Conceptual Site Model (CSM), as presented in the Phase II, represents a synopsis of our understanding of the conditions present at the Site, including sources of contamination, affected media (i.e., soil, groundwater, etc.), potential exposure pathways and potential exposure points. The Phase II CSM identifies contaminant source areas, consistent with typical MADEP CSM terminology and based on observed contaminant distribution and the current understanding of historic MGP operations. For this Phase III RAP, seven types of contamination have been identified at the Site to facilitate the development and evaluation of remedial alternatives. For detailed information relating to Site characterization information, refer to the Phase II. The types of contamination observed at the Site are described below in a summary of the Conceptual Site Model for the Site.



has extracted approximately 670 gallons of mobile DNAPL from the central portion of Parcel E between October 2001 and May 2003. These results support the hypothesis that a portion of the coal tar beneath the Site is extractable DNAPL.

1

Two surface water bodies lie on or adjacent to the portion of the Site impacted by TSM and Shallow DNAPL. The WEB Culvert transects the TSM area, flowing from the western boundary of TSM impacts toward the east, where it discharges into the MR Culvert, which is located along the eastern Site boundary. South of Charles Street, the MR Culvert discharges and the Malden River flows as an open channel toward the Mystic River. TSM impacts have been observed in sediments within the WEB and MR culverts and in the Malden River downstream of the culvert outfall. An IRA was initiated to address the migration of DNAPL into the culverts. As part of the IRA, catch basins and storm drains in DNAPL -impacted areas of the Site were lined, and expansion joints and weep holes in the culverts in DNAPL-impacted areas have been sealed to prevent DNAPL infiltration. Annual inspections of the culvert have been conducted to confirm that these measures have achieved the desired result of reducing or eliminating DNAPL infiltration to the culverts. As discussed in Section 1.04, the 2003 annual culvert inspection activities conducted in June 2003 indicated no observed coal tar sources (i.e., leaking joints, drain line outfalls, etc.) to the culvert and no sheens on the water surface in the culvert. Based on these observations, IRA activities appear to have been successful, and it is anticipated that an IRA Completion Report will be prepared and submitted to MADEP in the near future closing out the IRA.

Culvert IRA activities have also included limited assessment of the crushed stone layer beneath the WEB Culvert, which have indicated that DNAPL is present at some locations beneath the WEB Culvert. In response to these observations, grout was injected beneath the MR Culvert at Station 7+30 in December 1998 to limit or eliminate the potential for DNAPL migration beneath the MR Culvert. Additionally, test holes drilled through the base of the MR Culvert into the crushed stone base near the MR Culvert outfall did not detect DNAPL.

# C. Deep DNAPL

Coal tar residuals resulting from the production of manufactured gas at the former Malden MGP exists in the environment as a separate DNAPL phase, which tends to migrate vertically downward through soil and groundwater. A large portion of the former Malden MGP Site is underlain by a layer of organic deposits (the top of which ranges from 4 to 17.5 ft below ground surface), which impedes downward migration of coal tar DNAPL. In these areas, DNAPL tends to migrate horizontally along the upper surface of the organic deposit. As a result, DNAPL has not been detected in most deep wells on the Site. However, the organic deposit is absent in the southern portion of the Site, and in areas where it is present, it is discontinuous and has been penetrated or removed during various construction activities. These gaps in the organic deposit beneath the Site have likely allowed coal tar to migrate vertically as DNAPL into the deeper units below, resulting in both soil and groundwater impacts beneath the organic deposit.



The quantity of DNAPL that exists in the deeper subsurface is difficult to estimate. Eased on the deep exploration data collected at the Site, the extent of DNAPL in the subsurface beneath the organic deposit is limited. DNAPL has been detected in three of twelve deep wells on the Site, as reported in the Phase II and shown on Figure 13. In one of these three wells, B301L-OW, DNAPL was detected in past observations (1997 and 1998), however measurements made in July and August 2002 indicate less than 1/2 in. of DNAPL in the bottom of this well. In another of these three wells, B110A-OW, five measurements made between 1994 and 1999 observed a thickness of as much as 10 ft of DNAPL. Accumulated DNAPL was evacuated from B110A-OW using a vacuum truck in 1999. Subsequent DNAPL monitoring consisting of greater than 70 measurements between June 2000 and December 2001 indicated a thickness of less than 0.3 ft. The third of the deep monitoring wells on the Site that have been impacted by deep DNAPL, B108-OW, does contain a significant DNAPL thickness (77 measurements made between 1991 and 2001 detected an average DNAPL thickness of approximately 6.9 ft). However, observations made during installation of the test boring indicate that the source of the DNAPL in this well may be a section of scil 3.5 ft thick that is approximately 3 ft above the bottom of the well. Therefore, it is likely that DNAPL flow into the well from this zone may result in an accumulation of several ft of DNAPL in the well. Based on these observations, a limited quantity of deep DNAPL may require remediation.

# D. LINAPL

LNAPL at the former Malden MGP Site is comprised of oils that represent the lighter fraction of MGP residuals, which are less dense than water. The approximate limits of observed LNAPL impacts are shown on Figure 13; LNAPL impacts have been observed across an area of approximately 21,000 sf on the southern portions of Parcels A and E, and a portion of Commercial Street. LNAPL has also been detected on the northern portion of Parcel E in monitoring well B1-OW; however LNAPL has not been observed during recent monitoring in this well. The presence of LNAPL has resulted in impacts to shallow soils and groundwater on the Site. Based on the estimated areal extent and an observed average thickness of approximately 0.35 ft over the impacted area, the estimated quantity of LNAPL present on the Site is approximately 9,400 gallons. However, estimation of LNAPL quantity based on thickness in monitoring wells is dependent upon several factors, including LNAPL characteristics, soil type, water table fluctuations, and other factors. Therefore, this estimate should be considered approximate. It is possible that the actual distribution of LNAPL in the subsurface is more sporadic over the LNAPL area, and the quantity of LNAPL that actually exists is significantly less than the amount estimated above. Observations over the duration of Phase II investigations indicate that the LNAPL area is stable, and that LNAPL is not migrating to new areas of the Site. LNAPL impacts also do not appear to be contributing significantly to groundwater impacts at the Site based on observed groundwater concentrations in areas downgradient of the LNAPL-impacted area.

During Phase II investigations at the site, several LNAPL removal methods were pilot-tested. These efforts included testing of "siphon without a pump" (SWAP)



technology, which is a passive LNAPL removal system, use of absorbent pads for removal of LNAPL from monitoring wells, and vacuum extraction methods of LNAPL removal. The rate of LNAPL removal achieved by these methods is not anticipated to be applicable to removal of LNAPL on a site-wide basis.

# E. ETEXSN in Soil and Groundwater

Farcel B was used by the MMGLC/MVGC for gas purification beginning from approximately 1900 to the early 1970s. As a result, BTEXSN compounds and cyanide are the primary contaminants that are observed in soil and groundwater on Farcel B.

Data collected during the Phase II and subsequent investigations indicate the presence of BTEXSN contamination in soil and groundwater beneath the 129 Commercial Street building. Maximum concentrations of total BTEXSN observed in groundwater samples collected from wells and borings on Parcel B are presented on Figure 14. Subsurface investigations to date have encountered moderately contaminated soil at and below the water table elevation. Based on VOC concentrations observed in groundwater, heavily impacted soil may be present beneath the building, located either above or below the water table elevation, in areas where investigation is not currently possible. Subsurface investigation has not been possible in some areas of the facility due to the presence of large, immovable features inside the building. Therefore, although heavily impacted soil has not been observed to date above the water table elevation, it is considered possible that it may exist in some areas beneath the building. As shown on Figure 14, BTEXSN concentrations in groundwater are very low on the north (upgradient) side of the building, and are elevated in groundwater samples collected from explorations underneath and downgradient of the building. From these investigations it appears that shallow groundwater (i.e., approximately 10 to 12 ft below ground surface) dissolves BTEXSN constituents as it flows through source material in soil beneath the building located below the water table elevation. Several historic features have been identified on historic plans, including the drip house, drip tank, the unlabelled subsurface tank, and the purifier boxes shown on Figure 14, however investigation activities have not located these features.

As mentioned in Section 2.02, a commercial bakery currently occupies the 129 Commercial Street building. Several large structures associated with production and storage are located inside the facility, including a blast freezer (approximately 3,870 sf), a storage freezer (approximately 9,450 sf), and large baking ovens. The presence of these facilities precludes further exploration to define the location of the source material beneath the building.

The recent subsurface investigations conducted during March through August 2002 investigated possible locations of source material beneath the building. Test borings were installed at the estimated locations of several of the historic features listed above. Investigation locations are presented on Figure 14. Stained soil was encountered at and below the water table elevation (approximately 11 to 16 ft below the building



floor slab) in the central portion of the building. Borings installed at the estimated locations of historic MGP structures (an historic drip house, drip tank, and unlabelled subsurface tank) did not indicate that these locations were likely discrete sources of groundwater contamination. Based on visual observations and analytical data obtained during the recent investigation, the greatest degree of soil impact was observed in the central portion of the building, in borings 02B-B922-OW, 02B-B923-OW, and 02B-B924-OW. Further explorations within the building are precluded at this time due to the internal configuration and current use of the building. However, the results of recent investigations appear to indicate that the source of the groundwater contamination is likely beneath the central portion of the building. As shown on Figure 14, historic MMGLC purifier boxes were located in this general vicinity; the earliest available reference on which these appear is a plan dated 1912.

Eecause the cyanide contamination that is observed in groundwater samples from monitoring well 97B-B628-OW is upgradient of the observed VOC contamination, the source of the cyanide groundwater contamination on Parcel B is most likely different than the source of the VOC contamination. Furthermore, removal of cyanide at historic MGPs typically occurred in separate structures than those used for removal of VOC impurities. Because cyanide was not detected in soil samples from 97B-B628-OW, it is likely that the cyanide detected in groundwater at this monitoring location originates from upgradient of this location, either on Parcel B or in the Charles Street right-of-way. The Phase II Risk Characterization demonstrated a condition of No Significant Risk related to exposure to cyanide in soil and groundwater on Parcel B.

Impacted media on Parcel B include soil, groundwater, and indoor air. The most significant contamination has been detected in groundwater and indoor air. Based on our current understanding of subsurface conditions beneath the 129 Commercial Street building, indoor air impacts appear to result from volatilization of contaminants from impacted soil and groundwater. The presence of VOCs in groundwater is likely the result of dissolution of VOCs from source material (i.e., impacted soil) beneath the building. As a result, anticipated response actions at 129 Commercial Street will be linked to remediation of VOC impacts in soil and groundwater beneath the building. The area of impacted groundwater beneath the 129 Commercial Street building is indicated on Figure 14. Concentrations of total BTEXSN in groundwater samples collected from this area have been as high as approximately 295 mg/l. Additionally, contaminated soil has also been observed beneath the building in a portion of this area. The area of impacted soil is estimated to be approximately 21,000 sf. Assuming an average of a four-ft thick area of soil impact at the water table, as described above, the volume of impacted soil is approximately 3,100 cubic yards (cu yd).

# F. Fetroleum-impacted Soil

Fetroleum-impacted soil associated with the former tank farm is located in the vicinity of the historic above-ground petroleum and coal tar tanks on the northern half of Farcel E, just north of the WEB culvert, as shown on Figure 13. The primary



contamination observed in this area appears to be petroleum impacts associated with the historic fuel tanks. Generally, soils in this area are not saturated with coal tar, as is the case in the TSM-impacted areas on other portions of Parcel E. However, some soil samples in this area contained a mixture of tar-related compounds and petroleum compounds, indicating a mixture of contamination types in this area. This evidence is in agreement with historic MMGLC plans and with the CSM presented in this report, which indicate that the above-ground storage tanks were used for storage of both coal tar and petroleum during different periods of operation.

Based on the extent of petroleum impacts reported in the Phase II, the areal extent of the petroleum-impacted soil contamination is approximately 28,800 sf (3,200 square yards [sy]), and the vertical extent is variable. For the purposes of this Phase III RAP, the vertical extent is assumed to be approximately 12 ft below ground surface, the approximate depth of the upper surface of the organic deposit. Based on this assumption, the volume of petroleum-impacted soil is approximately 12,800 cu yd. As shown on Figure 13, a portion of this contamination is located beneath the KeySpan maintenance garage, which is located on the 100 Commercial Street property.

## 2.05 Risk Characterization Summary

A Method 3 Risk Characterization was conducted by AMEC Earth & Environmental (AMEC) of Westford, Massachusetts, in accordance with MCP requirements. The results of the Risk Characterization, which were presented in the Phase II Report, included a characterization of the risk to human health, public welfare and the environment attributable to the Former Malden MGP Site. Since submittal of the Phase II Report, AMEC has revised the Risk Characterization, based on new Site data described in Section 2.03. This work included a re-characterization of the risk to human health risk at the 51 and 129 Commercial Street properties, incorporating new Site information that was obtained during work associated with the Phase III evaluation, and a Substantial Hazard Evaluation for the Site. Documentation of the revised Risk Characterization (as revised) is presented in this section of the Phase III. As mentioned above, evaluation of remedial alternatives for the Malden River sediment portion of the Site is not included in this Phase III RAP. Therefore, results of the Risk Characterization pertaining to the Malden River have not been presented in this section.

#### A. General

The Site Risk Characterization involved evaluation of increased potential risk at the Site due to exposure to MGP residuals. Because land use was varied across the Site, different exposure pathways were evaluated at each property, based on current and anticipated future land use on each property. For the purposes of the Risk Characterization, each of these properties were treated as individual "sites" for which risks were calculated individually. The Charles Street parcel, which includes the section of Charles Street south of Callahan Park, is evaluated separate from the Governor House portion of the Site in the Risk Characterization. However, in this Phase III RAP, the Charles Street and the Former Governor House parcels are



tcgether referred to as the Governor House portion of the Site. See Figure 3 for the lccations associated with the addresses listed below. The properties evaluated for risk are summarized in Table VI, below:

#### TABLE VI

SUMMARY OF PROPERTIES INCLUDED IN RISK CHARACTERIZATION

Property Address	Parcel on which Property is Located	Current Property Use		
51 Commercial Street	Parcel A	Dentist Office and Chiropractor Office		
65 Commercial Street	Commercial Street Parcel A Muffler Service Station			
77 Commercial Street	Parcel A	Liquor Store		
89 Commercial Street	Parcel A	Auto Body Service Station		
99 – 109 Commercial Street	Parcel A	Auto Glass, tanning Salon, printing shop, medical offices and other office space		
100 Commercial Street Parcel E		KeySpan operations and vehicle maintenance facility		
129 Commercial Street	Commercial Street Parcel B Commercial Bakery			
Charles Street, between Commercial and Pearl Streets	N/A	City Street		
Giovernor House	Parcel D	Parking lot for Callahan Park		

# B. Human Health Risk Characterization

#### 1. Current Site Uses

The revised MCP Method 3 Human Health Risk Characterization (Appendix B) demonstrated a condition of No Significant Risk to Human Health for current Site uses on the terrestrial portion of the Site. Differences between the results of the Phase II Risk Characterization and the revised Risk Characterization presented in Appendix B are discussed below.

The original Phase II Risk Characterization did not demonstrate a condition of No Significant Risk to the current landscaper at 51 Commercial Street. However, this result was based on one sample collected from beneath a paved parking lot, which did not accurately represent the exposure point, and therefore, did not accurately characterize the risk to the landscaper. Additional soil samples were collected from the landscaped areas of the property, and the revised Risk Characterization was conducted using this data and the previously collected data point. The revised Risk Characterization demonstrated a condition of No Significant Risk to the current landscaper at 51 Commercial Street.



The original Phase II Risk Characterization for the 129 Commercial Street Property could not demonstrate a condition of No Significant Risk to the current commercial worker via inhalation of indoor air. This result was based on indoor air data collected between September 1999 and October 2001. However, the revised Risk Characterization, which was based on indoor air data collected between September 1999 and October 2002, demonstrated a condition of No Significant Risk to the current commercial worker via the indoor air pathway. This change is likely due to the lower VOC concentrations that have been observed during recent indoor air sampling events inside the facility undertaken as part of the ongoing RAM.

Potential Future Site Uses

The Risk Characterization for the Site was conducted based on the assumption that current use of land on the Site is continued into the future. Therefore, potential exposure pathways that were evaluated for future use in the Risk Characterization include those that may associated with the future commercial/industrial use of the Site (i.e., commercial/industrial workers, construction workers, and utility workers). In the event that a change in the use of a property is proposed (e.g., conversion of commercial or industrial properties to residential use), the remedial approach to that property would be re-evaluated in the context of the proposed use.

As previously discussed, the Phase II Risk Characterization demonstrated a condition of No Significant Risk at the Governor House portion of the Site and for the Charles Street public right-of-way. The Risk Characterization could not demonstrate a condition of No Significant Risk to Human Health for some potential future exposure pathways, as summarized in Table VII below:



#### TABLE VII

#### SUMMARY OF HUMAN HEALTH RISK CHARACTERIZATION FUTURE EXPOSURE PATHWAYS

Receptor	Location	Pathway <sup>(1)</sup>	Compound(s) of Concern	
Future Construction V/orker	100 Commercial Street	Ambient Inhalation from excavation	Naphthalene	
Future Construction Worker	100 Commercial Street (Hot Spot 3)	Ambient Inhalation from excavation	Naphthalene, Benzene, C <sub>9</sub> -C <sub>10</sub> Aromatics, Methylnaphthalene	
Future on-site Commercial/Industrial worker	100 Commercial Street	Soil Ingestion	PAHs, arsenic	
Future Construction Worker	99-109 Commercial Street	Ambient Inhalation from excavation	Naphthalene, Benzene, Aromatics	
Future Construction Worker	99-109 Commercial Street (Hot Spot 2)	Ambient Inhalation from excavation	Naphthalene, Benzene, C <sub>9</sub> -C <sub>10</sub> Aromatics, Trimethylbenzene, Methylnaphthalene	
Future On-site Commercial/Industrial Worker	99-109 Commercial Street	Soil ingestion	PAHs	
Future Construction Worker	89 Commercial Street 77 Commercial Street	Ambient Inhalation from excavation	Naphthalene, Methylnaphthalene	
Future On-site Commercial/Industrial Worker	89 Commercial Street 77 Commercial Street 65 Commercial Street 51 Commercial Street	Soil ingestion	PAHs	
Future Construction Worker	65 Commercial Street	Ambient Inhalation from excavation	Naphthalene, Methylnaphthalene, Cg-C10 Aromatics, C11-C12 Aromatics	
Future Construction Worker	51 Commercial Street	Ambient Inhalation from excavation	Naphthalene, Methylnaphthalene, Trimethylbenzene	

Notes:

Explanation of Exposure Pathway terms: Ambient Inhalation from excavation: Inhalation of ambient (i.e., outdoor) air containing VOCs volatilized from soil exposed during excavation



# C. Risk to Public Welfare

Risk to public welfare was also calculated in the Risk Characterization. As specified in the MCP, the risk to public welfare is based on the following: (1) a comparison of property-wide and hotspot average soil and groundwater concentrations to the UCLs; (2) comparison of indoor air and ambient vapor concentrations to odor thresholds; (3) the presence of NAPL having a thickness equal to or greater than ½ in.; and (4) comparison of estimated airborne vapor concentrations to odor thresholds. A condition of No Significant Risk to Public Welfare was demonstrated for 129 Commercial Street, 77 Commercial Street, Charles Street right-of-way, and the former Governor House.

۶.

A condition of No Significant Risk to Public Welfare was not demonstrated for 100 Commercial Street, 99-109 Commercial Street, 89 Commercial Street, 65 Commercial Street, and 51 Commercial Street based on exceedences of UCLs. NAPL greater than <sup>1</sup>/<sub>2</sub> in. has been identified in monitoring wells located at 100 Commercial Street, 99-109 Commercial Street, 89 Commercial Street, 65 Commercial Street, and 51 Commercial Street. In addition, site-wide soil concentrations for one or more constituents exceeded their respective UCLs in these areas.

Table VIII lists the properties at which a condition of No Significant Risk to Public Welfare could not be satisfied, based on the Risk Characterization.

#### TABLE VIII

SUMMARY OF PUBLIC WELFARE RISK CHARACTERIZATION RESULTS

Location	Compound(s) of Concern
100 Commercial Street	>0.5 in. of NAPL; concentrations of compounds > UCL
99-109 Commercial Street	>0.5 in. of NAPL; concentrations of compounds > UCL
89 Commercial Street	>0.5 in. of NAPL; concentrations of compounds > UCL
65 Commercial Street	>0.5 in. of NAPL; concentrations of compounds > UCL
51 Commercial Straet	>0.5 in. of NAPL; concentrations of compounds > UCL

#### D. Environmental Risk

Aside from the Malden River section of the Site, which has been excluded from this Phase III RAP, environmental resource areas have not been identified at the Site. Therefore, a condition of No Significant Risk to the environment was demonstrated for the terrestrial portion of the Site.

# E. Substantial Hazard Evaluation

An evaluation of the Substantial Hazards that exist on the Site is required for Sites at which a condition of No Significant Risk could not be satisfied for current exposure pathways, and for Sites at which a Temporary Solution may be appropriate.



Achievement of a Temporary Solution (i.e., a Class C RAO) requires elimination of substantial hazards that have been identified on the Site. AMEC conducted a Substantial Hazard Evaluation (SHE) for the terrestrial portion of the Site. A copy of the AMEC Substantial Hazard Evaluation is part of the revised Risk Characterization included in Appendix B.

The purpose of the SHE is to evaluate the level of potential risk that is present at the Site under current Site use. The SHE is comprised of two components: human health and ecological evaluation. For the human health portion, the SHE evaluates potential risk in a manner similar to the Risk Characterization, except the SHE evaluates potential risk posed over a reduced exposure period. The exposure period considered for the SHE is equal to the period of time elapsed between notification to MADEP of the presence of hazardous materials at the Site and the date that the SHE is conducted, plus an additional five years. The results of the SHE are compared to the same thresholds as the standard Risk Characterization; as a result, a risk that is considered a Substantial Hazard indicates a greater degree of risk to the exposed population than failure to demonstrate a condition of No Significant Risk. The Human Health SHE indicated a result of No Substantial Hazard for the terrestrial portion of the Site.

The ecological portion of the SHE is not as quantitative as the human health portion. The criteria listed in the MCP are listed in the AMEC report in Appendix B, and include the following:

- Evidence of stressed biota
- Visible presence of oil or hazardous materials (OHM) within 3 ft of the ground surface over two acres
- Continued discharge of contaminated groundwater to surface water where concentrations of Site-related OHM in surface water and/or sediment exceed Massachusetts surface water standards
- Continued discharge of contaminated groundwater to surface water where concentrations of Site-related OHM in surface water and/or sediment already pose a significant risk
- Migration of OHM to additional environmental media or resource area where exposures would potentially pose a significant risk of harm in the future, and
- Ecological risk or harm such that recovery would be substantially more difficult or would require more time if the Site was not remediated for even a short period of time

The ecological receptor associated with the Former Malden MGP Site is the Malden River. The Malden River itself has been excluded from this Phase III RAP and will be evaluated in the future in the context of the ongoing MRA and the Army Corps of Engineers Malden River Ecosystem Restoration Feasibility Study. However, the terrestrial portion of the Site that is evaluated in this Phase III RAP does include the MR Culvert, which conveys the Malden River along the eastern Site boundary, and the WEB Culvert, which empties into the MR Culvert. As described in Section 2.04A, TSM impacts have been observed within the WEB and MR Culverts, and coal tar DNAPL has been detected in the crushed stone bedding beneath the WEB Culvert.



Several remedial measures have been undertaken in response to these observations. An ongoing IRA was initiated in response to the observation of an intermittent sheen on the water surface in the MR Culvert. This response action has included the sealing of expansion joints and weep holes in the culverts, and re-lining of drain pipes that empty into the culverts. In addition, grout was injected beneath the MR Culvert at a location just downstream of the confluence with the WEB Culvert to prevent DNAPL migration in the crushed stone bedding. Annual inspections of the culverts have been conducted to check the integrity of these seals. The 2003 annual culvert inspection activities conducted in June 2003 indicated no observed coal tar sources (i.e., leaking joints, drain line outfalls, etc.) to the culvert and no sheen on the water surface in the culvert. These observations indicate that the response actions have successfully addressed the IRA conditions and it is anticipated that an IRA Completion Report will be prepared and submitted to MADEP in the near future closing out the IRA.

Because steps have been and continue to be taken to address the potential for Site contaminants to impact the WEB and MR Culverts and the Malden River, the SHE concludes that a condition of No Substantial Hazard exists at the Site, in accordance with the MCP.

# F. Conclusions Based on Risk Characterization Results

Based on the results of the revised Risk Characterization, a condition of No Significant Risk could not be satisfied for future potential exposure pathways at 51, 65, 77, 89, 99-109, and 100 Commercial Street. A condition of No Substantial Hazard was satisfied for the Site.

In Section III of this report, remedial objectives are outlined for achievement of a Permanent or Temporary Solution at the Site. The remedial objectives were developed based on the results of the revised Risk Characterization summarized in this section, with the goal of eliminating or mitigating potential risks caused by potential exposure to Site contamination.

The revised Risk Characterization for the Site demonstrated a condition of No Significant Risk to Human Health for the 129 Commercial Street property. However, the Risk Characterization for the indoor air inhalation pathway at this property was based on data that were collected during the operation of a pilot-scale mechanical subslab venting system, which was installed for the purpose of mitigating the migration of VOCs into indoor air, thereby reducing VOC concentrations in indoor air. This system was installed as a temporary measure and, as such, this Phase III RAP includes an evaluation of remedial alternatives for reducing concentrations of VOCs in indoor air.

## 2.06 Remedial Action Alternative Areas

In order to facilitate the evaluation of potential remedial technologies and alternatives, the Site has been divided into five Remedial Action Alternative Areas (referred to herein as "Areas") based on property boundaries, characteristics, and type of contamination present. A summary


of the five Areas, including a brief description and lists of the types of contamination and impacted media is presented below in Table IX.

#### TABLE IX

SUMMARY OF REMEDIAL ACTION ALTERNATIVE AREAS AND ASSOCIATED IMPACTED MEDIA

Site Area	Area Description	Properties Located Within The Area	Types Of Contamination Identified Within The Area	Impacted Media Contributing To Risk
AREA 1	Northern Portion of Parcel E	100 Commercial Street	TSM Shallow DNAPL Deep DNAPL LNAPL Petroleum-impacted Soil	DNAPL LNAPL Soil
AREA 2	Southern Portion of Parcel E (includes WEB Culvert)	100 Commercial Street	TSM Shallow DNAPL Deep DNAPL LNAPL	DNAPL LNAPL Soil
AREA 3	Northern portion of Parcel A	51 Commercial Street 65 Commercial Street 77 Commercial Street 89 Commercial Street	TSM Shallow DNAPL	DNAPL and Soil
AREA 4	Southern portion of Parcel A	99-109 Commercial Street	TSM LNAPL	LNAPL Soil
AREA 5	Parcel B	129 Commercial Street	BTEXSN in soil and groundwater	Soil Groundwater Indoor Air

The boundaries of the five Remedial Action Alternative Areas, along with the Site boundary, are shown on Figure 15. As can be seen on Figure 15, public rights of way, such as Commercial Street, Charles Street and Centre Street have not been included in the Remedial Action Alternative Areas. MGP impacts are believed to be less beneath the roadways than on gas manufacturing areas. These roadways existed at the time the MGP was in operation, and therefore it is considered unlikely that MGP waste is contained in shallow soils beneath the roadways. It is likely that mobile contaminants, such as LNAPL and shallow DNAPL, exist beneath the roadways due to flow from adjacent properties. These impacts would be located near the water table or the top surface of the organic deposit, and would likely be located below the depth of most utilities. To the extent practical, remedial measures on properties adjacent to these areas will be designed to reduce contamination in these areas, such as design of DNAPL and LNAPL systems with radii of influence that extend beneath the adjacent roadways. Aside from these efforts, implementation of remedial measures in these areas would be difficult and are considered infeasible at this time due to the presence of numerous utilities and heavy vehicular use. Furthermore, this Phase III RAP does not include an



evaluation of AULs to restrict exposures to contaminants located beneath public rights-ofway, as AULs are not required by the MCP, per 310 CMR 40.1012(3).

Based on Site monitoring data collected to date the potential for exposure to LNAPL and DNAPL beneath the streets is limited based on the probable location of these media. LNAPL has been detected in monitoring wells located along Commercial and Charles Streets (97A-B601-O'W, 99A-B816-OW and B109A-OW) at depths ranging from approximately 5.3 to 9.9 ft bgs, and shallow DNAPL has been detected in wells located adjacent to Commercial Street (E502-OW, 99E-B822-OW, and 00A-B914-OW) at depths ranging from approximately 11.2 to 14.5 ft bgs. Based on these observations, it is considered unlikely that LNAPL or DNAPL would be encountered during routine utility work beneath the public right-of way portions of the Site.

It is also noteworthy that, as shown in Table IX above, although groundwater contains elevated concentrations of contaminants, remediation of groundwater on Areas 1 through 4 is not required to achieve a condition of No Significant Risk. Groundwater is not used on these Areas as a resource (i.e., for drinking water or industrial use), and based on groundwater sampling data obtained during the Phase II investigation, plumes of groundwater contamination do not appear to be leaving the Site. Additionally, it is anticipated that remedial approaches to remediate LNAPL, DNAPL and soil will have a beneficial effect on groundwater quality. Therefore, Remedial Action Alternatives developed for Areas 1 through 4 will not include an evaluation of remedial components for groundwater. A groundwater monitoring program will be conducted during implementation of the selected remedy.

The boundaries, current use, characteristics and known extent of contamination on the five Remedial Action Alternative Areas, referred to herein as "Areas", are briefly described below. An evaluation of remedial alternatives is presented for each Area in Sections V through X.

#### A. Area 1, Northern Portion of Parcel E

Area 1 is the northern portion of 100 Commercial Street, the property currently owned by KeySpan. Area 1 is bounded by Commercial Street to the west, Centre Street to the North, the MR Culvert to the east and Area 2 to the south. Area 1 is approximately 2.3 acres, consisting of a maintenance garage (approximately 3,200 sf), parking areas, storage space for KeySpan and minimal landscaped space. Service trucks and other vehicles use Area 1; as a result there is relatively high truck traffic volume on the Area. KeySpan operations currently require use of the parking and storage areas and the maintenance garage on Area 1. However, KeySpan has indicated that use of Area 1 facilities may be discontinued to allow remedial actions to take place. Therefore, KeySpan business activities are not considered in the evaluation of the feasibility of remedial alternatives on Area 1.

1. Area 1 Characteristics

Area 1 is underlain by a relatively impermeable organic peat and silt deposit, ranging from approximately 4 to 17.5 ft below ground surface. The organic



deposit separates shallow groundwater flow from deeper groundwater beneath the organic deposit, and controls the migration of shallow DNAPL. Shallow DNAPL flow is interpreted to be controlled by pressure head and gravity (i.e., shape of the organic deposit). Groundwater flow above the organic deposit is separated from deeper groundwater flow by the organic deposit. An historic river channel (the Malden River, before it was channeled into the MR Culvert) is located on the northern half of this Area. The river channel, which is now filled, is located beneath the parking area on the northern portion of the Area. Water in the channel historically flowed from the alignment of the existing Malden River channel at the eastern end of Area 1 to the western Area boundary. It then flowed onto Parcel A (Area 3), where the channel meandered around to the southeast, at which point it merged with the alignment of the existing WEB culvert.

Historic MGP facilities on Area 1 include the Parcel E Tank Farm, which was located on the southern portion of Area 1, just north of the alignment of the existing WEB culvert. Known historic features on Area 1 include six aboveground storage tanks and the Hood Garage (unrelated to the MGP), located at the northwestern corner of Area 1.

#### Area 1 Conceptual Site Model

The types of contamination that have been identified on Area 1 include TSM, shallow DNAPL, deep DNAPL, LNAPE, and petroleum-impacted soil. Impacted media include soil and groundwater. Shallow DNAPL has been observed in the former river channel along the top surface of the organic deposit, and appears to have collected at depressions in the deposit. Deep DNAPL has been identified in one deep monitoring well on the northern portion of Area 1, along the alignment of the historic river channel. DNAPL has not been detected in this well during recent measurements. LNAPL was observed once in 1995 in one monitoring well in the historic river channel on the northern portion of Area 1; however LNAPL has not been observed during subsequent monitoring at this location. TSM and petroleum-impacted soils have been identified on Area 1, and are intermixed in some areas. Elevated PAH, TPH, VOC compounds, and cyanide have also been detected in soil. Based on sampling results in previous investigations, TSM is assumed to contain concentrations of PAHs or VOCs that exceed UCLs, and one soil sample collected from within the Parcel E Tank Farm Area exceeded UCLs for TPH. Observed groundwater impacts include VOCs, PAHs, TPH and cyanide, and appear to be limited in severity and extent. Based on groundwater elevation measurements, the crushed stone layer beneath the culverts drains groundwater in the vicinity of the culverts.

Summary of Risk Characterization Results Relevant to Area 1

The Risk Characterization, as it relates to Area 1, satisfied a condition of No Significant Risk to Human Health under current conditions. A condition of



29

## E. Area 5, Parcel B

Area 5 is the 129 Commercial Street Property, bounded by Commercial Street to the east, Charles Street to the north, the MBTA rail tracks to the west, and Adams Street to the south. Area 5 is approximately 2 acres in size and consists of an approximately 42,000 sf building, parking areas, and landscaped space along Commercial and Adams Streets. The 129 Commercial Street building is currently occupied by a commercial bakery with active operations and limited unused, available space inside or outside of the facility. Facility operations include regular manufacturing 6 days per week, with 2 shifts per day, with additional production during high demand periods, and large freezing facilities. Outside the facility are located employee parking areas and a large turn-around space to accommodate delivery trucks.

## 1. Area 5 Site Characteristics

Historically, Area 5 was the location of MGP gas purification operations. It is possible that historic MGP features, such as drip pots or a drip tank, are located beneath the building. However, such facilities have not been identified by the 10 borings conducted to date beneath the building. Further exploration beneath the building is precluded due to the presence of several large, immovable features located inside the facility. These features include baking ovens, a storage freezer and a blast freezer, and occupy approximately ¼ of the facility space.

Impacts in this Area consist of elevated concentrations of VOCs, specifically BTEXSN compounds and cyanide in soil and groundwater beneath the building. Elevated concentrations of some VOCs have been detected in indoor air inside the building. A RAM remains ongoing to reduce concentrations of VOCs in indoor air at the 129 Commercial Street property. Actions undertaken as part of the RAM have included the installation of a pilot-scale sub-slab ventilation system along a portion of the eastern side of the building, and the sealing of parts of the floor to prevent vapor migration through the floor slab. The results of periodic indoor air sampling conducted during the RAM have indicated a decrease in VOC concentrations in indoor air, resulting in a condition of No Significant Risk with the system operating.

Area 5 Conceptual Site Model

The contamination observed at the property consists of elevated BTEXSN compounds, which have been observed in soil, groundwater, and indoor air, and cyanide in soil and groundwater. Cyanide concentrations are highest at the northeastern corner of the property, beneath the parking lot north of the building. The highest BTEXSN concentrations have been observed in soil and groundwater beneath the central portion of the building. According to the current conceptual model for Area 5, dissolution of contaminants occurs as groundwater flows through contaminated soil that exists beneath the building. Subsurface investigations to date have encountered moderately contaminated



State State State State

35

future exposure pathways at the Site; however, the Risk Characterization did satisfy a condition of No Substantial Hazard. The selection and implementation of a remedial action alternative must result in conditions that will satisfy the MCP requirements for a Response Action Outcome (RAO), namely:

- For a Class A RAO, all sources of contamination that are resulting in or likely to result in increased concentrations in environmental media are eliminated or controlled, and all exposure pathways for which a condition of No Significant Risk could not be satisfied in the Site Risk Characterization are mitigated or eliminated.
- For a Class B RAO, no remedial actions are necessary at a Site because results of assessment actions have determined that a level of No Significant Risk exists.
- For a Class C RAO, all Substantial Hazards are eliminated.

Achievement of a Class A RAO would require removal or control of sources of contamination that result in future exposure pathways for which a condition of No Significant Risk could not be satisfied. With the exception of the Governor House portion of the Site, a Class B RAO is not appropriate for this Site, because the Risk Characterization was unable to satisfy a condition of No Significant Risk to future receptors at the Site. In the event that a Class A or B RAO cannot be achieved due to Site conditions, a Temporary Solution, or Class C RAO, is a potential response action outcome given that a condition of No Substantial Hazard was satisfied for the Terrestrial portion of the Site.

The specific requirements of Class A, B and C RAOs for the terrestrial portion of the Site are discussed in the following sections. The requirements outlined in this section will be used during evaluation of remedial alternatives at the Site, and for the evaluation of the practicality and feasibility of achieving a Permanent or Temporary Solution for the Site.

### 3.02 Remedial Objectives to Achieve a Permanent Solution

Achievement of a Permanent Solution at the Site requires elimination of exposures that result in potential current and future human health and environmental risk, and elimination of UCL exceedences in Site media. Based on the potential risks identified in the Risk Characterization (presented in Section 2.06 above) and distribution of UCL exceedences, the following remedial objectives must be met to achieve a Permanent Solution.

#### A. Elimination of Potential Exposure to Contaminated Soil

A Permanent Solution for the Site would require elimination of potential exposure to soil by the following receptors through soil remediation, placement of a cap, or AULs, as appropriate. A summary of soil exposure pathways requiring remediation or raitigation in order to achieve a Permanent Solution is provided below in Table X.



#### TABLE X SUMMARY OF SOIL EXPOSURE PATHWAYS REQUIRING REMEDIATION

Receptor	Exposure Pathway	Types of Contamination	Affected Properties	
Future commercial/industrial site worker	Ingestion of soil	TSM; Shallow DNAPL	100 Commercial Street 99-109 Commercial Street 51, 65, 77, 89 Commercial Street	
Future commercial/industrial site worker	Ingestion of soil	LNAPL	99-109 Commercial Street 100 Commercial Street	
Future commercial/industrial site worker	Ingestion of soil	Petroleum- impacted soil	100 Commercial Street	
Future construction worker	Inhalation of VOCs volatilized from soil	TSM; Shallow DNAPL	100 Commercial Street 99-109 Commercial Street 51, 65, 77, 89 Commercial Street	
Future construction worker	Inhalation of VOCs volatilized from soil	LNAPL	99-109 Commercial Street 100 Commercial Street	
Future construction worker	Inhalation of VOCs volatilized from soil	Petroleum- impacted soil	100 Commercial Street	

## B. Elimination of Potential Exposure to Indoor Air

Elevated VOC concentrations were detected in indoor air at 129 Commercial Street during Phase II investigations. A Release Abatement Measure was initiated in response to these elevated concentrations, and a pilot-scale sub-slab venting system was installed beneath the floor of the 129 Commercial Street building to reduce VOC concentrations in indoor air. The data evaluated in the Human Health Risk Characterization were collected as part of the RAM, to monitor the effectiveness of the system. Therefore, the indoor air data used in the Risk Characterization were collected while the system was running, and may be lower than those that would be observed in the absence of the system. In order to achieve a Permanent Solution, this Phase III RAP will include an evaluation of remedial alternatives for the long-term mitigation of the indoor air inhalation pathway. Based on the current conceptual Site model, elevated VOC concentrations in indoor air at the 129 Commercial Street building are likely due to volatilization of BTEXSN compounds from groundwater or soil beneath the building.

## C.

#### Elimination of soil UCL Exceedences.

In order to achieve a Permanent Solution, UCL soil exceedences must be eliminated through remediation or installation of an engineered barrier. Based on the results of the Phase II, TSM contains concentrations of PAHs and/or VOCs that exceed UCLs. The limits of the observed extent of TSM, which include a significant portion of

i ape



39

#### 3.04 Remedial Objectives for Area 1

Area 1 is the northern portion of Parcel E, the 100 Commercial Street Property. As described above, Area 1 contains impacts from TSM, shallow DNAPL, deep DNAPL, LNAPL, and petroleum-impacted soil. However, deep DNAPL and LNAPL have not been observed in monitoring wells on Area 1 during recent measurements. Therefore, LNAPL and deep DNAPL monitoring will be conducted, and in the event that they are detected consistently on the Area, remedial systems will be installed for their removal. Impacted media requiring remediation include DNAPL and soil. The estimated volume of shallow DNAPL on Area 1 is approximately 3,200 gallons, the estimated areal extent of TSM is approximately 1,900 square yards (sy), and the estimated areal extent of petroleum-impacted soils is approximately 3,200 sy. Remedial objectives were developed to mitigate risk related to current use of Area 1. Remedial objectives are listed below:

#### NAPL:

Reduce thickness of DNAPL to less than 1/2 in.

Soil:

- Reduce soil contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers, and construction workers.

#### 3.05 **Remedial Objectives for Area 2**

Area 2 is the southern portion of Parcel E, the 100 Commercial Street Property. As described above, Area 2 con ains impacts from TSM, shallow DNAPL, deep DNAPL and LNAPL. The estimated volume of NAPL in the subsurface includes approximately 11,500 gallons of shallow DNAPL and 3,800 gallons of LNAPL. The areal extent of TSM-impacted soils is estimated to be approximately 10,800 sy, and the areal extent of LNAPL-impacted soils is estimated to be approximately 400 sy. Remedial objectives are listed below:

#### NAPL:

Reduce thickness of DNAPL and LNAPL in monitoring wells to less than 1/2 in.

Soil:

- Reduce soi contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers, and construction workers.

Other:

Allow for continued operation of the on-site business.

#### 3.06 **Remedial Objectives for Area 3**

Area 3 is the northern portion of Parcel A, including the 55 through 89 Commercial Street properties. As described above, Area 3 contains impacts from TSM and shallow DNAPL. The volume of shallow DNAPL in the subsurface on Area 3 is estimated to be approximately

3,700 gallons, and the areal extent of TSM-impacted soils is estimated as 8,400 sy. Remedial objectives are listed below:

### NAPL:

Reduce thickness of DNAPL in monitoring wells to less than ½ in.

Soil:

- Reduce scil contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers and construction workers.

Other:

Allow for continued operation of the on-site businesses.

#### 3.07 Remedial Objectives for Area 4

Area 4 is the southern portion of Parcel A, including the 99-105 and 105-109 Commercial Street buildings. As described above, Area 4 contains impacts from TSM and LNAPL. The estimated volume of LNAPL on Area 4 is approximately 5,600 gallons of LNAPL, and the areal extent of TSM and LNAPL – impacted soils is approximately 560 sy and 640 sy, respectively. Remedial objectives are listed below:

NAPL:

- Reduce thickness of LNAPL in monitoring wells to less than 1/2 in.

Soil:

- Reduce soil contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers and construction workers.

## Other:

Allow for continued operation of the on-site businesses.

## 3.08 Remedial Objectives for Area 5

Area 5 consists of the 129 Commercial Street Property. As described above, contamination on Area 5 requiring remediation consists of BTEXSN in soil, groundwater and indoor air. Remedial objectives are listed below:

Soil:

 Reduce soil contamination such that groundwater concentrations are below MCP S-2/GW-2 standards.

#### Groundwater:

 Reduce VOC concentrations to below GW-2 to mitigate the source of indoor air contamination.



## Indoor Air: Red

Reduce concentrations of VOCs in indoor air that exceed MADEP risk thresholds.

# Other:

Allow the on-site business to continue to operate with as little disruption as possible during implementation of remedial action alternatives.

## IV. REMEDIAL ACTION ALTERNATIVE EVALUATION PROCESS

As described in Section 2.06, in order to facilitate evaluation of remedial alternatives for the Site, the Site has been divided into five Remedial Action Alternative Areas based on property boundaries, characteristics, and type of contamination present. Sections V through X present a summary of each Area's conditions and remedial objectives, and the development and evaluation of Remedial Action Alternatives for each of the five Areas. Sections V through IX include an Initial Screening of potentially applicable remedial technologies, development of Remedial Action Alternatives, and a Detailed Evaluation of Remedial Action Alternatives. Because the format of Sections V through IX for the five Areas is similar, a brief description of the individual components and definitions of terms used in the evaluation process is provided below.

## 4.01 Initial Screening Process

#### A. Initial Screening

An initial screening of remedial technologies is conducted for each Area to identify technologies that are reasonably likely to be feasible, based on contaminants present in the Area, contaminated media, and specific Area characteristics. Per 310 CMR 40.0856, remedial technologies are reasonably likely to be feasible if they are reasonably likely to achieve a Permanent or Temporary Solution, and if individuals with the necessary expertise to implement the technology are available.

Remedial technologies were screened for the contaminated media that contribute to exposure pathways for which a condition of No Significant Risk could not be satisfied for current or future receptors on properties located within each Area. The results of the initial screening are presented in a table, which includes a brief description of the technology and the rationale for retaining or eliminating the technology. A brief list of the remedial technologies that meet the initial screening criteria is provided in the Initial Screening text of the report.

## B. Elimination of Remedial Technologies Based on Site Characteristics

As discussed above, the Initial Screening criteria are a basic set of criteria intended to identify remedial technologies that may be applicable based on the general conditions at the Site. However, the specific characteristics of each Area (as listed above in Section 2.06) limit the feasibility of some remedial technologies. Therefore, those technologies that are not likely to be feasible for a specific Area are eliminated before Remedial Action Alternatives are developed. Remedial technologies that are eliminated based on Site characteristics are listed in this Section, either in bullet or tabular form, along with the rationale for elimination.



#### 4.02 Development of Remedial Action Alternatives

A Remedial Action Alternative, as used in this Phase III RAP, is defined as a combination of remedial technologies that is reasonably likely to meet the remedial objectives for the Area. Remedial Action Alternatives include components that remove, remediate, or mitigate exposure to contaminated media that result in exposure pathways for which a condition of No Significant Risk could not be satisfied for the Area. Remedial Action Alternatives are developed from the technologies that met the initial screening criteria and were not eliminated based on characteristics of the individual Area above.

### 4.03 Detailed Evaluation Process

In this Section, Remedial Action Alternatives that were developed for the Area are compared based on their potential to achieve the Area remedial objectives, and to meet the specific Area characteristics listed in Section 2.06. The Remedial Action Alternatives are compared using the eight criteria that are specified in the MCP, which include effectiveness, short and long term reliability, implementability, capital and long-term costs, risks, benefits, timeliness, and effect on non-pecuniary interests. These criteria are defined below. The detailed evaluation, which is summarized in a table for each Site Area, results in the selection of a Remedial Action Alternative for the Area, which is judged to be the most appropriate for the Site, based on known conditions in the Area and the evaluation criteria listed below.

## A. Evaluation Criteria

The alternatives were evaluated with respect to the following eight criteria as defined in 310 CMR 40.0858:

Effectiveness: This criterion addresses a remedial alternative's ability to achieve a Permanent or Temporary Solution; its effectiveness in reusing, recycling, destroying, detoxifying, or treating the hazardous materials; and its effectiveness in reducing levels of untreated hazardous material to concentrations that achieve or approach background.

Short-term and Long-term Reliability: This criterion is used to assess the degree of certainty that an alternative will be successful and the effectiveness of measures required to manage residues, remaining wastes, and control discharges to the environment. This criterion includes assessment of both the probability of success of risk reduction and/or contaminant mass reduction in the near term, as well as the probability that the alternative will be successful in achieving the long-term remedial goals.

Implementability: The Implementability criterion addresses the technical complexity of the alternative; integration with existing facility operations; integration with other remedial operations; maintenance and monitoring requirements; Site access requirements; availability of necessary services and materials; availability of off-site treatment, storage and disposal facilities; and whether the alternative meets regulatory requirements for permits required by local, state or federal agencies.



in.

**Cost:** Costs considered include design, construction, equipment, Site preparation, labor, permits, disposal, operation, energy requirements, maintenance, monitoring, environmental restoration and damages to natural resources. Costs are provided for comparison and feasibility purposes only, and are considered accurate to approximately +50/-30 percent.

The Net Present Value (NPV) is presented for alternatives that include ongoing activities, such as operation, maintenance and/or monitoring. An annual discount rate of 6 percent was used based on long-term typical United States public debt instruments. NPV estimates are provided for comparison only, and are considered accurate to approximately +50/-30 percent.

Risks: Short-term on-site and off-site risks during implementation are considered under this criterion. The Risk criterion also addresses on-site and off-site risks posed over the period of time required for the alternative to attain the remedial goal, and the long-term potential risk posed by hazardous material remaining at the Site after the completion of remedial activities.

Benefits: This criterion addresses the benefit of restoring natural resources; providing for productive Site reuse; avoided costs of relocating people; and avoided lost value of the Site.

Timeliness: The timeliness of eliminating uncontrolled sources or achieving a level of No Significant Risk or No Substantial Hazard is evaluated under this criterion.

Effect on Non-pecuniary Interests: This criterion addresses the effect of the alternative on non-monetary aspects, such as aesthetic values. It also includes impact of the installation of remedial measures on the use of properties not owned by MEC.

## B. Weighting of the Evaluation Criteria

In accordance with 40.0861(2)(b), the approach to the evaluation of Remedial Action Alternatives and the weighting of the criteria listed above is provided herein. The weighting of the Detailed Evaluation criteria varies slightly by Area; however the following general approach is taken in evaluation of the Remedial Alternatives according to the criteria listed above. The criterion that is given the greatest weight in this Phase III is implementability. Properties on the Site have features such as buildings and operating commercial businesses that limit the implementability of some remedial measures. The next tier of criteria includes effectiveness, cost and nonpecuniary interests. Remedial Alternatives that would result in a cost-effective Permanent Solution for the Site are considered more desirable than alternatives that may not result in a Permanent Solution, or may represent a higher cost burden to achieve a similar result (i.e., a Permanent or Temporary Solution). The nonpecuniary interest criterion is used in this evaluation to consider the potential for negative impacts on the businesses operating on the properties located within the Site limits, and the effect that the implementation of the remedy may have on the surrounding area. These potential impacts are important because the Site is located in



an urban area near Malden Center, and disruption to the area could have far-reaching financial implications to the businesses in the area and the community of Malden itself. The reliability, benefits and risk criteria are also given due consideration, as these are important factors in the selection of a remedial solution that will meet the remedial goals. Because a condition of No Substantial Hazard exists for the terrestrial portion of the Site, and a condition of No Significant Risk to Human Health and the Environment exists for current site conditions, the Timeliness criterion is given the least weight in this evaluation.

## 4.04 Selected Remedial Action Alternatives for the Site

Selected Remedial Action Alternatives and conceptual details regarding their implementation (such as time requirements and order of implementation) are described at the conclusion of the Detailed Evaluations for each Site Area. A summary of the selected overall Remedy for the Site, which will be a combination of the selected remedies for each Site Area, is then presented at the conclusion of this Phase III RAP, in Section X.

47

## V. EVALUATION OF REMEDIAL ALTERNATIVES FOR AREA 1

#### 5.01 Introduction

In this section, Remedial Action Alternatives to address contamination identified in Area 1 are developed and evaluated. A brief summary of the existing conditions and remedial objectives for Area 1 is provided below. Based on these conditions and objectives, an initial screening of remedial technologies is presented in Section 5.04 to identify technologies that may be reasonably likely to be feasible to address contamination identified on Area 1. Remedial technologies that meet the initial screening criteria but are not applicable on Area 1 due to specific site characteristics are eliminated from further consideration in Section 5.04 E. Remedial Action Alternatives are then developed in Section 5.05 by combining the remaining remedial technologies based on knowledge of Site conditions and engineering judgement. A detailed evaluation of the Remedial Action Alternatives is then conducted in Section 5.06, and the recoramended Remedial Alternative is presented in Section 5.07.

#### 5.02 Summary of Area 1 Conditions

Area 1 is the northern portion of the 100 Commercial Street property, which is currently used as part of the KeySpan Energy Delivery Operations Center. Area 1 is bounded by Centre and Commercial Streets and the MR and WEB culverts, and includes the portion of the MR Culvert that abuts it. Types of contamination that have been identified in Area 1 include TSM, shallow DNAPL, deep DNAPL, LNAPL and petroleum-impacted soil. The estimated volume of shallow DNAPL on Area 1 is approximately 3,200 gallons, the estimated areal extent of TSM is approximately 1,900 square yards (sy), and the estimated areal extent of petroleum-impacted soils is approximately 3,200 sy. As described in Section 2.06, deep DNAPL and LNAPL have not been identified in Area 1 monitoring wells during recent monitoring events.

#### 5.03 Summary of Area 1 Remedial Objectives

Remedial objectives were developed to mitigate risk related to future commercial use of Area 1. Remedial objectives are listed below:

#### NAPL:

Reduce thickness of DNAPL to less than 1/2 in.

#### Soil:

Reduce soil contaminant concentrations to less than the UCLs.

Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers, and construction workers.

Because LNAPL and deep DNAPL have not been detected during recent monitoring events, remedial measures for their removal from the subsurface will not be evaluated. Monitoring of wells in which LNAPL and deep DNAPL have been detected will be conducted during



implementation of the remedy. In the event that LNAPL or deep DNAPL are detected consistently in a monitoring well, removal methods will be evaluated.

## 5.04 Initial Screening of Remedial Technologies for Area 1

An initial screening of remedial technologies was conducted to identify technologies that are reasonably likely to be feasible at the Site, based on Site contaminants, contaminated media, and specific Site characteristics. Per 310 CMR 40.0856, remedial technologies are reasonably likely to be feasible if they are reasonably likely to achieve a Permanent or Temporary Solution, and if individuals with the necessary expertise to implement the technology are available.

Remedial technologies were screened for the contamination that contributes to exposure pathways for which a condition of No Significant Risk could not be satisfied for foreseeable future receptors on Area 1. A summary of the initial screening for Area 1 is presented in Table XI, including a brief description of the remedial technologies, and a rationale for retaining or eliminating the technology. Remedial technologies that meet the initial screening criteria are listed and briefly described below.

## A. Shallow DNAPL

- II DNAPL extraction using recovery wells: Installation of large (8-in.) diameter recovery wells, equipped with a submersible pump to extract DNAPL from the subsurface. Extracted DNAPL would be pumped from wells to small buildings on Site, in which it would be stored until it is transported off-site for disposal. A DNAPL extraction well has been operating on the southern portion of the 100 Commercial Street property (Area 2) since October 2001, and has collected approximately 670 gallons of DNAPL to date.
- II DNAPL extraction/migration control using trenches: Excavation of trenches that are keyed into the low-permeability organic deposit, and backfilled with gravel to collect DNAPL. A recovery well would be installed in the backfilled trench, equipped with a submersible pump to extract and collect DNAPL. Trenches could used to control shallow DNAPL migration from one portion of the Site to another through placement along property or area boundaries. Extracted DNAPL would be stored in small buildings on Site, and transported off-site for disposal or incineration.
- Vertical subsurface barrier for DNAPL migration control: Installation of a vertical barrier to DNAPL flow. Need depends on sequencing of DNAPL remediation on adjacent properties or areas. The type of barrier would depend on Phase IV design; potential options include a slurry wall, sheet piling, or a Waterloo<sup>TM</sup> Barrier.
- Disposal or incineration: Off-site transport of extracted NAPL to an appropriate receiving facility for disposal or incineration.



- B. Soil
  - Excavation and off-site treatment: Excavate and remove contaminated soil to an average depth of approximately 12 ft, along with DNAPL and LNAPL where present, and transport impacted soils off-site for treatment. Suitable excavated soils may be re-used as backfill on site. Implementation of this action would require support of excavations and building foundations during excavation, and may require demolition of on-site buildings if contamination is located beneath buildings. Implementation would also require an enclosure for control of vapors and odors.
  - Installation of an Engineered Barrier: RCRA-type Engineered Barrier consistent with 310 CMR 40.0000, to be installed over soil containing contaminant concentrations in excess of the UCL. Installation of an Engineered Barrier would require excavation of approximately 3 ft of soil, followed by placement of an HDPE impermeable membrane, which would be covered by approximately 1 ft of clean sand and a geo-composite drainage layer. The drainage layer would be covered by a marker fabric, and either pavement or a vegetated surface. Implementation of this alternative would require demolition of buildings, if present in areas of contamination.
  - Installation/Maintenance of a Direct Contact Barrier: Installation and/or maintenance of a surface barrier (i.e., 3 ft of soil, asphalt or concrete pavement), to prevent direct contact with surface soils.
  - Incineration: Off-site transport of excavated soil to an appropriate receiving facility for incineration
  - Ex-situ thermal descrption: Treatment of excavated soil material using thermal desorption with an afterburner. May be performed on or off-site; however on-site thermal desorption is not likely implementable due to the urban location and active use of the Area.
  - In-situ chemical oxidation: Injection of hydrogen peroxide, Fenton's Reagent, or other oxidant into soil using temporary injection wells to chemically break down contaminants to innocuous compounds. This remedial method is not feasible when a significant amount of free-phase product (i.e., LNAPL or DNAPL) exists in the subsurface, as uncontrollable reactions can result.
  - Activity and Use Limitations (AULs): Place restrictions on future property use and activities to prevent potential exposure through a pathway for which a condition of No Significant Risk could not be satisfied.



## C. Elimination of Remedial Technologies Based on Site Characteristics

Remedial technologies passed the initial screening portion of the evaluation process if the technology is reasonably likely to achieve a Permanent or Temporary Solution, and if experts are available to implement the technology. However, some technologies are not considered feasible in Area 1 due to specific site characteristics in Area 1. Therefore, the remedial technologies listed in Table XII below are not considered reasonably feasible in Area 1 and will not be incorporated into Remedial Action Alternatives for Area 1.

#### TABLE XII

REMEDIAL TECHNOLOGIES ELIMINATED BASED ON SITE CHARACTERISTICS AREA 1

Remedial Techriology	Technology Type	Target Media	Reason For Elimination
Extraction using trenches	Extraction	Shallow DNAPL	Significantly more disruptive and difficult to implement than extraction wells, which have been demonstrated to be effective on Area 2 for extraction of DNAPL.
DNAPL rnigration control using trenches	DNAPL migration control	Shallow DNAPL	Significantly more disruptive and difficult to implement than vertical subsurface barriers, which would likely be effective for DNAPL migration control.

#### 5.05 Development of Remedial Action Alternatives for Area 1

Remedial Action Alternatives for Area 1 are presented below. The Remedial Action Alternatives were developed using engineering judgement from the remedial technologies that passed the initial screening and were not eliminated based on specific Site characteristics. Approximate locations of the components of the Remedial Alternatives for Area 1 are shown on Figure 16.

 Alternative 1-1: Excavation of TSM, and shallow DNAPL, and petroleum-impacted soil, provisional shallow DNAPL migration control, and an AUL.

This alternative involves the excavation of the TSM and shallow DNAPL contamination on the northern portion of Area 1 and excavation of the petroleumimpacted soil on the southern portion of Area 1. Excavation of the petroleumimpacted soil would also require demolition of the KeySpan maintenance garage. Approximate locations of proposed remedial components for Area 1 are shown on Figure 16. In addition to the removal of contaminated soil and DNAPL, excavation of TSM located within the Area is intended to remove the source of shallow DNAPL in the Area. If needed, DNAPL migration onto Area 1 would be prevented using a vertical subsurface barrier, such as sheet piling or a Waterloo<sup>TM</sup> Barrier. It is anticipated that DNAPL migration control may be necessary in the event that the



estimated DNAPL volume of approximately 3,200 gallons beneath Area 1, and a net extraction rate of 35 gallons per month for the 5 extraction wells, it is estimated that approximately 5 to 7 years will be required to reach the DNAPL remedial goals on Area 1. Therefore, the estimated total time requirement to reach a Permanent Solution in Alternative 1-3 is approximately 6 to 9 years. However, the long-term rate of DNAPL recovery is difficult to estimate and may decrease with time. A vertical subsurface barrier would be installed to prevent the migration of shallow DNAPL from adjacent areas onto Area 1, if necessary. An AUL would be required to maintain the Engineered Barrier.

Implementation of this alternative would involve closing off the majority of the Area during excavation and installation of the Engineered Barrier, and therefore would result in significant disruption to facility operations. Additionally, an estimated 240 truckloads would be required to transport the excavated soil off-site. Use of the Area by the owner would likely be limited or completely discontinued during implementation of the remedy.

 Alternative 1-4: Shallow DNAPL recovery, provisional shallow DNAPL migration control, in-situ chemical oxidation of TSM area and petroleum-impacted soil, and an AUL

Alternative 1-4 would involve the installation of five shallow DNAPL recovery wells (as above in Alternative 1-3), and monitoring of deep monitoring wells for DNAPL and shallow wells for LNAPL. A small storage building would be required on Area 1 to store recovered DNAPL prior to off-site disposal. If necessary, the migration of shallow DNAPL from adjacent areas onto Area 1 would be prevented using a vertical subsurface barrier, and TSM and petroleum-impacted soil would be treated through injection of oxidants to promote in-situ chemical oxidation reactions. Chemical oxidation treatment of impacted soil in this alternative would be conducted as the treatment becomes feasible. In the case of TSM soils, chemical oxidation of soil would not be feasible until DNAPL has been reduced to the extent practicable. In the case of petroleum-impacted soils, LNAPL and DNAPL have not been identified; therefore the feasibility of the treatment of soil is dependent on the current owner's operational needs. As described above, it is estimated that approximately 5 to 7 years will be required to reach the DNAPL remedial goals. For the purposes of estimation, Alternative 1-4 includes implementation of chemical oxidation as one integrated program for the TSM and petroleum-impacted soils once the DNAPL remedial goals have been met. The chemical oxidation program is assumed to involve the installation of approximately 150 temporary injection wells (assuming 20-ft spacing between the wells), and three oxidant injections. Assuming chemical oxidation requires approximately 1 to 2 years to implement, the time requirement to reach a Permanent Solution in Alternative 1-4 is estimated to be approximately 6 to 9 years from the date of implementation of DNAPL extraction. AULs may be required to mitigate risk through future soil exposure pathways.

Disruption to facility operations during DNAPL extraction would involve temporary disruption during installation of extraction wells and a storage building, plus regular



(weekly or monthly) maintenance activities performed on wells and at the storage building and transport of extracted DNAPL off-site. Disruption to Area operations during chemical oxidation would involve limitations on use of the Area for three periods of approximately one month during installation of temporary injection wells and injection of oxidant.

 Alternative 1-5: Shallow DNAPL recovery, Monitoring of LNAPL and deep DNAPL, provisional shallow DNAPL migration control and AUL

Alternative 1-5 is a less intrusive remedial option than Alternatives 1-1 through 1-4 to address the presence of MGP contamination on Area 1. This alternative involves the installation of five shallow DNAPL recovery wells, monitoring of deep monitoring wells for the presence of DNAPL, and monitoring for the presence of LNAPL in monitoring wells. If shallow DNAPL migration control is required, vertical subsurface barriers may be installed to prevent migration of DNAPL onto Area 1 from adjacent properties. As described above, it is estimated that five DNAPL extraction wells would accomplish the DNAPL remedial goals in approximately 5 to 7 years. However, because this alternative does not include remediation of soil, a Permanent Solution is not achieved in Alternative 1-5. An AUL will be required to mitigate risk through future soil exposure pathways.

Disruption to facility operations during DNAPL extraction would involve temporary disruption during installation of extraction wells and a storage building, plus regular (weekly or monthly) maintenance activities performed on wells and at the storage building and transport of extracted DNAPL off-site.

Alternative 1-6: NAPL Monitoring and AUL

This alternative involves continued monitoring of LNAPL and DNAPL to ensure that NAPL located in the subsurface on Area 1 does not migrate off-site. Because this alternative does not involve removal of NAPL from the subsurface, it would not require a subsurface barrier to shallow DNAPL migration. This alternative involves minimal disruption to Area use during implementation of the remedy.

#### 5.06 Detailed Evaluation of Remedial Action Alternatives for Area 1

This section presents a Detailed Evaluation of the Remedial Action Alternatives presented above, based on the criteria listed in Section 4.03. As with the other Areas, this evaluation is based on the current property conditions and use, as described in Section 2.06 and summarized in Section 5.02. A summary of the detailed evaluation of the six remedial action alternatives developed for Area 1 is presented in Table XIII. Table XIII presents a summary of the likely effectiveness, reliability, implementability, cost (net present value, or NPV), risks, benefits, timeliness, and potential effect on non- pecuniary interests of each alternative. A comparison of the six remedial alternatives based on these criteria is presented below.

As indicated in Table XIII, each of the six remedial action alternatives (with the exception of Alternative 1-1) provisional shallow DNAPL migration control and an AUL. Achievement of



a Permanent Solution on Area 1 will require accomplishment of the NAPL reduction remedial goals. On Area 1, this goal is expected to be reasonably attainable with respect to deep DNAPL and LNAPL, as these contaminants have not been observed on Area 1 during recent monitoring events. Provisional shallow DNAPL migration control is intended to be a method of preventing the migration of shallow DNAPL from adjacent areas onto Area 1 once remediation on Area 1 has been completed. Because these remedial components are common to nearly all the remedial action alternatives proposed for Area 1, they do not have a significant impact on the comparison of the alternatives and are not included in the evaluation presented below. An AUL may be used to mandate specific land use requirements, such as a requirement that commercial/industrial use of the property be maintained, eliminating residential exposure pathways from consideration.

- Effectiveness: As shown on Figure 16, two distinct areas of contamination exist within Area 1: the contamination associated with TSM on the northern portion of the Area, and the petroleum-impacted soils on the southern portion of Area 1. Because they involve excavation of the contamination associated with TSM, Alternatives 1-1 and 1-2 are considered the most effective of the proposed alternatives. Alternative 1-3 (Engineered Barrier installation) is less effective than Alternatives 1-1 and 1-2, because although Alternative 1-3 eliminates the potential for exposure to contaminated soil through the installation of an Engineered Barrier, DNAPL extraction is likely to be a less effective means of DNAPL removal than excavation. Alternative 1-4 would likely have a similar level of effectiveness in the long term as Alternative 1-3, as both of these alternatives rely on DNAPL extraction wells to remove shallow DNAPL, and both would eliminate potential for exposure to soil containing UCL exceedences. A difference in effectiveness between Alternative 1-3 and 1-4 is that the Engineered Barrier proposed in Alternative 1-3 provides more rapid elimination of UCLs in the short term, however in-situ chemical oxidation proposed in Alternative 1-4 would likely result in a reduction in contaminant concentrations, a more effective solution in the long term. Alternative 1-5 is less effective than Alternatives 1-1, 1-2, 1-3, and 1-4, because it would not include measures (i.e., excavation or chemical oxidation) to reduce concentrations of contaminants in soil to less than the UCLs. Alternative 1-6, a "monitoring only" alternative, is the least effective of the six that were evaluated, because it would not include shallow DNAPL extraction or measures to reduce concentrations of contaminants to less than the UCLs. Alternatives 1-1, 1-2, 1-3, and 1-4 are likely Permanent Solutions for Area 1, although 1-3 and 1-4 would require a significantly longer time to reach a Permanent Solution than Alternatives 1-1 and 1-2 (refer to timeliness below). Alternatives 1-5 and 1-6 represent Temporary Solutions.
- Short and Long-term Reliability: In general, the alternatives that involve excavation of the contamination associated with TSM on the northern portion of Area 1 (i.e., Alternatives 1-1 and 1-2) are considered the most reliable remedial alternatives. Alternative 1-1 is considered the most reliable alternative, both in the short and long term, of the six evaluated alternatives because it involves excavation of both the contamination associated with TSM and the petroleum-impacted soil. Alternative 1-2, which involves excavation of the TSM contamination and in-situ chemical oxidation of petroleum-impacted soil, is also highly reliable; chemical oxidation has been used at



other MGP and petroleum contamination sites to reduce concentrations of PAHs and VOCs. Alternative 1-3 is ranked as moderately reliable, less reliable than Alternatives 1-1 and 1-2 in the short and long term. Alternative 1-3 provides relatively reliable elimination of the exposure pathway to UCLs in soil through installation of an Engineered Barrier. However, DNAPL extraction wells are a less certain method of DNAPL removal compared to excavation, and are likely to require a longer time period to accomplish the DNAPL remedial goals than excavation. Eased on the successful operation of a DNAPL extraction well on Area 2, it is likely that DNAPL extraction wells placed in proper locations on Area 1 will effectively remove DNAPL from the subsurface. However, it is less certain that the DNAPL remedial goals will be reached using DNAPL extraction wells than would be the case if the contamination were excavated. Alternative 1-4, which also relies upon DNAPL extraction wells to remove DNAPL and uses in-situ chemical oxidation to reduce concentrations of contaminants in soil, is also ranked as moderately reliable in the short and long term. Alternative 1-4 is considered more reliable than Alternative 1-3 in the long term, because the in-situ chemical oxidation component of Alternative 1-4 would likely result in a reduction in contaminant concentrations. Alternative 1-5 is considered moderately reliable in the short term, as DNAPL extraction wells are likely to function reliably once installed. However, over the long term Alternative 1-5 is ranked as a low reliability alternative, because it does not include a method to reduce contaminant concentrations in soil, which would be required for a Permanent Solution. Reliability of Alternative 1-6 is low, as remedial systems are not installed in this Alternative.

Implementability: Implementability is an important criterion in the selection of a remedial alternative for Area 1. Alternatives 1-1 and 1-2 would be difficult to implement, due to the large volume of highly impacted soil that would be excavated and transported off-site in these Alternatives. Nuisance odors and high truck traffic would be very disruptive to the surrounding area of Malden Center, which has a relatively high volume of public pedestrian and vehicular traffic. A temporary enclosure would likely be required to contain potentially hazardous vapors, and workers inside the enclosure would require Level B respiratory protection (i.e., supplied air). Excavation support, dewatering, and water treatment would also be required. The KeySpan maintenance garage would need to be removed in order to excavate petroleum-impacted soils in Alternative 1-1, and to install the Engineered Barrier in Alternative 1-3. Demolition of the maintenance garage would not be required in Alternatives 1-2 or 1-4, as oxidants may be delivered through the floor slab of the garage, assuming proper ventilation of sub-slab soil can be provided. Demolition of the Maintenance Garage also would not be required for Alternatives 1-5 and 1-6.

The alternatives that do not involve excavation of the TSM and shallow DNAPL are generally considered to be more implementable than Alternatives 1-1 and 1-2. Alternative 1-3 is considered moderately difficult to implement, due to the relatively large volume of shallow soil that would be excavated to install an engineered barrier. However, Alternative 1-3 would be considerably less disruptive than Alternatives 1-1



and 1-2, because shallow soils (i.e., 0 to 3 ft bgs) in Area 1 are less contaminated than deeper soils near the organic deposit. Alternative 1-3 would also not involve excavation dewatering. Alternatives 1-4 and 1-5 are considered readily implementable, because few difficulties are anticipated with the installation of DNAPL extraction wells, and chemical oxidation of TSM and petroleum-impacted soils after NAPL removal (included in Alternative 1-4) is expected to be relatively straightforward. Alternative 1-6 is considered readily implementable, as it includes only monitoring and does not call for the installation of remedial measures.

- Cost: As listed in Table XIII, Alternatives 1-1 and 1-2 are the most expensive, with estimated 30-year NPV of approximately \$4,900,000 (4.9M) and \$2.3M, respectively. The costs of Alternatives 1-1 and 1-2 are driven by the need for an enclosed structure for excavation and off-site transportation and disposal of contaminated soil and NAPL. The estimated 30-year NPV for Alternative 1-3, which involves the installation of an Engineered Barrier and shallow DNAPL extraction wells, is approximately \$2.1M. It is noteworthy that the estimated 30-year NPV for Alternatives 1-2 and 1-3 are within 10%, indicating that installation of an engineered barrier may not be a cost-effective solution on Area 1 due to the presence of shallow DNAPL. The estimated 30-year NPV for Alternatives 1-4, 1-5, and 1-6 are approximately \$1.1M, \$540K, and \$110K, respectively. At approximately \$1.1M, Alternative 1-4 is currently the most cost-effective potential Permanent Solution for Area 1.
  - Risks: The highest level of risk for the evaluated alternatives for Area 1 is associated with Alternatives 1-1 and 1-2, which have been given a "high" risk level for this evaluation. These risks include risk of inhalation of vapors by workers and passersby during excavation of TSM and contaminated soils. Mitigation of these risks would require an enclosure and Level B respiratory protection during excavation. In Level B excavation work inside an enclosure, workers would be subject to higher risks of injury due to limited fields of vision and fatigue, as well as exhaustion and dehydration. Additional risks to workers are involved with the handling and injection of chemical oxidants in Alternative 1-2; these risks include hazards associated with handling caustic reagents, and the potential for vapor, pressure and heat buildup beneath pavement and structures during in-situ chemical oxidation. Risk associated with the installation of an Engineered Barrier in Alternative 1-3 is considered to be relatively low, because although this alternative involves excavation of shallow soils, shallow soils on Area 1 are less impacted than deeper soils near the upper surface of the organic deposit. A moderate level of risk is associated with Alternative 1-4; risks associated with this alternative include the risks mentioned above for chemical oxidation. Risk associated with Alternative 1-5 is scored as low, as this alternative does not include excavation or chemical oxidation of TSM/soils. The lowest risk level is associated with Alternative 1-6, which is a "monitoring only" alternative, and involves very little potential for exposure to hazardous materials.

Long-term risks due to contamination left in-place are greatest in Alternative 1-6, which does not remove DNAPL or contaminated soil. These risks are less for Alternative 1-5, which removes DNAPL over the long term but leaves TSM in place.



Long-term risk for Alternatives 1-4 and 1-3 are less than those associated with Alternative 1-5 and similar to each other, as both remove DNAPL and eliminate the potential for exposure to UCLs in soil. Alternatives 1-2 and 1-1 present the lowest long-term risk, as they include removal or in-situ treatment of contamination.

- Benefits: Alternatives 1-1 and 1-2 provide the greatest benefit, as they provide the greatest reduction in contaminant mass (i.e., removal/treatment of DNAPL, LNAPL [if present] TSM, and impacted soil) on Area 1 in the shortest possible time frame. Benefits of Alternative 1-3 include elimination of the potential for exposure to TSM and contaminated soil in the short term, and removal of DNAPL, over time. Alternative 1-4 would result in removal of DNAPL and reduction of soil contaminant concentrations, both over the long term. Alternative 1-5 provides the benefit of the removal of DNAPL from Area 1, but does not reduce soil contaminant concentrations. Alternative 1-6 provides little benefit in terms of contaminant reduction.
- Timeliness: There is significant variation in the timeliness of the evaluated alternatives for Area 1. Alternatives 1-1 and 1-2 are the timeliest alternatives under consideration for Area 1. Implementation of these alternatives is expected to result in a Permanent Solution within approximately 1 to 3 and 2 to 4 years, respectively. The other alternatives are less timely. Alternative 1-3 would eliminate the potential for exposure to soil containing contaminant concentrations greater than the UCLs (through the installation of an Engineered Barrier) within 1 to 2 years; however attainment of the DNAPL remedial goals (and therefore a Permanent Solution) is anticipated to require approximately 5 to 7 years. Therefore, the time required to attain a Permanent Solution for Alternative 1-3 is approximately 6 to 9 years. The estimated time requirement to attain a Permanent Solution in Alternative 1-4 is the same as Alternative 1-3, approximately 6 to 9 years. Alternative 1-5 would require approximately 5 to 7 years to reach the DNAPL remedial goals, but does not reach a Permanent Solution because it does not include measures to reduce contaminant concentrations in soil to below UCLs. Alternative 1-6 also would not reach a Permanent Solution because it does not include measures to remove NAPL or to reduce contaminant concentrations to less than UCLs.
- Effect on Non-pecuniary Interests: Implementation of Alternatives 1-1 and 1-2 would have a significant impact on the use of Area 1 and on the surrounding area of Malden Center during excavation work. Excavation of TSM soil, because it contains high concentrations of volatile and semivolatile compounds such as benzene and naphthalene, would be conducted inside an enclosure to contain potentially harmful vapors. Despite this precaution, nuisance odors are likely during the excavation process. Additionally, transport of the anticipated volume of TSM soil (approximately 3,725 cu yd), which is included in both Alternatives 1-1 and 1-2, would require a minimum of approximately 150 truckloads (assuming approximately 25 cu yd per truck) to transport the soil to an off-site treatment facility. Transport of the petroleum-impacted soil (approximately 12,800 cu yd), excavation of which is included in Alternative 1-1, would require approximately 500 additional truckloads. Importation of ciean fill would likely require a similar number of truck trips, resulting



ir approximate total numbers of 1,300 truck trips for Alternative 1-1, and approximately 300 truck trips for Alternative 1-2. This volume of truck traffic, over a period of approximately six months to one year of excavation work, would have a significant impact on the local area. The impact of Alternative 1-3 on the local area would be less than Alternatives 1-1 and 1-2, due to the smaller anticipated excavation volume and because contaminant concentrations are generally lower in shallow soils. However, the excavation and Engineered Barrier construction work would still require a significant amount of construction traffic and would likely be very disruptive to the area. Additionally, construction of an Engineered Barrier would limit options for future development on the Area. Implementation of Alternatives 1-4 and 1-5 would be significantly less disruptive to property use and to the local area than Alternatives 1-1, 1-2 and 1-3, because Alternatives 1-4 and 1-5 do not involve excavation of contaminated soil.

### 5.07 Selected Remedial Action Alternative for Area 1

In summary, Alternatives 1-1 and 1-2, which involve excavation of the contamination related to TSM on the northern portion of the property, are likely the most effective, reliable, and timely alternatives and provide the most benefit in terms of reducing contaminant concentrations in the subsurface. However, these alternatives are also the most difficult to implement, have the greatest associated short-term risk, are the most costly to implement, and would have significant negative impacts on the community surrounding the Area during implementation. Alternative 1-3 provides effective, reliable elimination of potential exposure to UCLs at the site in a timely manner, but requires a lengthy period of time (i.e., approximately 6 to 9 years) to reach a Permanent Solution due to reliance on DNAPL extraction to reach DNAPL goals. Further, Alternative 1-3 limits future Area use options through the installation of an Engineered Barrier. Alternative 1-4 would reach a Permanent Solution in a similar timeframe as Alternative 1-3, and is easier and less costly to implement. Additionally, because Alternative 1-4 does not include excavation, it would have significantly less impact on the surrounding area than Alternative 1-3. Because Alternative 1-4 is a likely effective, reliable, and implementable Permanent Solution, it would be preferable over Alternatives 1-5 and 1-6, which are not likely Permanent Solutions, and Alternative 1-4 is the selected Remedial Action Alternative for Area 1. Alternative 1-4 involves the following components:

- Shallow DNAPL extraction using wells equipped with submersible pumps;
- Chemical oxidation of soil (when feasible),
- Control of shallow DNAPL migration onto the property (if necessary) and
- An AUL to maintain commercial/industrial land use on the property.

Approximately five 8-in. diameter, shallow DNAPL extraction wells are anticipated to extract DNAFL from the historic river channel on the northern portion of the property. Because LNAPL and deep DNAPL have not been observed recently, monitoring will be conducted to evaluate whether extraction of deep DNAPL and LNAPL is appropriate. Installation of DNAPL extraction wells can begin once when facility operations allow the installation of extraction wells and a NAPL storage facility on the property. In-situ chemical oxidation of



petroleum-impacted soil may be implemented as appropriate, given the requirements of the current property use. In-situ chemical oxidation of TSM soil, which is anticipated to require approximately 1 to 2 years to implement, cannot be conducted until such time as the thickness of DNAFL has been reduced to the extent practicable on the Site. This is anticipated to require an estimated 5 to 7 years, based on the estimated quantity of DNAPL beneath Area 1 and observed DNAPL extraction rates in RW-1, a DNAPL extraction well located on Area 2. Therefore the time requirement to attain a Permanent Solution on Area 1 is approximately 6 to 9 years. Shallow DNAPL migration control may be implemented to prevent the flow of DNAPL from adjacent properties (i.e., Area 3) onto Area 1, which may prolong the time required to achieve the remedial goals on Area 1. The projected Net Present Value of this alternative is estimated to be approximately \$1.1M.



## VI. EVALUATION OF REMEDIAL ALTERNATIVES FOR AREA 2

#### 6.01 Introduction

In this section, Remedial Action Alternatives to address contamination identified in Area 2, the southern portion of the 100 Commercial Street property, are developed and evaluated. A brief summary of the existing conditions and remedial objectives for Area 2 are provided below. Based on these conditions and objectives, an initial screening of remedial technologies is presented in Section 6.04 to identify technologies that may be applicable to address contamiration identified on Area 2. Remedial technologies that meet the initial screening criteria but are not applicable on Area 2 due to specific site characteristics are eliminated from further consideration in Section 6.04 D. Remedial Action Alternatives are then developed in Section 6.05 by combining the remaining remedial technologies based on knowledge of site conditions and engineering judgement. A detailed evaluation of the Remedial Action Alternatives is then conducted in Section 6.06, and the recommended Remedial Alternative is presented in Section 6.07.

#### 6.02 Summary of Area 2 Conditions

Area 2 is the southern portion of the 100 Commercial Street property currently owned by KeySpan, as shown in Figure 15. Area 2 is bounded by Charles and Commercial Streets, Area 1, and the MR Culvert. Area 2 includes a portion of the WEB Culvert, which transects the northern portion of Area 2, and the section of the MR Culvert that abuts it to the east. Area 2 was the location of several historic MGP features, including the historic condenser house, a relief holder, a tar filter and tar spray pond. Types of contamination that have been identified on Area 2 include TSM and shallow DNAPL, deep DNAPL, and LNAPL. The estimated volume of NAPL in the subsurface includes approximately 11,500 gallons of shallow DNAPL and 3,800 gallons of LNAPL. Based on the results of the Phase II investigations, recoverable deep DNAPL is likely present on Area 2 in the vicinity of B108-OW (refer to Figure 13). However, the volume is difficult to reliably estimate due to the sporadic nature of the detections and the geology at depth. The areal extent of TSM-impacted soils is estimated to be approximately 10,800 sy, and the areal extent of LNAPL-impacted soils is estimated to be approximately 400 sy. These estimates include DNAPL and impacted soil that likely exists beneath the 100 Commercial Street building. A DNAPL extraction well (RW-1) has been operating successfully on Area 2 since October 2001; to date approximately 670 gallons of DNAPL have been extracted from the subsurface using this well.

#### 6.03 Summary of Area 2 Remedial Objectives

Impacted media contributing to risk include DNAPL (shallow and deep), LNAPL, and soil. Remecial objectives are listed below:

#### NAPL:

Reduce thickness of DNAPL and LNAPL in monitoring wells to less than 1/2 in.



Soil:

- Reduce soil contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers, construction workers, and residents.

Other:

Allow for continued operation of the on-site businesses.

#### 6.04 Initial Screening of Remedial Technologies for Area 2

An initial screening of remedial technologies was conducted to identify technologies that are reasonably likely to be feasible for Area 2, based on the type of contamination observed within the Area, contaminated media present, and specific Area characteristics. Per 310 CMR 40.0856, remedial technologies are reasonably likely to be feasible if they are reasonably likely to achieve a Permanent or Temporary Solution, and if individuals with the necessary expertise to implement the technology are available to implement the technology.

Remedial technologies were screened for the contaminated media that contribute to exposure pathways for which a condition of No Significant Risk could not be satisfied for foreseeable future receptors on Area 2. A summary of the initial screening for Area 2 is presented in Table  $\lambda$ IV, including a brief description of the remedial technologies, and a rationale for retaining or eliminating the technology. Remedial technologies that meet the initial screening criteria are listed and briefly described below.

## A. DNAPL

- DNAPL extraction using recovery wells: Installation of large (8-in.) diameter recovery wells, equipped with a submersible pump to extract DNAPL from the subsurface. Extracted DNAPL would be pumped to small buildings on Site, stored and transported off-site for disposal or incineration. A DNAPL extraction well has been operating on the northern portion of this Area since October 2001, and has collected approximately 670 gallons of DNAPL to date.
- DNAPL extraction/DNAPL migration control using trenches: Excavation of trenches that are keyed into the low-permeability organic deposit, and backfilled with gravel to collect DNAPL. A recovery well would be installed in the backfilled trench, equipped with a submersible pump to extract and collect DNAPL. Trenches may be installed along Area boundary to cut off DNAPL flow and prevent migration of DNAPL onto the Area from adjacent Areas. Extracted DNAPL would be stored in small buildings on Site, and transported off-site for disposal or incineration.

.

Vertical subsurface barrier for DNAPL migration control: Installation of a barrier to DNAPL flow. The type of barrier would depend on the results of a feasibility assessment; potential options include a slurry wall, sheet piling, or a Waterloo<sup>TM</sup> Barrier. The barrier for Area 2 would also involve injection of



grout through the base of the WEB Culvert to prevent DNAPL migration through the crushed stone bedding beneath the culvert.

 Disposal or incineration: Off-site transport of extracted NAPL to an appropriate receiving facility for disposal or incineration.

## B. LNAPL

- Multi-phase Extraction (MPE): Extraction of groundwater, soil vapor and LNAPL through the application of a vacuum to specially designed extraction wells. Requires separation of NAPL from water and ex-situ treatment of both groundwater and soil vapor.
- Belt-skimmer: Extraction of LNAPL using belt-skimmers, which consist of two pulleys that drive a hydrophobic belt through the groundwater surface in a monitoring well and to the top of the well, bringing the LNAPL to the surface and skim the oil into a collection container. The belt-skimmer system is driven by an electrical motor.
- Disposal or Incineration: Off-site transport of extracted NAPL to an appropriate receiving facility for disposal or incineration
- C. Soil
  - Excavation and off-site treatment: Excavate and remove contaminated soil to an average depth of approximately 12 ft, along with DNAPL and LNAPL where present, and transport impacted soils off-site for treatment. Suitable excavated soils may be re-used as backfill on site. Implementation of this action would require support of excavations and building foundations during excavation, and may require demolition of on-site buildings if contamination is located beneath buildings. Implementation would also require an enclosure for control of vapors and odors.
  - Installation of an Engineered Barrier: RCRA-type Engineered Barrier consistent with 310 CMR 40.0000, to be installed over soil containing contaminant concentrations in excess of the UCL. Installation of an Engineered Barrier would require excavation of approximately 3 ft of soil, followed by placement of an HDPE impermeable membrane, which would be covered by approximately 1 ft of clean sand and a geo-composite drainage layer. The drainage layer would be covered by a marker fabric, and either pavement or a vegetated surface. Implementation of this alternative would require demolition of buildings, if present in areas of contamination. Would require deeper excavation and construction of a clean utility corridor at locations with subsurface utilities.



- Installation/Maintenance of a Direct Contact Barrier: Installation and/or maintenance of a surface barrier (i.e., 3 ft of soil, asphalt or concrete pavement), to prevent direct contact with surface soils.
- Incineration: Off-site transport of excavated soil to an appropriate receiving facility for incineration
- Ex-situ thermal desorption: Treatment of excavated soil material using thermal desorption with an afterburner. May be performed on or off-site; however on-site thermal desorption is not likely implementable due to the urban location and active use of the Area.
- In-situ chemical oxidation: Injection of hydrogen peroxide, Fenton's Reagent, or other oxidant into soil using temporary injection wells to chemically break down contaminants to innocuous compounds. This remedial method is not feasible when a significant amount of free-phase product (i.e., LNAPL or DNAPL) exists in the subsurface, as uncontrollable reactions can result.
- Activity and Use Limitations (AULs): Place restrictions on future property uses and activities to prevent potential exposure through a pathway for which a condition of No Significant Risk could not be satisfied.

## D. Elimination of Remedial Technologies Based on Site Characteristics

Area 2 represents a portion of the KeySpan Operations Center and consists of a large (37,000 sf) commercial building and a parking lot used to store a fleet of service vehicles. Remedial Action Alternatives for Area 2 are developed based on the assumptions that current use continues into the future (i.e., the existing building remains on site, and that the commercial business is not re-located for the installation of remedial components on the Area). Area 2 is located in a public, urban area, and experiences high traffic volume in its current use. Excavation of soil on Area 2 to achieve the remedial objectives is not considered reasonably feasible at this time due to the presence of the large building on the Area. Based on this reasoning, six remedial technologies that met Initial Screening criteria are not reasonably feasible on Area 2 and will not be incorporated into Remedial Action Alternatives for Area 2. These technologies are summarized below in Table XV:



## TABLE XV

IADLE AV						
REMEDIAL	<b>TECHNOLOGIES E</b>	LIMINATED	BASED (	ON SITE (	CHARACTER	ISTICS
AREA 2						

Remedial Technology	Technology Type	Target Media	Reason For Elimination	
Extraction using trenches	Extraction	Shallow DNAPL	Significantly more disruptive and difficult to implement than extraction wells, which have been demonstrated to be effective on Area 2 for extraction of DNAPL.	
DNAPL migration control using trenches	DNAPL migration control	Shallow DNAPL	Significantly more disruptive and difficult to implement than vertical subsurface barriers, which would likely be effective for DNAPL migration control.	
Excavation and off-site treatment	Physical Removal	Soil/TSM	Excavation beneath the building is infeasible; therefore this remedial method cannot achieve a Permanent Solution for the Site.	
Incineration	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.	
Thermal desorption	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.	
Engineered Barrier	Barrier	Soil/TSM	Installation of an Engineered Barrier would not result in a Permanent Solution for the Area due to the presence of a building on the Area; contamination is located beneath the building and the existing building cannot serve as a barrier to potential exposure.	

## 6.05 Development of Remedial Action Alternatives for Area 2

Three Remedial Action Alternatives that are reasonably likely to meet the remedial objectives have been developed for Area 2 and are described below. Approximate locations of the components of the Remedial Alternatives for Area 2 are shown on Figure 17.

 Alternative 2-1: Shallow and deep DNAPL recovery, LNAPL recovery using MPE, provisional shallow DNAPL migration control, in-situ chemical oxidation of TSM and LNAPL-impacted soil, and AUL

Alternative 2-1 involves the installation of shallow and deep DNAPL extraction wells equipped with submersible pumps, MPE wells for the extraction of LNAPL, provisional shallow DNAPL migration control using vertical subsurface barriers, insitu chemical oxidation of TSM soil, and AULs to ensure that current industrial/commercial property conditions are maintained on the property. In this alternative, approximately 13 DNAPL extraction wells would be installed in locations on Area 2 where DNAPL is known to exist based on subsurface investigations to date, and where low spots are known or inferred to exist in the organic peat and silt deposit. Based on the estimated DNAPL volume of approximately 11,500 gallons and



on observed recovery rates observed in RW-1, it is estimated that attainment of the DNAPL remedial goals will require approximately 7 to 10 years.

Alternative 2-1 also includes the installation of approximately 12 MPE wells. Because it applies a very high vacuum in the vicinity of the extraction wells, MPE typically results in rapid LNAPL extraction rates and is included as an aggressive approach to remove LNAPL from the subsurface. It is estimated that MPE would require approximately 2 years to reach the LNAPL remedial goals on Area 2.

In-situ chemical oxidation is proposed in Alternative 2-1 to reduce concentrations of contaminants in accessible soil once the free-phase product is removed. In-situ chemical oxidation would be implemented using a network of approximately 245 remporary injection wells (assuming a 20-ft radius of influence), and is anticipated to require approximately 2 to 3 years to implement. It is assumed that chemical oxidation is not reasonably feasible beneath the 100 Commercial Street building or beneath the WEB Culvert due to the disruption this would cause to the commercial business located in the building, and the potential for the build-up of heat and vapors beneath the building and the culvert. Therefore, in-situ chemical oxidation would be conducted in accessible soils only. Chemical oxidation of TSM and LNAPL-impacted soils is not feasible until DNAPL and LNAPL have been removed, to the extent practicable; therefore it would not be implemented until the NAPL remedial goals have been reached on the Area. Implementation of chemical oxidation on Area 2 would require significant reductions in use of the Area surrounding the 100 Commercial Street building during installation of temporary injection wells and injection of oxidants.

Vertical subsurface barriers to shallow DNAPL flow may be installed to prevent migration of shallow DNAPL onto Area 2 from adjacent areas. This may be desired in the event that remedial goals are met and a Permanent Solution is reached on Area 2 before they are reached on adjacent areas. The vertical barrier for Area 2 would involve driven sheet piling or a Waterloo<sup>TM</sup> Barrier, as shown on Figure 17, and grout injected through the base of the WEB Culvert to prevent potential DNAPL migration in the crushed stone beneath the culvert.

Alternative 2-1 is anticipated to require approximately 9 to 13 years to complete. However, contaminants (TSM and shallow DNAPL) would likely remain beneath the 100 Commercial Street building following implementation of Alternative 2-1, necessitating an AUL. Because TSM (i.e., UCL exceedences in soil) and shallow DNAPL would likely remain beneath the building in this alternative, Alternative 2-1 would not achieve a Permanent Solution and therefore represents a Temporary Solution.

 Alternative 2-2: Shallow and deep DNAPL recovery, LNAPL recovery using beltskimmers, provisional shallow DNAPL migration control, and AUL

Alternative 2-2 involves the installation of 13 shallow and deep DNAPL extraction wells, 6 LNAPL extraction wells equipped with belt-skimmers, provisional shallow



DNAPL migration control using vertical subsurface barriers, and AULs to ensure that current property conditions are maintained on the Area. It is anticipated that the LNAPL remedial goals may be met in approximately 4 to 6 years, and the DNAPL remedial goals may be met in approximately 7 to 10 years. However, because DNAPL likely exists beneath the 100 Commercial Street building, it is possible that the DNAPL remedial goals may not be achievable on Area 2 under current conditions and use. TSM would also remain on the Area in Alternative 2-2. Therefore, Alternative 2-2 also represents a Temporary Solution.

Alternative 2-3: LNAPL and shallow and deep DNAPL monitoring and AUL

Alternative 2-3 is a monitoring alternative. Based on Site data collected to date, the LNAPL and DNAPL plume dimensions appear stable; this alternative proposes monitoring to ensure that the extents of LNAPL and DNAPL impacts do not expand.

#### 6.06 Detailed Evaluation of Remedial Action Alternatives for Area 2

This section presents a Detailed Evaluation of the three Remedial Action Alternatives presented above, based on the criteria listed in Section 4.03. As with the other Areas, this evaluation is based on the current property conditions and use, as described in Section 2.06 and summarized in Section 6.02. A summary of the detailed evaluation of the three Remedial Action Alternatives developed for Area 2 is presented in Table XVI. Table XVI presents a summary of the likely effectiveness, reliability, implementability, cost (net present value, NPV), risks, benefits, timeliness, and potential effect on non- pecuniary interests of each alternative. The presence of the existing KeySpan operations building currently located on Area 2 limits the implementability and effectiveness of some remedial alternatives on Area 2. Because it is likely that DNAPL and TSM soil containing contaminant concentrations greater than U/CLs are located beneath the building, a Permanent Solution is not proposed for Area 2; the remedial alternatives evaluated below represent Temporary Solutions. A comparison of the three remedial alternatives based on these criteria is presented below.

Effectiveness: Because Alternative 2-1 includes LNAPL extraction using MPE, Alternative 2-1 is likely to provide more effective and more rapid LNAPL removal than Alternative 2-2. MPE is generally more effective in low-permeability soils; the effectiveness of MPE on Area 2 will therefore depend on the permeability of soils within the LNAPL area. However, the same method of DNAPL extraction is proposed in Alternatives 2-1 and 2-2; therefore DNAPL extraction would occur at the same rate in both of these alternatives. It is unlikely that either Alternatives 2-1 or 2-2 would reach the DNAPL remedial goals, because DNAPL would likely remain beneath the 100 Commercial Street building; however, they would likely be effective at removing significant DNAPL mass from the subsurface in this Area. Additionally, it is likely that TSM exists beneath the building, and would also remain beneath the building under these alternatives. Therefore, neither Alternative 2-1 nor 2-2 would reach a Permanent Solution. Both Alternatives 2-1 and 2-2 are ranked with a "moderate" level of effectiveness, because although they do not reach a Permanent Solution, they will reduce the volume of DNAPL and LNAPL present on the Area. Alternative 2-3 also would not provide a Permanent Solution, as it is a monitoring-



only alternative, and is given a "low" effectiveness ranking, because it does not reduce the volume of NAPL on Area 2. Each of the evaluated alternatives represents a Temporary Solution for Area 2.

Short and Long Term Reliability: The technologies proposed in Alternatives 2-1 (DNAPL extraction wells, MPE, and in-situ chemical oxidation) and 2-2 (DNAPL extraction wells and belt-skimmers) are considered reliable remedial technologies. Both alternatives are considered moderately reliable in the short and long term, although MPE is considered a slightly more reliable method of LNAPL extraction than belt-skimmers. The reliability criterion is not relevant for Alternative 2-3, as a remedial system is not proposed in this Alternative.

2

- Implementability: Alternative 2-1 is considered somewhat difficult to implement, because the proposed MPE system would require extraction equipment and a groundwater and vapor treatment system on Area 2 and the chemical oxidation component would require approximately 245 injection points. Injection of oxidants beneath the buildings on Area 2 is not considered reasonably feasible due to the disruption this would cause to the commercial businesses located in the buildings, and the potential for the build-up of heat and vapors beneath the buildings. Therefore it is assumed that chemical oxidation would not be conducted beneath the buildings on Area 3. These technologies could present significant logistical problems for the Area facility operations. Alternative 2-2 is readily implementable; the belt-skimmers included in Alternative 2-2 would involve considerably less intrusion than MPE into the commercial operations on the Area, as significantly less space and equipment are required to implement this remedial component. Additionally, Alternative 2-2 does not include in-situ chemical oxidation, which is not fully implementable due to the presence of the building on Area 2. Alternative 2-3, which includes monitoring of LNAPL and DNAPL on Area 2, is readily implementable.
- Cost: Alternative 2-1, which is the most aggressive remedial alternative, is also the most costly alternative, with an estimated NPV of approximately \$2.4M. The estimated NPV for Alternative 2-2 is approximately \$1.3M. The estimated NPV for Alternative 2-3, which is a monitoring alternative, is approximately \$110K.
- Risks: Low to moderate risk is associated with the three alternatives evaluated for Area 2. Moderate risks associated with Alternative 2-1 include those associated with handling of extracted LNAPL and DNAPL, and handling the caustic oxidants during in-situ chemical oxidation. Low risk is associated with implementation of Alternative 2-2, including the handling of extracted LNAPL and DNAPL. Risks associated with Alternative 2-2 are somewhat less than those associated with Alternative 2-1, as Alternative 2-2 does not involve the handling of caustic oxidants. Alternative 2-3 involves few risks of exposure to hazardous chemicals.

Long-term risks due to potential exposure to contamination left in place are higher in Alternatives 2-2 and 2-3, because these alternatives do not include measures to reduce contaminant concentrations in soil. This long-term risk is reduced somewhat in



Alternative 2-1, which includes in-situ chemical oxidation to reduce contaminant concentrations in accessible soils.

- Benefits: Alternatives 2-1 and 2-2 both provide the benefit of reducing the quantity of LNAPL and DNAPL in the subsurface. The completeness of the removal of DNAPL is dependent on the ability of DNAPL extraction wells surrounding the 100 Commercial Street building to remove DNAPL from beneath the building. Alternative 2-1 includes some additional benefit due to the in-situ chemical oxidation component, which would reduce contaminant concentrations in accessible soil. However, soil beneath the building would not be accessible for chemical oxidation treatment. Alternative 2-3 produces no significant beneficial reduction of contaminants, as it includes neither NAPL extraction nor soil remediation.
- Timeliness: LNAPL removal using MPE in Alternative 2-1 is likely to require approximately 2 years, which would be more rapid than removal using belt-skimmers in Alternative 2-2, which is anticipated to require approximately 4 to 6 years. The method and therefore the rate of DNAPL extraction is the same in both Alternatives 2-1 and 2-2; DNAPL extraction is expected to continue for approximately 7 to 10 years on Area 2. It is unlikely that the DNAPL extraction wells proposed in Alternatives 2-1 and 2-2 will accomplish the DNAPL remedial goals, as a portion of the DNAPL on Area 2 is located beneath the 100 Commercial Street building. Chemical oxidation of TSM and LNAPL-impacted soil, which is proposed as a remedial component in Alternative 2-1, would not be implementable until LNAPL and DNAPL have been reduced to the extent practicable. In-situ chemical oxidation is anticipated to require approximately 2 to 3 years to reach the soil remedial goals; however soils beneath the 100 Commercial Street building are not accessible for chemical oxidation treatment. Therefore the remedial goals will not be met for soils beneath the building. Alternative 2-3 could be implemented immediately and would continue for the foreseeable future, until a Permanent Solution can be implemented.
- Effect on Non-pecuniary Interests: Alternative 2-1 results in more adverse effects on non-pecuniary interests, because the MPE system and chemical oxidation injection proposed in this alternative would require more equipment and would have a greater impact on property use than Alternative 2-2. Both Alternatives 2-1 and 2-2 would require the construction of a small NAPL storage building on the Area, which is not expected to have a significant effect on the aesthetics or use of the Area. The requirement for a small DNAPL storage building may be satisfied through modifications to the existing building located on the northern side of the KeySpan Operations building, which was constructed in association with the installation of the existing DNAPL recovery well. Alternative 2-3 would have very little effect on use of the Area or aesthetics.

## 6.07 Selected Remedial Action Alternative for Area 2

In summary, although Alternative 2-1 may be slightly more effective and reliable and provide marginal additional benefit over Alternative 2-2, neither alternative would provide a



Permanent Solution for Area 2. Implementation of Alternative 2-1 would likely provide slightly more effective, reliable, and rapid LNAPL removal than Alternative 2-2, and would reduce the quantity of DNAPL on the Area. Alternative 2-1 would also provide additional benefit of reduced contaminant concentrations in accessible soil. However, Alternative 2-1 would not reach a Permanent Solution because DNAPL and TSM would likely remain beneath the 100 Commercial Street building. The belt-skimmers proposed for LNAPL extraction in Alternative 2-2 are considered to be a relatively reliable technology, and would provide effective LNAPL extraction over time. Implementation of Alternative 2-2 would also provide for effective reduction in the quantity of DNAPL in accessible areas, and like Alternative 2-1, Alternative 2-2 would not result in a Permanent Solution. Alternative 2-3, which involves monitoring of LNAPL and DNAPL, also would not result in a Permanent Solution. Based on the successful operation of a DNAPL extraction well on Area 2, DNAPL extraction is feasible on the property and therefore Alternative 2-3, which does not include DNAPL extraction, is not considered a viable alternative. Both Alternatives 2-1 and 2-2 would have the same result (i.e., would result in a Temporary Solution due to the contamination that would remain beneath the 100 Commercial Street building). Therefore, because of the implementation difficulties, impact on use, and higher cost of Alternative 2-1, Alternative 2-2 is the selected remedy for Area 2.

Implementation of Alternative 2-2 would involve the installation of approximately 13 DNAPL recovery wells, placed in locations of likely DNAPL accumulation, as shown on Figure 17. Approximately 6 belt-skimmer wells would be installed on the southwestern corner of the Area to remove LNAPL, as shown on Figure 17. A vertical barrier to DNAPL migration may also be installed, if necessary. The anticipated time requirement to reach the DNAPL remedial goals is approximately 7 to 10 years, based on assumed DNAPL extraction rates. However, it is likely that DNAPL will remain beneath the 100 Commercial Street building on Area 2.



## VII. EVALUATION OF REMEDIAL ALTERNATIVES FOR AREA 3

#### 7.01 Introduction

In this section, Remedial Action Alternatives to address contamination identified in Area 3, the north ern portion of Parcel A. Area 3 is bounded by Centre Street to the north, the MBTA Orange line to the west, Area 4 to the south, and Commercial Street to the east. Area 3 is comprised of four separately owned properties, including 51 Commercial Street, 65 Commercial Street, 77 Commercial Street, and 89 Commercial Street. Use of these properties is currently commercial, and involves a high degree of public access. The WEB Culvert crosses Area 3 between the 77 Commercial Street and the 89 Commercial Street properties. A brief summary of the existing conditions and remedial objectives for Area 3 are provided below. Based on these conditions and objectives, an initial screening of remedial technologies is presented in Section 7.04 to identify technologies that may be applicable to address contamination identified on Area 3. Remedial technologies that meet the initial screening criteria but are not applicable on Area 3 due to specific site characteristics are eliminated from further consideration in Section 7.04. Remedial Action Alternatives are then developed in Section 7.05 by combining the remaining remedial technologies based on knowledge of site conditions and engineering judgement. A detailed evaluation of the Remedial Action Alternatives is then conducted in Section 7.06, and the recommended Remedial Alternative is presented in Section 7.07.

#### 7.02 Summary of Area 3 Conditions

TSM and shallow DNAPL contamination has been identified on Area 3. This contamination is likely residual from historic MGP operations and the operation of the American Tar Company, which operated on the northern portion of Area 3 and land to the north of Area 3, which is now Centre Street. The historic location of the American Tar Company is shown on Figure 5. According to the conceptual model for the Site, DNAPL flow beneath Area 3 is by gravity along the upper surface of the organic deposit. The majority of the DNAPL in the subsurface on Area 3 is believed to be located in the historic Malden River channel, which is a low area in the organic deposit and is shown on Figure 7. The volume of shallow DNAPL in the subsurface on Area 3 is estimated to be approximately 3,700 gallons, and the areal extent of TSM-impacted soils is estimated as 8,400 sy. These estimates include DNAPL and impacted soil that likely exist beneath the buildings located on Area 3.

#### 7.03 Summary of Area 3 Remedial Objectives

Impacted media on Area 3 include shallow DNAPL and soil. Remedial objectives are listed below:

#### NAPL:

Reduce thickness of DNAPL in monitoring wells to less than 1/2 in.

#### Soil:

Reduce soil contaminant concentrations to less than the UCLs.


Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers and construction workers.

#### Other:

Allow for continued operation of the on-site businesses.

#### 7.04 Initial Screening of Remedial Technologies for Area 3

An initial screening of remedial technologies was conducted to identify technologies that are reasonably likely to be feasible for Area 3, based on the type of contamination observed within the Area, contamination present, and specific Area characteristics. Per 310 CMR 40.0856, remedial technologies are reasonably likely to be feasible if they are reasonably likely to achieve a Permanent or Temporary Solution, and if individuals with the necessary expertise to implement the technology are available to implement the technology.

Remedial technologies were screened for the contaminated media that contribute to exposure pathways for which a condition of No Significant Risk could not be satisfied for future receptors on Area 3. A summary of the initial screening for Area 3 is presented in Table XVII, including a brief description of the remedial technologies, and a rationale for retaining or eliminating the technology. Remedial technologies that meet the initial screening criteria are listed and briefly described below by the medium they address.

- A. DNAPL
  - DNAPL extraction using recovery wells: Installation of large (8-in.) diameter recovery wells, equipped with a submersible pump to extract DNAPL from the subsurface. Extracted DNAPL would be pumped to small buildings on Site, stored and transported off-site for disposal or incineration. A DNAPL extraction well has been operating on the northern portion of Area 2 since October 2001, and has collected approximately 670 gallons of DNAPL to date.
  - DNAPL extraction/DNAPL migration control using trenches: Excavation of trenches that are keyed into the low-permeability organic deposit, and backfilled with gravel to collect DNAPL. A recovery well would be installed in the backfilled trench, equipped with a submersible pump to extract and collect DNAPL. Trenches may be installed along Area boundary to cut off DNAPL flow and prevent migration of DNAPL onto the Area from adjacent Areas. Extracted DNAPL would be stored in small buildings on Site, and transported off-site for disposal or incineration.
  - Vertical subsurface barrier for DNAPL migration control: Installation of a barrier to DNAPL flow. The type of barrier would depend on the results of a feasibility assessment; potential options include a slurry wall, sheet piling, or a Waterloo<sup>TM</sup> Barrier. The barrier for Area 3 could also include injection of grout through the base of the WEB Culvert to prevent DNAPL migration through the crushed stone bedding beneath the culvert.



#### C. Elimination of Remedial Technologies Based on Site Characteristics

As described above, Area 3 is comprised of four individually owned properties with five operating commercial businesses. Each property consists of a small building a parking lot, and minimal landscaped space. Area 3 is located in a public, urban area, with a high degree of public access in its current use. Therefore, remedial technologies that result in significant disruption to business operations or public nuisance are considered infeasible. Likewise, DNAPL extraction or remediation of TSM or soil beneath buildings to achieve the remedial objectives is not considered feasible. Based on this reasoning, six remedial technologies are not reasonably feasible on Area 3 and will not be incorporated into Remedial Action Alternatives for Area 3. These remedial technologies are summarized below in Table XVIII:

#### TABLE XVIII

REMEDIAL TECHNOLOGIES ELIMINATED BASED ON SITE CHARACTERISTICS AREA 3

Remedial Technology	Technology Type	Target Media	Reason For Elimination
Extraction using trenches	Extraction	Shallow DNAPL	Significantly more disruptive and difficult to implement than extraction wells, which have been demonstrated to be effective on Area 3 for extraction of DNAPL.
DNAPL migration control using trenches	DNAPL migration control	Shallow DNAPL	Significantly more disruptive and difficult to implement than vertical subsurface barriers, which would likely be effective for DNAPL migration control.
Excavation and off-site: Disposal	Physical Removal	Soil/TSM	Excavation beneath the building is infeasible; therefore this remedial method cannot achieve a Permanent Solution for the Site.
Incine ation	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.
Thermal desorption	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.
Engin:ered Barrier	Barrier	Soil/TSM	Installation of an Engineered Barrier would not result in a Permanent Solution for the Area due to the presence of four buildings on the Area; contamination is likely located beneath the existing buildings and the buildings cannot serve as barriers to potential exposure.

#### 7.05 Development of Remedial Action Alternatives for Area 3

Three Remedial Action Alternatives developed to meet the remedial goals for Area 3 are discussed below. Approximate locations of the components of the Remedial Alternatives for Area 3 are shown on Figure 18.



Alternative 3-1: Shallow DNAPL recovery, provisional shallow DNAPL migration control, in-situ chemical oxidation of TSM and soil, and AUL

Alternative 3-1 would involve the installation of approximately 12 shallow DNAPL recovery wells, and construction of a small storage building to store recovered DNAPL prior to off-site disposal. Based on the estimated volume of DNAPL on Area 3 (approximately 3,700 gallons) and observed recovery rates for DNAPL recovery RW-1 on Area 2, it is estimated that the DNAPL remedial goals may be attained in approximately 3 to 5 years. However, because of the presence of the four buildings on Area 3, DNAPL may remain in the subsurface, preventing the attainment of a Permanent Solution for the Area. The migration of shallow DNAPL from adjacent areas onto Area 3 could be prevented using a vertical subsurface barrier, which would consist of driven sheeting as indicated on Figure 18. The vertical barrier could also involve injection of grout through the base of the WEB Culvert to prevent DNAPL migration through the crushed stone base beneath the culvert. The locations of the vertical barriers coincide with the historic river channels that previously flowed from east to west in the northern portion of Area 3, then meandered to the south and back to the west at the approximate location of the current WEB culvert. Given the variations in the historic river bottom, as expressed by the depressed area of the organic deposit, it is not possible to make a direct correlation between the former flow direction of the river and the migration direction of shallow DNAPL.

Following completion of DNAPL recovery, TSM soil would be treated in-situ through injection of oxidants in accessible soils, using a network of approximately 180 injection wells (assuming 20-ft radius of influence) to promote chemical oxidation reactions. It is assumed that chemical oxidation will not be conducted beneath buildings on Area 3 or beneath the WEB Culvert, due to the possibility for the build-up of heat and vapors beneath these structures. It is anticipated to require three oxidant injections over a period of approximately 2 to 3 years. During implementation of chemical oxidation on Area 3, use of the parking areas surrounding the buildings on the Area would be significantly reduced, due to temporary well installation and oxidant injection activities. AULs may be required to mitigate risk through future soil exposure pathways.

 Alternative 3-2: Shallow DNAPL recovery, provisional shallow DNAPL migration control and AUL

Alternative 3-2 is a less intrusive remedial option to address the presence of TSM and shallow DNAPL contamination on Area 3. This alternative involves the installation of approximately 12 shallow DNAPL recovery wells to recover DNAPL from the subsurface, as proposed in Alternative 3-1. If shallow DNAPL migration control is required, vertical subsurface barriers may be installed to prevent migration of DNAPL onto Area 3 from adjacent properties. This alternative does not include a remedial component to address residual soil TSM contamination and will not achieve a Permanent Solution. An AUL may be required to mitigate risk through future soil exposure pathways.



#### Alternative 3-3: Shallow DNAPL monitoring and AUL

Alternative 3-3 is a monitoring alternative. Based on Site data collected to date, the DNAPL plume dimensions appear stable; this alternative proposes monitoring to ensure that the extent of DNAPL impact does not expand.

#### 7.06 Detailed Evaluation of Remedial Action Alternatives for Area 3

This section presents a Detailed Evaluation of the three Remedial Action Alternatives presented above, based on the criteria listed in Section 4.03. As with the other Areas, this evaluation is based on the current property conditions and use, as described in Section 2.06 and summarized in Section 7.02. A summary of the detailed evaluation of the three Remedial Action Alternatives developed for Area 3 is presented in Table XIX. Table XIX presents a summary of the likely effectiveness, reliability, implementability, cost (net present value, or NPV), risks, benefits, timeliness, and potential effect on non- pecuniary interests of each alternative. The presence of the existing buildings currently located on Area 3 limits the implementability of Alternatives 3-1 and 3-2. Because it is likely that DNAPL and soil containing contaminant concentrations greater than UCLs are located beneath these buildings, a Permanent Solution is not proposed for Area 3; the remedial alternatives evaluated below represent Temporary Solutions. A comparison of the three remedial alternatives based on these criteria is presented below.

- Effectiveness: It is unlikely that Alternatives 3-1 or 3-2, both of which include DNAPL extraction using extraction wells, would reach the DNAPL remedial goals, because DNAPL would likely remain beneath the buildings located on Area 3. Additionally, it is likely that TSM exists beneath the buildings, and would also remain beneath the building under Alternatives 3-1 and 3-2. Therefore, although Alternatives
  - 3-1 and 3-2 would reduce the quantity of DNAPL on Area 3, neither would reach a Permanent Solution. Both Alternatives 3-1 and 3-2 are ranked with a "moderate" effectiveness. Alternative 3-3 also would not provide a Permanent Solution, as it is a monitoring-only alternative, and does not reduce the quantity of DNAPL. Therefore Alternative 3-3 is ranked as "low" under the effectiveness category. Each of the three evaluated alternatives represents a Temporary Solution for Area 3.
- Short and Long Term Reliability: The technologies proposed in Alternatives 3-1 (DNAPL extraction wells and in-situ chemical oxidation of soil) and 3-2 (DNAPL extraction wells) are considered reliable remedial technologies. Both alternatives 3-1 and 3-2 are considered moderately reliable in the short and long term. Reliability is not considered relevant to Alternative 3-3, as a remedial system is not installed under this Alternative.

Implementability: The DNAPL extraction components of Alternatives 3-1 and 3-2 are readily implementable, although implementability of a DNAPL extraction system would be contingent upon agreements with property owners to construct a small DNAPL storage building on one of the properties on Area 3. Additionally, installation of 180 chemical oxidation wells on Area 3, as proposed in Alternative 3-1, would likely cause significant disruption to the operating businesses on the Area. As



component in Alternative 3-1, would not be implementable until the quantity of DNAPL has been reduced to the extent practicable using DNAPL extraction wells. Once it is implementable, in-situ chemical oxidation is anticipated to require approximately 2 to 3 years to reach the soil remedial goals; however soils beneath the buildings on Area 3 are not accessible for chemical oxidation treatment. Therefore the remedial goals would not likely be met for soils beneath the building. Alternative 3-3 could be implemented immediately and would continue for the foreseeable future, until a Permanent Solution can be implemented.

Effect on Non-pecuniary Interests: Alternatives 3-1 and 3-2 would both have a moderate impact on property use and aesthetics. Both of these alternatives would require the construction of a small NAPL storage building on a property within the Area. Alternative 3-1 would have additional impact on property use, as this alternative would involve three injections of chemical oxidant into soil beneath the Area. These injections may not be acceptable to the property owners, due to the public nature of the businesses operating on these properties. Alternative 3-3 would have very little effect on use of the Area or aesthetics.

#### 7.07 Selected Remedial Action Alternative for Area 3

Implementation of either Alternatives 3-1 and 3-2 would result in effective, reliable reduction in the quantity of DNAPL in accessible portions of Area 3. Neither alternative would result in timely achievement of the DNAPL remedial goals; this is anticipated to require approximately 3 to 5 years for both of these alternatives. The in-situ chemical oxidation component included in Alternative 3-1 would provide additional benefit of reduction in contaminant concentrations; however this remedial component cannot be implemented until the quantity of DNAPL has been reduced to the extent practicable. Neither Alternative 3-1 nor 3-2 would reach a Permanent Solution because DNAPL and TSM would likely remain beneath the buildings located on Area 3. Alternative 3-3, which involves monitoring of DNAPL, also would not result in a Permanent Solution. Based on the successful operation of a DNAPL extraction well on Area 2, DNAPL extraction is feasible on the Site. Therefore, DNAPL extraction should be included in the selected remedy for Area 3 and Alternative 3-3 is not considered a viable alternative. Both Alternatives 3-1 and 3-2 would have the same outcome (i.e., would result in a Temporary Solution due to the contamination that would remain beneath the buildings). Therefore, because of the implementation difficulties, impact on use, and higher cost of Alternative 3-1, Alternative 3-2 is the selected remedy for Area 3.

Alternative 3-2 involves the installation of approximately 12 shallow DNAPL extraction wells, which would be installed in locations at which the top of the organic deposit is relatively low. Proposed locations for DNAPL extraction wells are shown on Figure 18; DNAPL extraction well locations are proposed along the historic Malden River channel, in order to maximize the effectiveness of this alternative.



#### VIII. EVALUATION OF REMEDIAL ALTERNATIVES FOR AREA 4

#### 8.01 Introduction

In this section, Remedial Action Alternatives are developed to address contamination identified in Area 4, which is the southern portion of Parcel A, as shown on Figure 19. Area 4 is bounded by Charles Street to the south, the MBTA Orange line to the west, Area 3 to the north, and Commercial Street to the east. Area 4 is comprised of one property on which two buildings are located, containing medical offices and other businesses involving a high degree of public use. A brief summary of the existing conditions and remedial objectives for Area 4 are provided below. Based on these conditions and objectives, an initial screening of remedial technologies is presented in Section 8.04 to identify technologies that may be applicable to address contamination identified on Area 4. Remedial technologies that meet the initial screening criteria but are not applicable on Area 4 due to specific site characteristics are eliminated from further consideration in Section 8.04. Remedial Action Alternatives are then developed in Section 8.05 by combining the remaining remedial technologies based on knowledge of site conditions and engineering judgement. A detailed evaluation of the Remedial Action Alternatives is then conducted in Section 8.06, and the recommended Remedial Alternative is presented in Section 8.07.

#### 8.02 Summary of Area 4 Conditions

LNAPL has been identified in one monitoring well on the southwestern portion of Area 4, and in a monitoring well located in the Commercial Street ROW, just east of Area 4. A small area of TSM contamination has been identified in the southern portion of Area 4, as well as along the northern boundary of Area 4 with Area 3. This contamination is likely residual from historic MGP operations; historic MGP features that have been located on Area 4 include gas production buildings, condensers, tar filters, gas holders and purification facilities. The historic location of MGP features is shown on Figure 5. The volume of LNAPL in the subsurface on Area 4 is estimated to be approximately 5,600 gallons. The areal extent of TSM-impacted soil is estimated as approximately 560 sy, and the volume of LNAPL-impacted soils is estimated as approximately 640 sy. These estimates do not include contaminated soils that may be located beneath the buildings located on Area 4.

#### 8.03 Summary of Area 4 Remedial Objectives

Area 4 contains impacts from TSM and LNAPL. Remedial objectives are listed below:

#### NAPL:

Reduce thickness of LNAPL in monitoring wells to less than 1/2 in.

#### Soil:

- Reduce soil contaminant concentrations to less than the UCLs.
- Prevent exposure to soil within 15 ft of the ground surface to address potential risk to future on-site industrial/commercial workers and construction workers.



Other:

Allow for continued operation of the on-site businesses.

# 8.04 Initial Screening of Remedial Technologies for Area 4

An initial screening of remedial technologies was conducted to identify technologies that are reasonably likely to be feasible for Area 4, based on the type of contamination observed within the Area, contamination present, and specific Area characteristics. Per 310 CMR 40.0856, remedial technologies are reasonably likely to be feasible if they are reasonably likely to achieve a Permanent or Temporary Solution, and if individuals with the necessary expertise to implement the technology are available to implement the technology.

Remedial technologies were screened for the contaminated media that contribute to exposure pathways for which a condition of No Significant Risk could not be satisfied for current or future receptors on Area 4. A summary of the initial screening for Area 4 is presented in Table XX, including a brief description of the remedial technologies, and a rationale for retaining or eliminating the technology. Remedial technologies that meet the initial screening criteria are listed and briefly described below by the medium they address.

- A. LNAPL
  - Multi-phase Extraction (MPE): Extraction of groundwater, soil vapor and LNAPL through the application of a vacuum to specially designed extraction wells. Requires separation of NAPL from water and ex-situ treatment of both groundwater and soil vapor.
  - Belt-skimmer: Extraction of LNAPL using belt-skimmers, which consist of two pulleys that drive a hydrophobic belt through the groundwater surface in a monitoring well and to the top of the well, bringing the LNAPL to the surface and skim the oil into a collection container. The belt-skimmer system is driven by an electrical motor.
  - Disposal or Incineration: Off-site transport of extracted NAPL to an appropriate receiving facility for disposal or incineration.
- B. Soil
  - Excavation and off-site treatment: Excavate and remove contaminated soil to an average depth of approximately 12 ft, along with LNAPL where present, and transport impacted soils off-site for treatment. Suitable excavated soils may be re-used as backfill on site. Implementation of this action would require support of excavations and building foundations during excavation, and may require demolition of on-site buildings if contamination is located beneath buildings. Implementation would also require an enclosure for control of vapors and odors.



- Installation of an Engineered Barrier: RCRA-type Engineered Barrier consistent with 310 CMR 40.0000, to be installed over soil containing contaminant concentrations in excess of the UCL. Installation of an Engineered Barrier would require excavation of approximately 3 ft of soil, followed by placement of an HDPE impermeable membrane, which would be covered by approximately 1 ft of clean sand and a geo-composite drainage layer. The drainage layer would be covered by a marker fabric, and either pavement or a vegetated surface. Implementation of this alternative would require demolition of buildings, if present in areas of contamination. Would require deeper excavation and construction of a clean utility corridor at locations with subsurface utilities.
- Installation/Maintenance of a Direct Contact Barrier: Installation and/or maintenance of a surface barrier (i.e., 3 ft of soil, asphalt or concrete pavement), to prevent direct contact with surface soils.
- Incineration: Off-site transport of excavated soil to an appropriate receiving facility for incineration
- Ex-situ thermal desorption: Treatment of excavated soil material using thermal desorption with an afterburner. May be performed on or off-site; however on-site thermal desorption is not likely implementable due to the urban location and active use of the Area.
- In-situ chemical oxidation: Injection of hydrogen peroxide, Fenton's Reagent, or other oxidant into soil using temporary injection wells to chemically break down contaminants to innocuous compounds. This remedial method is not feasible when a significant amount of free-phase product (i.e., LNAPL) exists in the subsurface, as uncontrollable reactions can result.
- Activity and Use Limitations (AULs): Place restrictions on future property uses and activities to prevent potential exposure through a pathway for which a condition of No Significant Risk could not be satisfied.

# C. Elimination of Remedial Technologies Based on Site Characteristics

As described above, Area 4 consists of a privately owned property, on which two buildings are located with operating commercial businesses. The property consists of two small buildings, parking space, and minimal landscaped space. Area 4 is located in a relatively public, urban area, with a high degree of public access in its current use. Therefore, remedial technologies that result in significant disruption to business operations or public nuisance are considered infeasible. Likewise, LNAPL extraction or remediation of TSM or soil beneath buildings to achieve the remedial objectives is not considered feasible due to the presence of the two buildings on the Area. Based on this reasoning, four remedial technologies are not reasonably feasible on Area 4 and will not be incorporated into Remedial Action Alternatives for Area 4. These technologies are listed below in Table XXI:



#### TABLE XXI

Trubble Port					
REMEDIAL	<b>TECHNOLOGIES ELIMINATED</b>	BASED	ON SITE	CHARACTERISTICS	3
AREA 4					

Remedial Technology	Technology Type	Target Media	Reason For Elimination
Excaviation and off-site Disposal	Physical Removal	Soil/TSM	Excavation beneath the building is infeasible; therefore this remedial method cannot achieve a Permanent Solution for the Site.
Incineration	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.
Thermal desorption	Ex-situ treatment	Soil/TSM	Not necessary if soil is not excavated.
Engineered Barrier	Barrier	Soil/TSM	Installation of an Engineered Barrier would not result in a Permanent Solution for the Area due to the presence of two buildings on the Area; contamination is likely located beneath the buildings and the existing buildings cannot serve as barriers to potential exposure.

#### 8.05 Development of Remedial Action Alternatives for Area 4

Three Remedial Action Alternatives that are reasonably likely to meet the remedial objectives have been developed for Area 4 and are described below. Approximate locations of the comportents of the Remedial Alternatives for Area 4 are shown on Figure 19.

 Alternative 4-1: LNAPL recovery using MPE, in-situ chemical oxidation of TSM and LNAPL-impacted soil, and AUL

Alternative 4-1 involves the installation of MPE wells for the extraction of LNAPL, in-situ chemical oxidation of TSM and LNAPL-impacted soils, and AULs to ensure that current conditions are maintained on the Area. Implementation of MPE on Area 4 would involve the installation of approximately 20 MPE wells. Because it applies a high vacuum, MPE results in rapid LNAPL extraction rates and is included as an aggressive approach to remove LNAPL from the subsurface. It is anticipated that approximately 2 years would be required to reach the LNAPL remedial goals in Alternative 4-1. Chemical oxidation of LNAPL-impacted soils is not feasible until LNAPL has been removed, to the extent practicable; therefore it would not be implemented in areas impacted by LNAPL until the LNAPL remedial goals have been reached for the Area. Further, it is assumed that chemical oxidation of LNAPLimpacted soils would not be conducted beneath the buildings on Area 4. Chemical oxidation of TSM-impacted soils on the northern portion of Area 4 may be implemented once access has been obtained, as NAPL has not been observed in this portion of the Area. Chemical oxidation of the TSM and LNAPL-impacted soils on Area 4 is anticipated to require approximately 1 to 2 years.



#### Alternative 4-2: LNAPL recovery using belt-skimmers and AUL

Alternative 4-2 involves the installation of approximately 13 LNAPL extraction wells equipped with belt-skimmers and an AUL to ensure that current property conditions are maintained on the Area. Based on the estimated volume of LNAPL on Area 4 and anticipated rates of LNAPL recovery, it is estimated that the LNAPL remedial goals may be attained in approximately 3 to 5 years. However, it is considered likely that LNAPL may exist beneath the southern building on Area 2; therefore attainment of the LNAPL remedial goals may not be attainable on Area 4. This alternative does not include in-situ chemical oxidation, as this technology is not likely to be feasible beneath the buildings on the Area.

Alternative 4-3: LNAPL monitoring and AUL

Alternative 4-3 is a monitoring alternative; this alternative proposes monitoring to ensure that the extent of LNAPL impacts does not expand.

#### 8.06 Detailed Evaluation of Remedial Action Alternatives for Area 4

This section presents a Detailed Evaluation of the three Remedial Action Alternatives presented above, based on the criteria listed in Section 4.03. As with the other Areas, this evaluation is based on the current property conditions and use, as described in Section 2.06 and summarized in Section 8.02. A summary of the detailed evaluation of the three Remedial Action Alternatives developed for Area 4 are presented in Table XXII. Table XXII presents a summary of the likely effectiveness, reliability, implementability, cost (net present value, or NPV), risks, benefits, timeliness, and potential effect on non-pecuniary interests of each alternative. The presence of the existing buildings currently located on Area 4 limits the implementability and effectiveness of some remedial alternatives for Area 4. Because it is likely that LNAPL and soil containing contaminant concentrations greater than UCLs are located beneath buildings, a Permanent Solution is not proposed for Area 4; the remedial alternatives evaluated below represent Temporary Solutions. A comparison of the three remedial alternatives based on these criteria is presented below.

 Effectiveness: Because Alternative 4-1 includes LNAPL extraction using MPE, Alternative 4-1 is likely to provide more effective and more rapid LNAPL removal than Alternative 4-2. MPE is generally more effective in low-permeability soils; the effectiveness of MPE on Area 4 will therefore depend on the permeability of soils within the LNAPL area. However, it is unlikely that either Alternatives 4-1 or 4-2 would reach the LNAPL remedial goals, because LNAPL would likely remain beneath the buildings located on Area 4. Additionally, it is likely that contaminated soil, potentially containing soil contamination in excess of UCLs, exists beneath the buildings on Area 4, and would also remain beneath the building under these alternatives. Therefore, neither Alternative 4-1 nor 4-2 would reach a Permanent Solution. Because Alternatives 4-1 and 4-2 would reduce the volume of LNAPL on Area 4 but would only achieve a Temporary Solution, both are ranked with a

"moderate" level of effectiveness. Alternative 4-3 also would not provide a

Permanent Solution, as it is a monitoring-only alternative, but would not reduce the



volume of TSM or LNAPL on the Area. Therefore Alternative 4-3 is given a "low" effectiveness ranking. Each of the evaluated alternatives represents a Temporary Solution for Area 4.

÷

- Short and Long Term Reliability: The technologies proposed in Alternatives 4-1 (MPE and in-situ chemical oxidation) and 4-2 (belt-skimmers) are considered reliable remedial technologies. Both alternatives are considered moderately reliable in the short and long term, although MPE is considered a slightly more reliable method of LNAPL extraction than belt-skimmers because MPE is a more aggressive system and has been used successfully at other sites for LNAPL removal. The reliability criterion is not relevant for Alternative 4-3, as a remedial system is not proposed in this Alternative.
- Implementability: Alternative 4-1 is considered moderately difficult to implement, because the proposed MPE system would require approximately 20 MPE wells, extraction equipment and a groundwater and vapor treatment system on Area 4. Additionally, 35 chemical oxidation wells would be required in Alternative 4-1. Remedial facilities such as these would disrupt the current use of the Area. Alternative 4-2 is readily implementable; the belt-skimmers included in Alternative 4-2 would involve considerably less intrusion than MPE into the commercial operations on the Area, as significantly less space and equipment are required to implement this remedial component. Additionally, Alternative 4-2 does not include in-situ chemical oxidation, which is not fully implementable due to the presence of the building on Area 4. Alternative 4-3, which includes monitoring of LNAPL on Area 4, is readily implementable.
- Cost: The estimated NPV for Alternative 4-1, which includes MPE and chemical oxidation of TSM and LNAPL-impacted soils is approximately \$1.3M. The NPV for Alternative 4-2 is significantly lower, \$510K, as it includes a less aggressive approach to the removal of LNAPL (i.e., belt-skimmers) and does not include chemical oxidation of TSM and LNAPL-impacted soils. The estimated NPV for the monitoring alternative 4-3 is approximately \$100K.
- Risks: Moderate to low risks are associated with the three alternatives evaluated for Area 4. Moderate risks associated with Alternative 4-1 include those associated with handling of extracted LNAPL, and handling the caustic oxidants during in-situ chemical oxidation. Risk associated with implementation of Alternative 4-2 is somewhat lower than in Alternative 4-1, and includes the handling of extracted LNAPL only. For alternatives 4-1 and 4-2, in which LNAPL extraction is proposed, measures would be required to ensure that people on the property (including employees, medical patients or other visitors) do not come into contact with extraction equipment or extracted LNAPL. Alternative 4-3 involves few risks of exposure to hazardous chemicals.

Long-term risks associated with contamination left in place are highest for Alternatives 3-2 and 3-3; Alternative 3-3 does not include removal of LNAPL from the subsurface, and neither Alternative 3-2 nor Alternative 3-3 includes measures to



reduce concentrations of contaminants in soil. Long-term risks are lowest in Alternative 3-1, which includes in-situ chemical oxidation to reduce contaminant concentrations in accessible soils.

- Benefits: Alternatives 4-1 and 4-2 both provide the benefit of reducing the quantity of LNAPL on the Area. The completeness of the removal of LNAPL is dependent on the ability of LNAPL extraction wells adjacent to the building to remove DNAPL from beneath the building. As mentioned above under "Effectiveness", MPE (proposed in Alternative 4-1) may be more effective at reducing the quantity of LNAPL beneath the buildings on Area 4. Alternative 4-1 also includes some additional benefit due to the in-situ chemical oxidation component, which would reduce contaminant concentrations in accessible soil. However, soil beneath the building would not be accessible for chemical oxidation treatment. Alternative 4-3 produces little beneficial reduction of contaminants, as it includes neither LNAPL extraction nor soil remediation.
- Timeliness: LNAPL removal using MPE in Alternative 4-1 is likely to require approximately 2 years, which would be more rapid than removal using belt-skimmers in Alternative 4-2, which is anticipated to require approximately 3 to 5 years. However, it is unlikely that the LNAPL extraction wells proposed in Alternatives 2-1 and 2-2 will accomplish the LNAPL remedial goals, as it is believed that a portion of the LNAPL on Area 4 is located beneath the buildings on Area 4. Chemical oxidation of TSM and LNAPL-impacted soil, which is proposed as a remedial component in Alternative 4-1, would not be implementable until LNAPL quantity has been reduced to the extent practicable. In-situ chemical oxidation is anticipated to require approximately 2 years to reach the soil remedial goals; however soils beneath the buildings are not accessible for chemical oxidation treatment. Therefore the remedial goals will not be met for soils beneath the building. Alternative 4-3 could be implemented immediately and would continue for the foreseeable future, until a Permanent Solution can be implemented.
- Effect on Non-pecuniary Interests: Alternative 4-1 is likely to have greater adverse effects on non-pecuniary interests, because the MPE system and chemical oxidation injection proposed in this alternative would require more equipment and would have a greater impact on property use than Alternative 4-2. Both Alternatives 4-1 and 4-2 would require the construction of a small NAPL storage building on the Area. It is not known whether the construction of this building on Area 4 would be acceptable to the property owner. Alternative 4-3 would have very little effect on use of the Area or aesthetics.

#### 8.07 Selected Remedial Action Alternative for Area 4

In summary, although Alternative 4-1 may be slightly more effective and reliable and provide marginal additional benefit over Alternative 4-2, neither alternative would provide a likely Permanent Solution for Area 4. Implementation of Alternative 4-1 would likely provide slightly more effective, reliable, and rapid LNAPL removal than Alternative 4-2, and would also provide additional benefit of reduced contaminant concentrations in accessible soil.



However, Alternative 4-1 would not reach a Permanent Solution because LNAPL and contaminated soil would likely remain beneath the buildings located on Area 4. The belt-skimmers proposed for LNAPL extraction in Alternative 4-2 are considered to be a relatively reliable technology, and would provide effective LNAPL extraction over time. Like Alternative 4-1, Alternative 4-2 would not result in a Permanent Solution. Alternative 4-3, which involves monitoring of LNAPL and no contaminant removal or reduction, also would not result in a Permanent Solution. Therefore Alternative 4-3, which does not include an LNAPL extraction component, is not a recommended alternative. Both Alternatives 4-1 and 4-2 would result in a Temporary Solution due to the contamination that may remain beneath the buildings on the Area. Therefore, because of the implementation difficulties, impact on use and higher cost of Alternative 4-1, Alternative 4-2 is the selected remedy for Area 4.



ē.,

# IX. EVALUATION OF REMEDIAL ALTERNATIVES FOR AREA 5

# 9.01 Introduction

Brown and Caldwell conducted a focused evaluation of remedial alternatives for Area 5. This section presents a summary of Brown and Caldwell's Phase III RAP for Area 5. The Brown and Caldwell evaluation, which is included in Appendix E of this Report, includes a statement of Remedial Action Objectives, an Identification and Initial Screening of Technologies and Development of Remedial Alternatives, a Detailed Evaluation of Remedial Action Alternatives, and Selection of a Remedial Action Alternative for Area 5.

-

#### 9.02 Remedial Action Objectives

#### A. Risk Characterization Summary

The Human Health Risk Characterization satisfied a condition of No Significant Risk for current and future exposure pathways on Area 5. A condition of No Significant Risk to Public Welfare and the environment was also satisfied.

Although a condition of No Significant Risk has been satisfied for the indoor air inhalation pathway, the level of potential risk presented by the elevated levels of VOCs is approximately equivalent to the MADEP criteria at which a condition of No Significant Risk cannot be satisfied. Additionally, the data used in the Risk Characterization were collected during operation of a pilot-scale sub-slab ventilation system. Therefore, an evaluation of remedial alternatives to address indoor air was conducted.

#### B. Remedial Action Objectives

Remedial alternatives were developed to mitigate potential risk due to volatilization of benzene and naphthalene into indoor air. This may be accomplished through reduction of contaminant concentrations in groundwater and/or controlling vapor migration. MEC is conducting an evaluation of the HVAC system inside the building to assess the feasibility of making modifications to the existing system. The evaluation is being conducted to assess whether implementation of modifications to the existing system or installation of a new system is a feasible measure to control the infiltration of vapors from the subgrade. Therefore, the Brown and Caldwell evaluation did not include remedial alternatives for the HVAC system.

# 9.03 Identification, Initial Screening, and Development of Remedial Action Alternatives

#### A. Identification and Screening of Technologies

The Brown and Caldwell evaluation recommends the use of horizontal directional drilling (HDD) technology to install remedial measures on Area 5. HDD would



involve installation of horizontal wells or vapor extraction systems beneath the building. The evaluation states that a thorough utility location effort would be required prior to HDD installation and information regarding the building foundation system would be necessary. If this information is not available, a field investigation is recommended. The evaluation also states that if historic MGP structures shown on historic maps are encountered during drilling operations, the well installation layout may have to be revised. Brown and Caldwell stated that it may be possible to drill through the former MGP structures, depending on the condition and types of material used for construction. Finally, the Brown and Caldwell evaluation stated that the presence of the structures may provide preferential pathways for air or oxidants, and may reduce treatment efficiencies.

Remedial measures evaluated in the Brown and Caldwell evaluation would be installed using HDD methods. The remedial technologies that met the initial screening criteria include the following:

- Air Sparging: Injection of air into groundwater at relatively high flow rates to promote volatilization and biodegradation of VOCs in groundwater.
- Biosparging: Injection of air or oxygen into groundwater at relatively low flow rates to promote biodegradation of VOCs in groundwater.
- Chemical Oxidation using Ozone: Chemical conversion of hazardous constituents to less toxic compounds that are more stable, less mobile or inert. Ozone would be generated on-site and may be sparged or dissolved in water and injected into the subsurface.

#### B. Development of Remedial Action Alternatives

Each of the technologies listed above were identified by Brown and Caldwell as potential remedial alternatives for the treatment of groundwater on Area 5. It was assumed that a SVE system would be installed in each remedial alternative as a vapor management system to control vapor migration into the facility. The following are the remedial action alternatives evaluated by Brown and Caldwell:

- Alt-1: No Further Action This alternative would include no further remedial measures to address groundwater and soil vapor impacts on Area 5. This alternative includes groundwater monitoring at Area 5.
- Alt-2: Air Sparging with SVE This alternative would include horizontal air sparging wells spaced at 30-ft intervals beneath the building and an SVE system to control migration of vapors. The estimated air sparge flow rate would be 750 scfm; the SVE system is described below.
- Alt-3: Biosparging with SVE



This alternative would include biosparging wells spaced at 30-ft intervals beneath the building and an SVE system to control migration of vapors. The estimated biosparge rate would be 375 scfm.

- Alt-4: Chemical Oxidation using Ozone with SVE This alternative would include application of ozone similar to an air sparging system or a biosparging system, with an SVE system to control migration of vapors.
- SVE Component: An SVE system would be installed using horizontal wells in the unsaturated zone in Alt-2, Alt-3, and Alt-4 to capture vapors that may be migrating into indoor air. SVE extraction systems generally have extraction rates 1.25 to 5 times greater than the biosparging rate.

# 9.04 Detailed Evaluation of Remedial Alternatives

The following is a summary of the detailed evaluation of the remedial alternatives presented by Brown and Caldwell.

- Effectiveness: Alt-1 would not achieve a Permanent Solution. Alt-2, Alt-3, and Alt-4 would effectively treat the contaminants and achieve a Permanent Solution, however the timeframe in which this would be accomplished is uncertain because areas of elevated concentrations are not well defined and cannot be directly assessed.
- Reliability: Alt-1 would not successfully achieve the remedial action objectives. For the remaining alternatives, Brown and Caldwell stated that the reliability of the treatment mechanisms has been shown to be effective for many sites with the types of contaminants on Area 5. For this application, the reliability of the alternatives would be more dependent on the ability of the horizontal well injection system to deliver the air, nutrients or ozone to the treatment area.
- Difficulty in Implementation: The horizontal well system would have to be designed to not interfere with subgrade utilities, the foundation structure of the building and possibly historic MGP structures. Field testing prior to the design would be necessary in order to determine well spacing. Monitoring would also be required to assess the effectiveness of the alternative. There is sufficient availability of services, materials, equipment and specialists for implementation of the alternatives, and operation of the alternatives could be integrated with existing facility operations and conditions. The horizontal well system in Alt-2, Alt-3, and Alt-4 would have to be designed to not interfere with subgrade utilities and the foundation structure of the building, and well spacing would need to be calculated based on field studies.
- Costs: Capital installation costs for the alternatives are as follows: Alt-1, \$0; Alt-2, \$900,000; Alt-3, \$840,000; Alt-4, \$1,100,000. The estimated total present net worth of monitoring costs for the alternatives are as follows: Alt-1, \$440,000; Alt-2, \$100,000; Alt-3, \$130,000; Alt-4, \$100,000.



- Risks: Alt-1 does not have short-term risks associated with implementation, however long-term risks are associated with leaving contamination in place. Short-term risks in Alt-2, Alt-3, and Alt-4 would be the limited potential exposures to soils from cuttings during well installation. For Alt-4, there is the potential for exposures to ozone gas, should leakage occur.
- Benefits: Alt-1 does not reduce concentrations in a reasonable timeframe. Alt-2, Alt-3, and Alt-4 would lower the concentrations of contaminants in groundwater and therefore would enhance the restoration of Area 5. Implementation of the alternatives would not affect facility operations because equipment could be located in a non-obtrusive location.
- Timeliness: Alt-1 would require the longest time frame to achieve the remedial objectives. Alt-2 (air sparging) and Alt-4 (chemical oxidation with ozone) would likely require the shortest time frame, approximately 3 years. Alt-3 (biosparging) would likely require approximately 5 years.
- Non-pecuniary interests: Alt-1 would not have any effect on non-pecuniary interests, The remaining alternatives would require installation of horizontal wells and an equipment building to house equipment. Monitoring activities would be required as well.

# 9.05 Selection of a Remedial Action Alternative

The following is a summary of Brown and Caldwell's comparison of the alternatives evaluated in the detailed evaluation, feasibility evaluation and the recommended alternative to address ground water and soil impacts on Area 5.

# A. Comparison of Alternatives

Alt-1 was rated the least favorable alternative and not considered a viable alternative. The remaining alternatives were rated favorably for this application. Costs are comparable, and implementation of any of the alternatives would probably result in successful reduction of COCs. Alt-4 (chemical oxidation) requires more intense training and higher level field personnel due to the complexity of the equipment involved. Alt-2 (air sparging) may accomplish volatilization of benzene, but volatilization of naphthalene may not be easy to accomplish. Alt-2 also involves higher operational costs, and has a higher potential for migration of VOCs into the facility. Therefore, Alt-3 was favored over the other alternatives.

#### **B.** Feasibility of Implementing a Permanent Solution

The recommended remedial alternative is intended to reduce constituents in indoor air. However, Brown and Caldwell stated that since areas of elevated concentrations in the soil that are most likely contributing to groundwater and indoor air contamination has not been identified, there is uncertainty as to the timeframe that



would be required to reduce the concentrations to a level where No Significant Risk could be demonstrated.

#### C. Feasibility of Achieving or Approaching Background

The Brown and Caldwell evaluation states that because areas of elevated concentrations in soil that are most likely contributing to groundwater contamination have not been definitively located, reduction of groundwater and soil concentrations to background cannot be expected and is not probable.

# D. Feasibility of Reducing Concentrations in Soil to Below UCLs

Phase II Investigations for Area 5 did not detect UCLs in soil. However, the areas of elevated concentrations that may be contributing to groundwater contamination has not been definitively located, and further investigation is not feasible. It is possible that constituents are present at concentrations above UCLs. However, without knowing the location of the elevated concentrations, the feasibility of reducing the concentrations of to below UCLs is not predictable.

# E. Selection of Alternatives

Brown and Caldwell selected remedial alternative Alt-3, biosparging with SVE, installed in horizontal wells beneath the building using HDD techniques. The conceptual design includes three horizontal delivery wells for biosparging, connected to a manifold and blower station located on the southwest corner of the property. The SVE system would include installation of four horizontal SVE wells and six vertical extraction wells located in front of the building. The SVE blower and vapor treatment system would be located in a shed on-site. The SVE system should operate at an air flow rate approximately four times greater than the air flow rate of the biosparging system. Groundwater monitoring would be conducted on a quarterly basis at the site for an estimated five years. Implementation and accomplishment of the remedial goals is anticipated to require approximately six to seven years.

Field tests would be required to obtain specific site information to determine the horizontal well spacing. Additionally, Brown and Caldwell stated that while the HDD drilling assembly can be steered to predetermined points in both the vertical and horizontal plane, it should be noted that a thorough utility location effort is required and also, information regarding the foundation system is necessary. If this information is not available, a field investigation to obtain as much site information as possible would be beneficial.

As previously mentioned in Section 9.02B, MEC is conducting an evaluation of the HVAC system for the 129 Commercial Street facility, and therefore an evaluation of the installation of a new HVAC system or modifications to the existing system were not included in the Brown and Caldwell evaluation. If, however, site conditions or other parameters interfere with the successful installation of a horizontal well system that would meet the design requirements of the selected remedial alternative, then



further effort will be directed toward the consideration of the implementation of an HVAC system for the 129 Commercial Street facility that would minimize the infiltration of vapors from the subgrade.



# X. SUMMARY OF SELECTED REMEDIAL ACTION ALTERNATIVES

#### 10.01 Fhase III RAP Overview

This Phase III RAP has presented an evaluation of remedial alternatives for the former Malden MGP Site. As required by 310 CMR 40.0859(1), a Site-wide remedial action alternative that addresses the various impacted media at the Site has been selected based on the detailed evaluation criteria listed in 310 CMR 40.0858. This alternative is described below in Section 10.03. Based on the evaluation contained in this Phase III RAP, a Permanent Solution for the Site is not attainable at the present time, and a Temporary Solution (i.e., Class C RAO) is suitable and timely for the Site. The remedial action alternative proposed in this section is intended to meet the requirements of the MCP, while allowing current use of the properties on the Site to be maintained. This section presents a summary of Site conditions, evaluates the feasibility of achieving a Permanent Solution remedial alternative, and describes the anticipated approach for implementation of the proposed remedy. Justification for the selection of a Temporary Solution is presented, including a list of definitive and enterprising steps toward identifying and achieving a Permanent Solution in the future.

For the purposes of this Phase III RAP, the former Malden MGP Site has been divided into five Remedial Action Alternative Areas, the limits of which are shown on Figure 15. Contamination identified at the Site has been categorized into six types: TSM, shallow DNAPL, deep DNAPL, LNAPL, BTEXSN in soil and groundwater, and petroleum-impacted Soil. The Site Risk Characterization (as revised, refer to Appendix B) was conducted to identify exposure pathways for which a condition of No Significant Risk could not be satisfied. Remedial alternatives were developed and evaluated in this Phase III RAP to address contamination in media that contribute to the exposure pathways identified in the Risk. Characterization. A summary of the media and types of contamination for which remedial alternatives were developed on each Site Area is presented below in Table XXIII.



#### TABLE XXIII

# SUMMAFLY OF REMEDIAL ACTION ALTERNATIVE AREAS AND ASSOCIATED IMPACTED MEDIA

Site Area	Area Description	Properties Located Within The Area	Types Of Contamination Identified	Media For Which Remedial Alternatives Were Evaluated <sup>(2)</sup>
AREA 1	Northern portion of Parcel E	100 Commercial Street	TSM Shallow DNAPL Deep DNAPL <sup>(1)</sup> LNAPL <sup>(1)</sup> Petroleum- impacted soil	Soil
AREA 2	Southern portion of Parcel E	100 Commercial Street	TSM Shallow DNAPL Deep DNAPL LNAPL	Soil
AREA 3	Northern portion of Parcel A	51 Commercial Street 65 Commercial Street 77 Commercial Street 89 Commercial Street	TSM Shallow DNAPL	Soil
AREA 4	Southern portion of Parcel A	99-109 Commercial Street	TSM LNAPL	Soil
AREA 5	Parcel B	129 Commercial Street	BTEXSN in soil and groundwater	Soil Groundwater Indoor Air

Notes:

 LNAPL and deep DNAPL have been identified in monitoring wells in the past on Area 1; however recent monitoring has not observed these contaminants on Area 1.

2. Although groundwater contains elevated concentrations of contaminants, remediation of groundwater on Areas 1 through 4 is not required to achieve a condition of No Significant Risk. Groundwater is not used on these Areas as a resource (i.e., for drinking water or industrial use), and based on groundwater sampling data obtained during the Phase II investigation, plumes of groundwater contamination do not appear to be leaving the Site. Additionally, it is anticipated that remedial approaches to remediate LNAPL, DNAPL and soil will have a beneficial effect on groundwater quality.

# 10.02 Feasibility of Achieving a Permanent Solution or Background Concentrations

To achieve a Permanent Solution for the Site, a condition of No Significant Risk must be satisfied for the Site, including elimination of soil UCL exceedences through treatment or installation of an engineered barrier, and all source areas must be controlled or eliminated. At the former Malden MGP Site, this would involve the following:



- Removal of LNAPL and DNAPL present in thickness greater than <sup>1</sup>/<sub>2</sub> in. from the Site subsurface
- Elimination of soil material in which UCL exceedences have been observed, including contaminated soil and tar-saturated material (TSM), through either excavation, treatment, or placement of an engineered barrier to eliminate potential exposure pathways to UCL exceedences.
- Remediation of soil with contaminant concentrations that preclude satisfaction of a condition of No Significant Risk, or mitigation of potential risk through AULs

At the former Malden MGP Site, attainment of the NAPL remedial goal listed above is not reasonably feasible in the near term. DNAPL has been identified in monitoring wells on Areas 1, 2 and 3, and LNAPL has been identified on Areas 2 and 4. Based on NAPL thickness measurements, the total volume of DNAPL is estimated to be approximately 13,000 to 22,000 gallons, and the total LNAPL volume is estimated to be approximately 8,000 to 11,000 gallons. Based on the performance of the DNAPL extraction well operating on Area 2 since October 2001, DNAPL extraction is feasible at the Site, but a significant time period (approximately 10 years) will be required to reach the DNAPL remedial goals on each Site Area. Likewise, removal of LNAPL from the subsurface may be feasible at the Site, however LNAPL extraction typically requires an extended period of time as well. Furthermore, it is likely that DNAPL and LNAPL may be located beneath buildings on Areas 2, 3 and 4, and therefore may not be extractable under current site conditions.

As indicated above, achievement of a Permanent Solution would also require elimination of the potential for exposure to UCL exceedences at the Site. Exceedences of UCLs in soil are considered to exist within the limits of TSM contamination, which has been observed on Areas 1 through 4. Achievement of a Permanent Solution through excavation of soil contamination is not reasonably feasible because a significant portion of this contamination is overlain by buildings that are currently occupied by operating businesses. It is considered likely that TSM contamination exists beneath buildings located on Areas 2, 3, and potentially Area 4, and the soil contamination on Area 5 is located beneath a building. Demolition of the KeySpan Maintenance Garage would be required to excavate petroleum-impacted soils on Area 1. Likewise, installation of an Engineered Barrier would not reach a Permanent Solution at the Site, because the presence of the buildings on top of soils contaminated at levels greater than the UCLs does not eliminate potential exposure pathways, according to the MCP. The buildings on the Site would not satisfy MADEP requirements for an Engineered Barrier, and because the barrier must be contiguous across contaminated areas in order to remove potential exposure pathways to UCL exceedences, the presence of the buildings on the Site would render the barrier ineffective. In-situ remediation methods for soil would not be effective until NAPL-phase contaminants have been removed from the subsurface, and are likely not implementable beneath the buildings on the Site. Therefore, reduction of contaminant concentrations on the Site to less than the UCLs is not feasible at this time and a Permanent Solution for the Site is not reasonably feasible at present.

As described above, based on the detailed evaluation conducted in this Phase III RAP, a feasible Permanent Solution does not currently exist for the Site, and a Temporary Solution is proposed. Because a Permanent Solution cannot be attained, background concentrations of



contaminants in soil and groundwater at the Site cannot be reached at the present time. The presence of contaminants beneath buildings and NAPL in the subsurface precludes restoration of soil and groundwater to background concentrations. Therefore, reduction of contaminant concentrations to background concentrations is not an attainable goal at present.

# 10.03 Selected Remedial Action Alternative

#### A. Selected Remedial Solution Components

The components of the proposed remedial alternative, which were selected in the evaluations for the Areas presented in Sections V through IX, are intended to result in progress toward the Site remedial goals. Specifically, the proposed remedy is intended to reduce the quantity of DNAPL and LNAPL in the subsurface at the Site, and reduce VOC concentrations in soil, groundwater and indoor air that contribute to exposure pathways for which a condition of No Significant Risk could not be satisfied. The remedial action alternatives selected for Areas 1 through 5 are listed below.

- Area 1: Alternative 1-4; Shallow DNAPL extraction using approximately five DNAPL extraction wells, shallow DNAPL migration control (if necessary), in-situ chemical oxidation of TSM and petroleum-impacted soils (once NAPL quantity has been sufficiently reduced), and an AUL to prevent exposure to contaminated soils. Wells in which LNAPL and deep DNAPL have been detected will be monitored for NAPL during implementation of the remedy.
- Area 2: Alternative 2-2; Shallow and deep DNAPL extraction using approximately 13 extraction wells, LNAPL extraction using approximately six wells equipped with belt-skimmers, shallow DNAPL migration control (if necessary), and AUL to prevent exposure to contaminated soils
- Area 3: Alternative 3-2; Shallow DNAPL extraction using approximately 12 DNAPL extraction wells, shallow DNAPL migration control (if necessary), and an AUL to prevent exposure to contaminated soils.
- Area 4: Alternative 4-2; LNAPL extraction using approximately 13 wells equipped with belt-skimmers and an AUL to prevent exposure to contaminated soils.
  - Area 5: The focused evaluation conducted by Brown and Caldwell selected Alt-3, biosparging and SVE, installed using HDD methods beneath the 129 Commercial Street building. This alternative includes three horizontal biosparging delivery wells, four horizontal SVE wells, and six vertical SVE wells in front of the building.
- With the exception of Area 5, impacted groundwater does not contribute to risk and therefore remedial alternatives were not developed specifically to reduce contaminant concentrations in groundwater. However, implementation.



of the selected remedy described above is likely to have a positive impact on groundwater concentrations over the long term. In order to monitor groundwater quality on the Site, implementation of a Site-wide groundwater monitoring program is recommended. This program is anticipated to involve the sampling of approximately 20 to 25 monitoring wells on an annual basis, to ensure that site conditions remain stable during the implementation of the Temporary Solution on the Site.

AULs may be implemented on properties located within the boundaries of the Site to mitigate potential exposure to contaminated soil. AULs may mandate the maintenance of a direct contact barrier (i.e., asphalt pavement) to prevent contact of Site employees, visitors and trespassers with contaminated soil.

Approximate locations of remedial components (including DNAPL and LNAPL extraction wells and potential locations for DNAPL migration control barriers) are shown on Figure 20. The proposed remedial measures are anticipated to reduce the quantity of LNAPL and DNAPL and reduce concentrations of OHM at the Site, and therefore reduce potential risk posed to human health and the environment by OHM at the Site. Due to the limitations discussed above, the selected remedy represents a Temporary Solution for the Site. The requirement for a Temporary Solution, as outlined in Section 3.03, is the elimination of Site Substantial Hazards. Because the Site Risk Characterization satisfied a condition of No Substantial Hazard for the terrestrial portion of the Site, this requirement has already been met for Areas 1 through 5.

Although not a required component of the proposed Temporary Solution, AULs may be instituted on selected properties as a risk reduction measure to reduce the potential for exposure to MGP residuals. Based on DEP guidance<sup>1</sup>, in circumstances "where a Temporary Solution is expected to be in place for a long period of time, an AUL may be useful as a means of providing notice of the residual contamination to future holders of an interest in property at the disposal site." The AULs that may be placed on Site properties would require maintenance of a direct contact barrier (e.g., maintenance of the existing asphalt surfaces) to mitigate potential exposure to contaminated soil. Additionally, AULs may be used to mandate health & safety procedures to protect on-site commercial/industrial facility workers and construction workers during work to mitigate exposure hazards associated with contaminated soil. Such health and safety procedures may include the use of personal protective equipment and/or engineering controls that would reduce exposure to contaminated soil. AULs may also require that current property use be maintained (i.e., prohibit changes to more sensitive property uses, such as residential use) and mandate soil management plans unless necessary measures, such as engineering controls, are implemented to preclude future exposures.

#### B. Definitive and Enterprising Steps

The selected remedial alternative for the Site constitutes a Temporary Solution, because the requirements of a Permanent Solution cannot be met at the present time.



"Definitive and enterprising steps", as described in 310 CMR 40.0861(h), are nemedial measures that represent progress toward a Permanent Solution. Several of the components of the selected remedy described above represent definitive and enterprising steps toward a Permanent Solution. The following "definitive and enterprising steps" are proposed:

- II DNAPL recovery using extraction wells on Areas 1, 2 and 3
- LNAPL recovery using belt-skimmers on Areas 2 and 4
- Reduction of BTEXSN concentrations in soil, groundwater and indoor air through the installation of biosparging and SVE systems on Area 5
- Review of the performance of the Temporary Solution every five years, and assess whether new or emerging technologies may be able to contribute further toward the achievement of a Permanent Solution
- Monitor performance and effectiveness of the selected remedial components.

# C. Anticipated Remedial Implementation Approach

The Temporary Solution components and definitive and enterprising steps listed in the Sections above are proposed to reduce concentrations of MGP residuals at the Site, and to progress toward accomplishment of these remedial objectives. The following key aspects have been identified as "critical path" elements of the remedial process, and are likely to dictate the schedule of the remedial implementation:

- Property access and property use: The Former Malden MGP Site is comprised of seven properties, which are owned by entities other than MEC. Therefore, implementation of the remedy must be coordinated with the needs of property owners. The properties located within the limits of the Site are occupied by operating businesses, which must be allowed to function during implementation of the remedy.
- Appropriate sequencing of remedial technologies: On Areas 1, 2, 3 and 4, the quantity of NAPL in the subsurface must be reduced to the extent practicable prior to the remediation of TSM and soil. Remediation of TSM or soil in the presence of LNAPL or DNAPL would be ineffective.
- 3. Other technical considerations: Technical factors regarding the effectiveness of the proposed remedial components will be taken into consideration during implementation of the remedy. For example, proper location of DNAPL extraction wells is dependent upon the elevation of the organic deposit, which has been observed to be highly irregular during subsurface investigations. Based on these observations, DNAPL extraction wells may be installed in stages, such that the locations of the second stage of DNAPL extraction wells would be determined based on the results obtained in the first stage. The most efficient placement of DNAPL or LNAPL extraction wells may result in the installation of extraction wells on one property before another.



4. Provisional installation of shallow DNAPL migration control barriers: Installation of vertical shallow DNAPL migration control barriers may be implemented in the event that certain conditions are encountered. In the event that the DNAPL remedial goals are accomplished on one Area before they are accomplished on adjacent Areas, a DNAPL migration control barrier may be installed if the potential exists for the migration of shallow DNAPL from the adjacent Areas. This remedial component is provisional and may not be required if the DNAPL remedial goals are attained in a similar timeframe on Areas 1, 2 and 3, or if the potential for DNAPL migration from one Area to another is considered small, based on the results of DNAPL

-

The proposed remedy for the Site will be implemented in accordance with and consideration of these critical elements. Based on the complexity of the contamination and ownership situation of the Site, it is likely that a phased approach to Site remediation will be followed. Components of the proposed remedy will be implemented as they become feasible in the overall Site remedial plan.

# D. Estimated Net Present Value of the Selected Remedial Alternative

The estimated net present value of the selected remedial alternative, as summarized in Table XXIV, is approximately \$4.95M. Assumed time requirements for the selected remedial components are also listed in Table XXIV.

# 10.04 Justification of Temporary Solution

A Permanent Solution for the Site cannot be reached at this time, because hazardous materials such as PAHs, VOCs, and TSM are present in soil at concentrations greater than MADEP UCLs and NAPL exists in several monitoring wells at a thickness greater than 0.5 in. As discussed in Section 10.02 above, removal through excavation, placement of an engineered barrier or remediation of these materials is not feasible at this time due to the current property use. In the event that there are future changes in property use, remedial technologies can be re-evaluated to assess the feasibility of a Permanent Solution. As part of the Temporary Solution for the Site, definitive and enterprising steps will be taken to (1) contain, and as possible remove DNAPL and LNAPL from the subsurface on the Site and (2) reduce VOC concentrations in groundwater and indoor air on the 129 Commercial Street property.



#### REFERENCES

- MADEP/BWSC MCP Master Q&A, dated 25 March 1999. No. 185, p. 56.
- Haley & Aldrich, Inc., "Report on Phase II Comprehensive Site Assessment, Former Malden MGP Site, Malden, Massachusetts", dated 28 December 2001
- Haley & Aldrich, Inc., Tier II Extension Submittals, Former Malden Manufactured Gas Plant, dated 17 February 1995, 16 April 1996, 6 February 1997, 19 February 1998, and 17 February 1999
- Haley & Aldrich, Inc., "Tier Reclassification Submittal and Tier 1A Permit Application, Former Malden MGP, Malden, Massachusetts", dated 20 August 1999
- Haley & Aldrich, Inc., "Release Abatement Measure Plan, Former Manufactured Gas Site, Parcel B, 129 Commercial Street, Malden, Massachusetts, RTN-3-0362 and Linked RTNs 3-3757, 3-13310, and 3-13345", dated 2 July 1998
- Haley & Aldrich, Inc., "Modification to Release Abatement Measure Plan, Former Malden MGP Site, Parcel B, 129 Commercial Street, Malden, Massachusetts", dated 9 April 1999
- Haley & Aldrich, Inc., "Release Abatement Measure Status Report", Nos. 1 through 10, RTN 3-0362 and linked RTN 3-3757, dated 22 January 1999 through 7 April 2003
- Haley & Aldrich, Inc., "Immediate Response Action Plan Update, RTN 3-0362 & 3-13754, West End Brook & Malden River Culverts", dated 12 March 1998
- Haley & Aldrich, Inc., "Immediate Response Action Status Report", Nos. 1 through 14, RTN 3-0362 & 3-13754, West End Brook & Malden River Culverts, dated 11 September 1996 through 12 March 2003.
- Haley & Aldrich, Inc., "Report on Oil and Hazardous Material Site Evaluation, Boston Gas Company, Commercial and Center Streets, Malden, Massachusetts", dated 26 August 1988.
- Haley & Aldrich, Inc., "Report on Oil and Hazardous Material Site Evaluation, Boston Gas Company, Commercial and Charles Streets, Malden, Massachusetts", dated 20 September 1988.
- Haley & Aldrich, Inc., "Report on Oil and Hazardous Material Site Evaluation and Conceptual Remedial Recommendation, Boston Gas Company, 100 Commercial Street, Malden, Massachusetts", Volumes I through III, dated 21 September 1988.

G:\06558\634 Phase III\Final\_Phase\_III\_Report\Malden\_MGP\_PhIII reviewed.doc



Page 1 of 4

Y

TABLE II SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 51 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

LOCATION ID	MCP	MCP	MCP	SS-101	101-SS	SS-102	SS-102	SS-103
Depth Range	METHOD 1	METHOD 1	METHOD 1	0 to 3 feet	0.5 to 1 feet	0 to 3 feet	0.5 to 1 feet	0 to 3 feet
Sample ID	RCS-1	RCS-2	ncr	SS-101 C1	<b>SS-101 SI</b>	SS-102 C1	SS-102 S1	SS-103 CI
Sample Date	(ug/kg)	(uğ/kg)	(10,/5n)	6/9/2002	6/9/2002	6/9/2002	6/9/2002	6/9/2002
VOCs (ug/kg)	0007	0000001	0000001		0001		0001	
Naphinalene (VUC)	000+	TODOOT	nonononi	;	noci	:	1900	1
p/m-Xylene	200000	50000	200000	:	ND(180)	1	300	:
SVOCs (ug/kg)			100 M					
1-Methylnaphthalene	N/A	NIA	N/A	ND(1100)	8	940	;	ND(570)
2-Methylnaphthalene	4000	100000	1000000	ND(1800)	1	980	:	ND(920)
Acenaphthylene	100000	100000	1000000	4200	ł	2400	;	ND(570)
Anthracene	1000000	1000000	1000000	3400	ł	2600	;	ND(570)
Benzo(a)anthracene	700	1000	100000	8100	;	0009	;	1100
Benzo(a)pyrene	200	700	100000	8300	:	6200	;	1100
Bcnzo(e)pyrene	N/A	N/A	N/A	6000	:	4200	;	760
Benzo(b)fluoranthene	700	1000	100000	7300	:	4900	:	920
Benzo(ghi)perylene	100000	2500000	1000000	6600	;	4600	1	760
Benzo(k)fluoranthene	7000	10000	400000	6800	1	4900	;	890
Chrysene	2000	10000	400000	8400	;	6300	;	1100
Dibenzo(a,h)anthracene	200	700	100000	0061	;	1300	3	ND(570)
Fluoranthene	1000000	1000000	1000000	11000	;	9500	:	1600
Fluorene	400000	200000	10000000	ND(1100)	;	800	;	ND(570)
Indeno(1,2,3-cd)pyrene	200	1000	100000	5800	:	4100	1	012
Naphthalene	4000	1000000	10000000	ND(1100)	:	1400	1	ND(570)
Perylene	NIA	N/A	N/A	2300	;	1700	:	ND(570)
Phenanthrene	100000	100000	10000000	4700	;	8400	1	820
Pyrenc	700000	2000000	10000000	14000	:	12000	:	1800

Page 2 of 4

TABLE II SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 51 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

LOCATION ID	MCP	MCP	MCP	SS-101	SS-101	SS-102	SS-102	SS-103
Depth Range	METHOD 1	METHOD 1	METHOD 1	0 to 3 feet	0.5 to 1 feet	0 to 3 feet	0.5 to 1 feet	0 to 3 feet
Sample ID	RCS-1	RCS-2	ncr	SS-101 C1	SS-101 S1	SS-102 C1	SS-102 S1	SS-103 C1
Sample Date	(ug/kg)	(ug/kg)	(ug/kg)	6/9/2002	6/9/2002	6/9/2002	6/9/2002	6/9/2002
Metals (ug/kg)								
Arsenic	30000	30000	300000	0091	į	9500	;	8900
Barium	1000000	2500000	1000000	72000	:	53000	1	62000
Cadmium	30000	80000	800000	850	;	ND(460)	;	470
Chromium	100000	250000	1000000	38000	1	17000	;	16000
Lead	300000	600000	6000000	300000	:	340000	;	30000
Mercury	20000	00009	000009	410	:	380	ł	250
Cyanide								
Cyanide, Total	100000	100000	4000000	5400	1	6500	;	1600
Cyanide, Physiologically Available				870	:	1900	1	400
EPH (ug/kg)								
C9-C18 Aliphatics	100000	2500000	2000000	ND(102000)	;	65400	:	ND(34500)
C19-C36 Aliphatics	250000	500000	2000000	155000	;	151000	ŝ	122000
C11-C22 Aromatics	200	200000	1000000	542000	:	1730000		191000

# Notes:

1. ND (5): Compound not detected above detection limit;

detection limit is shown in parentheses.

 VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

3. Table lists detected compounds only; compounds

that were not detected are not listed.

Page 3 of 4

TABLE II SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 51 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

Depth Range         METHOD 1         No 1 feet         0.5 to 1 feet	LOCATION ID	MCP	MCP	MCP	SS-103	SS-103	SS-103
Sample ID         RCS-1         RCS-1         RCS-2         UCL         SS-103 S1 DU         SS-103 S1 DU <th>Depth Range</th> <th>METHOD 1</th> <th>METHOD 1</th> <th>METHOD 1</th> <th>0 to 3 feet</th> <th>0.5 to 1 feet</th> <th>0.5 to 1 feet</th>	Depth Range	METHOD 1	METHOD 1	METHOD 1	0 to 3 feet	0.5 to 1 feet	0.5 to 1 feet
Scampte Date         (ug/kg)         (ug/kg)         (ug/kg)         (ug/kg)         (ug/kg)         (s/2002         6/9/2002         <	Sample ID	RCS-1	RCS-2	NCL	SS-103 CI DUP	SS-103 S1	SS-103 S1 DUP
VOCs (ng/kg)         VOCs (ng/kg)         ND(860)         ND(860)         ND(860)         ND(860)         ND(860)         ND(860)         ND(860)         ND(860)         ND(860)         ND(170)	Sample Date	(ug/kg)	(ug/kg)	(ug/kg)	6/9/2002	6/9/2002	6/9/2002
Naphthaltere (VOC)         4000         1000000         1000000          ND(890)         ND(800)           p(m-Xylene         50000         50000         50000         50000         50000          ND(80)         ND(80)           SVOC5 (ag/8g)         N/A         N/A         N/A         N/A         ND(1100)           ND(800)           ND(800)           ND(170)           ND(170)           ND(170)           ND(170)           ND(170)            ND(170)            ND(170)                           ND(170) </td <td>VOCs (ug/kg)</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	VOCs (ug/kg)						
p(in:Xylene         50000         5000         5000         50000	Naphthalene (VOC)	4000	1000000	10000000	:	ND(890)	ND(860)
SVOCs (ug/kg)         N/A         N/A <thn a<="" th=""> <thn a<="" th=""> <th< td=""><td>p/m-Xylene</td><td>50000</td><td>50000</td><td>\$0000</td><td>:</td><td>ND(180)</td><td>ND(170)</td></th<></thn></thn>	p/m-Xylene	50000	50000	\$0000	:	ND(180)	ND(170)
1-Methylraphtralene         N/A	SVOCs (ug/kg)						
2-Methylraphthalene         400         100000         1000000         24000 <th< td=""><td>1-Methylnaphthalene</td><td>N/A</td><td>N/A</td><td>N/A</td><td>ND(1100)</td><td>;</td><td>:</td></th<>	1-Methylnaphthalene	N/A	N/A	N/A	ND(1100)	;	:
Accaraphthylene         100000         1000000         1000000         1000000         1000000         10011000	2-Methylnaphthalene	4000	0000001	10000000	ND(1800)	;	;
Antiracene         100000         100000         100000         100000         100000         2400         -	Acenaphthylene	100000	100000	10000000	ND(1100)	;	:
Berizo(a)antifracene         700         1000         10000         2400         -         <	Anthracene	100000	100000	1000000	ND(1100)	:	:
Berzo(a)pyrene         700         700         700         100000         2400	Benzo(a)anthracene	700	1000	100000	2400	;	:
Berrzo(c)pyrene         N/A         N/A         N/A         1700	Benzo(a)pyrene	700	700	100000	2400	;	:
Benzo(b)fluoranthene         700         1000         10000         1800             Benzo(ghi)perylene         1000000         2500000         100000         1700             Benzo(ghi)perylene         7000         10000         400000         1700             Benzo(k)fluoranthene         7000         10000         400000         2600             Chrysene         7000         10000         400000         2600              Dibenzo(s,h)anthracene         700         100000         1000000         1000000         2600             Fluoranthene         700         100000         1000000         1000000              Fluoranthene         700         1000000         1000000         1000000              Naphthalene         700         100000         1000000         1000000              Naphthalene         700         100000         1000000         1000000         1500             Perylene	Benzo(c)pyrene	NIA	N/A	N/A	1700	;	:
Berzo(ghi)perylene         1000000         2500000         10000         21000             Berzo(k)fluoranihene         7000         100000         400000         2100 <td< td=""><td>Benzo(b)fluoranthene</td><td>700</td><td>1000</td><td>100000</td><td>1800</td><td></td><td>;</td></td<>	Benzo(b)fluoranthene	700	1000	100000	1800		;
Benzo(k)fluoranthene         7000         10000         400000         2100             Chrysene         7000         10000         400000         2600              Chrysene         700         10000         400000         2600              Dibenzo(a,h)anthracene         700         1000000         1000000         1000000         3600             Fluoranthene         1000000         1000000         1000000         1000000         3600             Fluorene         700         100000         10000000         1000000         1500             Naphthalene         7/0         1000000         10000000         10000000         1500             Perylene         N/A         N/A         N/A         N/A         N/A             Perylene         1000000         10000000         10000000         10000000             Perylene         N/A         N/A         N/A         N/A             Prene         70000	Benzo(ghi)perylene	100000	2500000	10000000	1700	1	;
Chrystene         7000         10000         400000         2600	Benzo(k)fluoranthene	2000	10000	400000	2100	:	;
Diberzo(a,h)anthracene         700         700         100         700         100000         100000         1001100 <th< td=""><td>Chrysene</td><td>7000</td><td>10000</td><td>400000</td><td>2600</td><td>;</td><td>:</td></th<>	Chrysene	7000	10000	400000	2600	;	:
Fluoranthene         1000000         1000000         1000000         3600             Fluorene         100000         100000         1000000         100000         1500	Dibenzo(a,h)anthracene	700	700	100000	ND(1100)	;	:
Fluorene         40000         2000000         1000000         ND(1100)             Indeno(1,2,3-cd)pyrene         700         1000         100000         1500             Naphthalene         700         100000         1000000         1000000              Perylene         N/A         N/A         N/A         ND(1100)             Prevlene         1000000         1000000         10000000         2000             Pvreme         700000         1000000         10000000         10000000	Fluoranthene	100000	1000000	1000000	3600	;	;
Indeno(1,2,3-cd)pyrene         700         1000         100000         1500             Naphthalene         4000         1000000         1000000         1000000	Fluorenc	400000	2000000	1000000	ND(1100)	1	;
Naphthalene         4000         1000000         1000000         ND(1100)             Perylene         N/A         N/A         N/A         N/A         NO	Indeno(1,2,3-cd)pyrene	200	1000	100000	1500	:	:
Perylene         N/A         N/A         N/A         ND(1100)	Naphthalene	4000	1000000	10000000	ND(1100)	ł	;
Phenanthrene 100000 100000 1000000 2000	Pervlene	NIA	NIA	N/A	ND(1100)	;	:
Purente 700000 2000000 1000000 4400	Phenanthrene	100000	100000	10000000	2000	ł	;
	Pyrene	700000	2000000	10000001	4400	;	:

1

Haley & Aldrich, Inc. G (06558\634 Phase III\Phase III Tables.xls Page 4 of 4

I

1

TABLE II SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 51 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

LOCATION ID	MCP	MCP	MCP	SS-103	SS-103	SS-103
Depth Range	METHOD 1	METHOD 1	METHOD 1	0 to 3 feet	0.5 to 1 feet	0.5 to 1 feet
Sample ID	RCS-1	RCS-2	ncr	SS-103 C1 DUP	SS-103 S1	SS-103 S1 DUP
Sample Date	(ug/kg)	(ug/hE)	(ug/kg)	6/9/2002	6/9/2002	6/9/2002
Metals (ug/kg)						
Arsenic	30000	30000	300000	9500	;	;
Barium	1000000	2500000	10000000	65000	:	:
Cadmium	30000	80000	800000	510	:	:
Chromium	100000	250000	10000000	14000	;	:
Lead	300000	600000	6000000	330000	1	:
Mercury	20000	00009	000009	240	E	3
Cyanide						
Cyanide, Total	100000	100000	4000000	1300	;	:
Cyanide, Physiologically Available				450	:	;
EPH (ug/kg)						
C9-C18 Aliphatics	100000	250000	20000000	ND(11500)		;
C19-C36 Aliphatics	250000	500000	2000000	00018	1	:
C11-C22 Aromatics	200	200000	10000000	127000	:	:

Notes:

- 1. ND (5): Compound not detected above detection limit;
  - detection limit is shown in parentheses. 2. VOCs: Volatile Organic Compounds;
- SVOCs: Semivolatile Organic Compounds;
- Table lists detected compounds only; compounds that were not detected are not listed.

Page 1 of 8

I

ł

1

I

1

ł

1

1

TABLE III SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

LOCATION ID	MCP	MCP	MCP	02B-B917	028-8917	028-B918	028-8918	028-8919
SAMPLE ID	METHOD 1	METHOD 1	METHOD 1	02B-917-MW	02M-1001-MW	02B-918-MW	02B-918-MW/S10	02B-919-MW/S2
SAMPLE DATE	RCS-1	RCS-2	NCL	2/18/2002	2/18/2002	2/18/2002	2/19/2002	2/19/2002
DEPTH INTERVAL	(ug/kg)	(ug/kg)	(ug/kg)	2.6 to 4.6 ft	2.6 to 4.6 ft	14.2 to 16.2 ft	18.2 to 20.2 ft	3 to 5 ft
VOCs (ug/kg)								
1,2,4-Trimethylbenzene	1000001	1000000	NIA	ND(220)	ND(230)	ND(3500)	ND(350)	ND(300)
1,3,5-Trimethylbenzene	10000	100000	NIA	ND(220)	ND(230)	ND(3500)	ND(350)	ND(300)
n-Propylbenzene	N/A	N/A	N/A	ND(44)	ND(46)	ND(710)	ND(70)	ND(61)
Acetone	3000	60000	1000000	ND(440)	ND(460)	ND(7100)	ND(700)	ND(610)
Benzene	10000	60000	200000	280	240	35000	1200	011
Chloroform	100	10000	500000	ND(65)	ND(69)	(0011)ON	ND(100)	ND(92)
Ethylbenzene	80000	500000	1000000	ND(44)	ND(46)	15000	510	150
Isopropylbenzene	1000001	10000000	NIA	ND(44)	ND(46)	ND(710)	ND(70)	ND(61)
Methylene chloride	100	200000	7000000	ND(440)	ND(460)	ND(7100)	ND(700)	ND(610)
Naphthalene	4000	1000000	1000000	6300	3700	15000	1400	360
Styrene	2000	20000	1000000	ND(44)	ND(46)	010	200	ND(61)
Toluene	00006	500000	10000001	240	230	42000	2200	240
o-Xvlene	50000	500000	500000	ND(44)	ND(46)	2900	220	ND(61)
p/m-Xylene	50000	50000	50000	100	120	10000	760	160
SVOCs (ug/kg)								
1-Methylnaphthalene	NIA	N/A	N/A	ND(13000)	ND(5400)	2600	ND(600)	ND(5300)
2-Methylnaphthalene	4000	1000000	1000000	ND(22000)	ND(8600)	3500	ND(950)	ND(8400)
Acenaphthene	20000	250000	1000000	ND(13000)	ND(5400)	ND(610)	ND(600)	ND(5300)
Acenaphthylene	100000	1000000	10000001	14000	ND(5400)	ND(610)	ND(600)	ND(5300)
Anthracene	100000	1000000	1000000	21000	ND(5400)	ND(610)	ND(600)	ND(5300)
Benzo(a)anthracene	200	1000	100000	48000	ND(5400)	ND(610)	ND(600)	0009
Benzo(a)pyrene	200	700	100000	40000	ND(5400)	ND(610)	ND(600)	6500
Benzo(e)pyrene	NIA	N/A	N/A	25000	ND(5400)	ND(610)	ND(600)	ND(5300)
Benzo(b)fluoranthene	700	1000	100000	32000	ND(5400)	ND(610)	ND(600)	ND(5300)
Benzo(ghi)perylene	100000	250000	1000000	24000	ND(5400)	ND(610)	ND(600)	ND(5300)
Benzo(k)fluoranthene	7000	10000	400000	34000	ND(5400)	ND(610)	ND(600)	ND(5300)
Chrysene	7000	10000	400000	46000	ND(5400)	ND(610)	ND(600)	0009
Dibenzo(a,h)anthracene	700	700	100000	ND(13000)	ND(5400)	ND(610)	ND(600)	ND(5300)

6/27/2003

Haley & Aldrich, Inc. G:00558'634 Phase III\Phase III Tables xls Page 2 of 8

1

9

TABLE III SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLÁN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

ľ

LOCATION ID	MCP	MCP	MCP	028-8917	02B-B917	02B-B918	02B-B918	028-8919
SAMPLE ID	METHOD 1	METHOD 1	METHOD 1	02B-917-MW	02M-1001-MW	02B-918-MW	02B-918-MW/S10	02B-919-MW/S2
SAMPLE DATE	RCS-1	RCS-2	NCL	2/18/2002	2/18/2002	2/18/2002	2/19/2002	2/19/2002
DEPTH INTERVAL	(ug/kg)	(ug/kg)	(ug/kg)	2.6 to 4.6 ft	2.6 to 4.6 ft	14.2 to 16.2 ft	18.2 to 20.2 ft	3 to 5 ft
Fluoranthene	100000	100000	1000000	95000	7400	ND(610)	(009)(UN	12000
Fluorene	400000	200000	10000000	ND(13000)	ND(5400)	ND(610)	ND(600)	ND(5300)
Indeno(1.2.3-cd)ovrene	700	1000	100000	23000	ND(7500)	ND(850)	ND(830)	ND(7400)
Nanhthalene	4000	100000	1000000	ND(13000)	ND(5400)	5300	(009) ON	ND(5300)
Phenanthrene	100000	100000	10000000	72000	ND(5400)	ND(610)	ND(600)	0006
Pervlene	NIA	NIA	N/A	ND(13000)	ND(5400)	ND(610)	ND(600)	ND(5300)
Pyrene	700000	200000	1000000	86000	7000	ND(610)	ND(600)	15000
Metals (ug/kg)			100					
Arsenic	30000	30000	300000	4500	6200	3300	1800	5200
Barium	1000000	2500000	1000000	62000	41000	6500	10000	25000
Cadmium	30000	80000	800000	ND(430)	ND(430)	ND(480)	ND(470)	ND(420)
Chromium	1000000	250000	10000000	16000	22000	6200	9800	10000
Lead	300000	600000	6000000	110000	66000	ND(2400)	ND(2400)	38000
Mercury	20000	60000	600000	760	220	ND(100)	(06)UN	280
Celenium	400000	250000	1000000	ND(860)	ND(860)	ND(960)	ND(940)	ND(830)
Silver	100000	200000	200000	ND(430)	ND(430)	ND(480)	ND(470)	ND(420)
Cyanide							100	
Cyanide, Total	100000	100000	400000	15000	0099	1200	440	3500

Notes:

 ND (5): Compound not detected above detection limit; detection limit is shown in parentheses.

2. VOCs: Volatile Organic Compounds;

SVOCs: Semivolatile Organic Compounds;

Table lists detected compounds only; compounds that were not detected are not listed. Page 4 of 8

TABLE III SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

-

....

i

ł

L

1

Ľ

1

E

LOCATION ID	MCP	MCP	MCP	028-8920	028-8920	028-8920	028-8921	028-8921
SAMPLE ID	METHOD 1	METHOD 1	METHOD 1	02B-920-MW (.8-2.8) (	02B-920-MW (14-16)	02B-920-MW/S9	02B-921-MW/S10	02B-921-MW/S2
SAMPLE DATE	RCS-1	RCS-2	UCL	2/20/2002	2/20/2002	2/21/2002	2/21/2002	2/21/2002
DEPTH INTERVAL	(ug/kg)	(ug/kg)	(ug/kg)	0.8 to 2.8 ft	14 to 16 ft	18 to 20 ft	19 to 21 ft	2.5 to 4.5 ft
Fluoranthene	100000	100000	1000000	2200	ND(620)	ND(640)	ND(620)	ND(520)
Fluorene	400000	200000	1000000	ND(1100)	ND(620)	ND(640)	ND(620)	ND(520)
Indeno(1,2,3-cd)pyrene	700	1000	100000	2800	ND(620)	(006)QN	ND(860)	ND(720)
Naphthalene	4000	1000000	1000000	(0011)ON	ND(620)	ND(640)	ND(620)	ND(520)
Phenanthrene	100000	100000	10000000	3300	ND(620)	ND(640)	ND(620)	ND(520)
Pervlene	N/A	N/A	NIA	1100	ND(620)	ND(640)	ND(620)	ND(520)
Pyrene	70000	200000	1000000	7200	ND(620)	ND(640)	ND(620)	ND(520)
Merals (ug/kg)								
Arsenic	30000	30000	300000	6800	1700	3900	2700	11000
Barium	100000	250000	10000001	27000	7900	46000	26000	28000
Cadmium	30000	80000	80000	ND(430)	ND(490)	ND(500)	ND(480)	ND(410)
Chromium	100000	250000	1000000	10000	6200	12000	14000	20000
Lead	300000	600000	6000000	110000	ND(2400)	3000	2600	4000
Mercury	20000	60009	600000	640	ND(100)	ND(100)	ND(100)	ND(80)
Selenium	400000	250000	10000000	ND(850)	ND(980)	ND(1000)	(010)(UN)	ND(820)
Silver	100000	200000	200000	ND(430)	ND(490)	ND(500)	ND(480)	ND(410)
Cyanide								
Cvanide, Total	100000	100000	400000	3800	8400	860	460	ND(170)

yanide

-----

٠,

Notes: 1. ND (5): Compound not detected above detection limit;

1

detection limit is shown in parentheses.

 VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

ų,

1

 Table lists detected compounds only; compounds that were not detected are not listed.

TABLE III SUMMARY OF SUPPLEMENTAL SOIL SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

LOCATION ID	MCP	MCP	MCP	028-8923	028-B924	02B-B924
SAMPLE ID	METHOD 1	METHOD 1	METHOD 1	02B-923-MW/S8	02B-B924-S7	02B-B924-S8
SAMPLE DATE	RCS-1	RCS-2	NCL	2/22/2002	7/28/2002	7/28/2002
DEPTH INTERVAL	(ug/kg)	(gg/gg)	(ug/kg)	15 to 17 ft	12.5 to 14.5 ft	14.5 to 16.5 ft
Fluoranthene	1000000	100000	1000000	ND(60000)	ND(600)	ND(620)
Fluorene	400000	200000	10000001	ND(60000)	ND(600)	ND(620)
Indeno(1,2,3-cd)pyrene	700	1000	100000	ND(830000)	ND(600)	ND(620)
Naphthalene	4000	1000000	10000000	4300000	ND(600)	ND(620)
Phenanthrene	100000	10000	1000000	ND(60000)	770	ND(620)
Perylene	N/A	N/A	N/A	ND(60000)	ND(600)	ND(620)
Pyrene	700000	200000	1000000	ND(60000)	ND(600)	ND(620)
Metals (ug/kg)						
Arsenic	30000	30000	300000	2900	1400	1200
Barium	100000	250000	1000000	13000	12000	0096
Cadmium	30000	80000	800000	ND(470)	ND(470)	ND(480)
Chromium	1000000	250000	1000000	7200	8200	7400
Lead	30000	600000	6000000	ND(2300)	2900	2400
Mercury	20000	60000	000009	(06)(IN	ND(90)	ND(90)
Selenium	40000	2500000	1000000	ND(930)	ND(940)	ND(970)
Silver	100000	200000	200000	ND(470)	ND(470)	ND(480)
Cyanide						
Cvanide, Total	100000	100000	4000000	620	:	:

Notes:

1. ND (5): Compound not detected above detection limit;

detection limit is shown in parentheses. 2. VOCs: Volatile Organic Compounds;

SVOCs: Semivolatile Organic Compounds;

 Table lists detected compounds only; compounds that were not detected are not listed.

Haley & Aldrich, Inc. G:00558\634 Phase III\Phase III Tables.xls

Page 8 of 8

Page 1 of 18

1

TABLE IV SUMMARY OF SUPPLEMENTAL, GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

ł

4

WELL/LOCATION ID	METHOD I	METHOD 1	METHOD I	02B-B918-OW	02B-B918-OW	02B-B918-OW	02B-B918-OW
SAMPLE ID	MCP	MCP	MCP	02B-B918-OW	02B-B900-DUP	02B-B918-OW	02B-B918-OW
SAMPLE DATE	I-WD	GW-2	ncr	3/10/2002	3/10/2002	5/5/2002	7/27/2002
SAMPLING NOTES	(I/g/I)	(I/gu)	(I/gu)		Dup. Of 02B-B918-OW		
VOC5 (HB/I)							
1,2,4-Trimethylbenzene	00001	100000	N/A	ND(5000)	ND(5000)	ND(1200)	ND(1200)
1,3,5-Trimethylbenzene	100	1000	N/A	ND(5000)	ND(5000)	ND(1200)	ND(1200)
2-Chlorotoluene	NIA	NIA	N/A	ND(5000)	ND(5000)	ND(1200)	ND(1200)
4-Isopropyltoluene	1000	10000	N/A	ND(1000)	ND(1000)	ND(250)	ND(250)
Acetone	3000	50000	100000	(00001)CIN	ND(1000)	ND(2500)	ND(2500)
Benzene	5	2000	70000	56000	00099	55000	41000
Chlorobenzene	100	500	10000	ND(1000)	ND(1000)	ND(250)	ND(250)
Chloroform	S	400	100000	ND(1500)	ND(1500)	ND(380)	ND(380)
cis-1,2-Dichloroethene	70	30000	100000	(0001)CIN	ND(1000)	ND(250)	ND(250)
trans-1,2-Dichloroethene	100	20000	100000	ND(1500)	ND(1500)	ND(380)	ND(380)
trans-1,3-Dichloropropene	N/A	NIA	N/A	ND(1000)	ND(1000)	ND(250)	ND(250)
Ethylbenzene	700	4000	100000	7100	10000	8200	8600
Isopropylbenzene	10000	100000	N/A	(0001)(IN	(0001)QN	ND(250)	ND(250)
2-Butanone	400	50000	100000	ND(10000)	(00001)GN	ND(2500)	ND(2500)
Naphthalene	20	0009	60000	ND(5000)	ND(5000)	4200	ND(1200)
n-Butylbenzene	N/A	N/A	N/A	ND(1000)	ND(1000)	ND(250)	ND(250)
n-Propylbenzene	1000	10000	NIA	(0001)QN	ND(1000)	ND(250)	ND(250)
m.p-Xylene	0009	0009	N/A	3700	5100	4400	2800
o-Xylene	0009	0009	N/A	1200	1600	1300	940
sec-Butylbenzene	N/A	N/A	N/A	ND(1000)	ND(1000)	510	ND(250)
Styrene	100	906	100000	ND(1000)	(0001)CN	480	400
tert-Butylbenzene	1000	10000	N/A	ND(5000)	ND(5000)	ND(1200)	ND(1200)
Toluene	1000	6000	000001	30000	38000	31000	24000
Tetrachloroethene	5	3000	50000	ND(1000)	ND(1000)	ND(250)	ND(250)
Trichloroethene	S	300	100000	ND(1000)	ND(1000)	ND(250)	ND(250)

Haley & Aldrich, Inc. G:\06558\634 Phase III\Phase III Tables.xls

6/27/2003
Page 2 of 18

ļ

-

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

WELL/LOCATION ID	METHOD I	METHOD 1	METHOD 1	02B-B918-OW	02B-B918-OW	02B-B918-OW	02B-B918-OW
SAMPLE ID	MCP	MCP	MCP	02B-B918-OW	02B-B900-DUP	02B-B918-OW	02B-B918-OW
SAMPLE DATE	GW-1	GW-2	NCL	3/10/2002	3/10/2002	5/5/2002	7/27/2002
SAMPLING NOTES	(l/gn)	(l/gn)	(l/gn)		Dup. Of 02B-B918-OW		
SVOUS (ug/l)							
1-Methylnaphthalene	N/A	N/A	N/A	420	260	430	330
2-Methylnaphthalene	10	3000	100000	ND(400)	ND(200)	200	170
2-Methylphenol	5000	50000	N/A	ND(300)	ND(150)	ND(120)	:
Acenaphthene	20	5000	50000	ND(250)	ND(130)	ND(100)	ND(50)
Bis(2-ethylhexyl)phthalate	9	30	100000	ND(500)	ND(250)	ND(200)	;
Pluorene	300	3000	30000	ND(250)	ND(130)	ND(100)	ND(50)
Naphthalene	20	0009	60000	2700	1700	2700	1500
Phenol	4000	30000	100000	ND(350)	180	220	;
Metals (ug/l)							
Arsenic	50	400	4000	28	25	16	17
Barium	2000	30000	100000	80	80	66	80
Fluoride	N/A	N/A	N/A	3	1	:	;
Cyanide (ug/l)	Ş	4	TIN	201			000
Cyanide, 10tal	10	10	NIA	00/	830	132	871
Cyanide, Physiologically Available	NIA	N/A	2000	94	87	57	29

Notes:

1. ND (5): Compound not detected above detection limit; detection limit

-

is shown in parentheses.

VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

Table lists detected compounds only; compounds that were not detected are not listed.

 Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples.

Page 3 of 18

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS MALDEN, MASSACHUSETTS

ľ

and the second test	1 Accession 1	1 ADDRESS OF	- Announced a	AAD DOOD 000	UND DOOD DOO	UND BOAT OW	AND BOAT OW	WO DOUT OW
WELL/LUCATION ID	METHOD 1	METHOD I	METHOD I	MO-0769-970	MO-0769-970	MO-1769-970	MO-1769-970	MO-7760-070
SAMPLE ID	MCP	MCP	MCP	02B-B920-OW	02B-B920-OW	02B-B921-OW	02B-B921-OW	02B-B922-OW
SAMPLE DATE	I-M9	GW-2	NCL	3/10/2002	5/5/2002	3/10/2002	5/5/2002	5/5/2002
SAMPLING NOTES	(l/gn)	(l/gn)	(l/gn)					
VOCs (ug/l)								
1,2,4-Trimethylbenzene	10000	100000	N/A	5.3	ND(2.5)	28	ND(12)	ND(2500)
1,3,5-Trimethylbenzene	100	1000	N/A	ND(2.5)	ND(2.5)	5.8	ND(12)	ND(2500)
2-Chlorotoluene	N/A	NIA	N/A	ND(2.5)	ND(2.5)	ND(2.5)	ND(12)	ND(2500)
4-Isopropyltoluene	1000	10000	N/A	ND(0.5)	ND(0.5)	9.3	5	ND(500)
Acetone	3000	50000	100000	ND(5)	ND(5)	ND(5)	ND(25)	ND(5000)
Benzene	5	2000	70000	80	4.5	40	160	12000
Chlorobenzene	100	500	10000	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	ND(500)
Chloroform	5	400	100000	ND(0.75)	ND(0.75)	ND(0.75)	ND(3.8)	ND(750)
cis-1.2-Dichloroethene	10	30000	100000	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	ND(500)
trans-1,2-Dichlorocthene	100	20000	100000	ND(0.75)	ND(0.75)	ND(0.75)	ND(3.8)	ND(750)
trans-1,3-Dichloropropene	N/A	N/A	N/A	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	ND(500)
Ethylbenzene	700	4000	100000	ND(0.5)	0.6	29	49	3900
Isopropylbenzene	10000	100000	N/A	1.4	0.79	01	8.8	ND(500)
2-Butanone	400	50000	100000	ND(5)	ND(5)	ND(5)	ND(25)	ND(5000)
Naphthalene	20	0009	60000	ND(2.5)	ND(2.5)	20	180	6400
n-Butvibenzene	N/A	N/A	N/A	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	ND(500)
n-Propylbenzenc	1000	10000	N/A	ND(0.5)	ND(0.5)	2.6	3.6	ND(500)
m.p-Xylene	6000	6000	N/A	ND(0.5)	ND(0.5)	1.9	4.8	4800
o-Xvlenc	6000	6009	NIA	1.1	ND(0.5)	4	7.2	2100
sec-Butylbenzene	N/A	N/A	N/A	ND(0.5)	0.82	2.7	ND(2.5)	ND(500)
Styrene	100	006	100000	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	1800
tert-Butylbenzene	1000	10000	N/A	ND(2.5)	ND(2.5)	ND(2.5)	ND(12)	ND(2500)
Toluene	1000	0009	100000	ND(0.75)	ND(0.75)	1.6	ND(3.8)	27000
Tetrachloroethene	5	3000	50000	ND(0.5)	ND(0.5)	ND(0.5)	7.8	ND(500)
Trichloroethene	2	300	100000	ND(0.5)	ND(0.5)	ND(0.5)	ND(2.5)	ND(500)
				5				

Haley & Aldrich, Inc. G:\06558\634 Phase III\Phase III Tables.xls

Page 4 of 18

)

J

SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS 129 COMMERCIAL STREET TABLE IV

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	02B-B920-OW	02B-B920-OW	02B-B921-OW	02B-B921-OW	02B-B922-OW
SAMPLE ID	MCP	MCP	MCP	02B-B920-OW	02B-B920-OW	02B-B921-OW	02B-B921-OW	02B-B922-OW
SAMPLE DATE	GW-1	GW-2	UCL	3/10/2002	5/5/2002	3/10/2002	5/5/2002	5/5/2002
SAMPLING NOTES	(I/gn)	(I/gn)	(I/gu)					
SVOCs (ug/l)	207							
I-Methylnaphthalene	N/A	N/N	N/A	ND(5)	ND(5)	430	300	550
2-Methylnaphthalene	10	3000	100000	ND(8.1)	ND(8)	ND(40)	ND(40)	ND(800)
2-Methylphenol	5000	50000	N/A	ND(6.1)	ND(6)	ND(30)	ND(30)	ND(600)
Acenaphthene	20	5000	50000	7.5	6.4	ND(25)	ND(25)	ND(500)
Bis(2-ethylhexyl)phthalate	9	30	100000	ND(10)	ND(10)	ND(50)	ND(50)	ND(1000)
Fluorene	300	3000	30000	12	12	ND(25)	ND(25)	ND(500)
Vaphthalene	20	0009	60000	ND(5)	ND(5)	ND(25)	110	0016
Phenol	4000	30000	100000	ND(7.1)	ND(7)	ND(35)	ND(35)	ND(700)
Metals (ug/l)								
Arsenic	50	400	4000	32	24	(01)QN	ND(10)	ND(10)
<b>arium</b>	2000	30000	100000	130	110	140	210	8
ituoride	N/A	N/A	N/A	:	:	;	:	ł
Syanide (ug/l) Vanide Total	01	0	NIA	513	LCL	040	1150	650
yanide. Physiologically Available	N/A	NIA	2000	81	3	76	86	02

Notes:

1. ND (5): Compound not detected above detection limit; detection limit

1

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

3. Table lists detected compounds only; compounds that were not detected

are not listed.

4. Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples.

G:\06558\634 Phase III\Phase III Tables.xls Haley & Aldrich, Inc.

Page 5 of 18

J

1

]

J

]

J

ļ

1

ļ

ļ

1

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

,

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	02B-B922-OW	02B-B923-OW	02B-B923-OW	02B-B923-OW
CAMPIEID	MCD	MCD	MCD	WO-COG-BOD	WO 2002 DW	WO LCOD BCD	DUID A
	1 mor	C MO	INC	M0-7760-070	WD-0769-070	MO-0768-070	V-JOO
SAMFLE DATE	1-40	7-40		700711711	ZNUTICIC	700711711	7007/6/6
SAMPLING NOTES	(I/gu)	(I/gn)	(I/gn)				Dup. Of 02B-B923-OW
VOCS (ug/i)							
1,2,4-Trimethylbenzene	10000	100000	NIA	ND(2500)	280	ND(250)	ND(500)
1,3,5-Trimethylbenzene	100	1000	N/A	ND(2500)	ND(250)	ND(250)	ND(500)
2-Chlorotoluene	N/A	N/A	N/A	ND(2500)	ND(250)	ND(250)	ND(500)
4-Isopropyltoluene	1000	10000	NIA	ND(500)	ND(50)	ND(50)	ND(100)
Acetone	3000	50000	100000	ND(5000)	ND(500)	ND(500)	ND(1000)
Benzene	5	2000	70000	6500	2200	370	2300
Chlorobenzene	100	500	10000	ND(500)	ND(50)	ND(50)	ND(100)
Chloroform	5	400	100000	ND(750)	ND(75)	ND(75)	ND(150)
cis-1,2-Dichloroethene	70	30000	000001	ND(500)	ND(50)	ND(50)	ND(100)
trans-1,2-Dichloroethene	100	20000	100000	ND(750)	ND(75)	ND(75)	ND(150)
trans-1,3-Dichloropropene	N/A	NIA	N/A	ND(500)	ND(50)	ND(50)	ND(100)
Ethylbenzene	700	4000	100000	7400	160	130	160
Isopropythenzene	10000	100000	N/A	ND(500)	ND(50)	ND(50)	ND(100)
2-Butanone	400	50000	100000	ND(5000)	ND(500)	ND(500)	ND(1000)
Naphthalene	20	0009	60000	5900	5100	3600	3600
n-Butylbenzene	N/A	NIA	N/A	ND(500)	ND(50)	ND(50)	ND(100)
n-Propylbenzene	1000	00001	N/A	ND(500)	ND(50)	ND(50)	ND(100)
m,p-Xylene	0009	6000	N/A	7800	890	330	920
o-Xylene	0009	0009	N/A	3000	490	180	500
sec-Butylbenzene	N/A	NIA	N/A	ND(500)	ND(50)	ND(50)	ND(100)
Styrene	100	006	100000	2700	ND(50)	ND(50)	ND(100)
tert-Butylbenzene	1000	10000	N/A	ND(2500)	ND(250)	ND(250)	ND(500)
Toluene	1000	0009	100000	54000	680	011	740
Tetrachloroethene	S	3000	50000	ND(500)	55	ND(50)	ND(100)
Trichloroethene	s	300	100000	ND(500)	ND(50)	ND(50)	ND(100)

Page 6 of 18

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

ł

WELL/LOCATION ID	METHOD I	METHOD 1	METHOD 1	02B-B922-OW	02B-B923-OW	02B-B923-OW	02B-B923-OW
SAMPLE ID	MCP	MCP	MCP	02B-B922-OW	02B-B923-OW	02B-B923-OW	DUP-A
SAMPLE DATE	GW-1	GW-2	NCL	7/27/2002	5/5/2002	7/27/2002	5/5/2002
SAMPLING NOTES	(l/gn)	(I/gu)	(1/gn)				Dup. Of 02B-B923-OW
510C5 (ag/1)							
1-Methylnaphthalene	N/N	NIA	NIA	420	340	220	310
2-Methylnaphthalene	10	3000	100000	540	450	230	400
2-Methylphenol	5000	50000	N/A	;	ND(300)	:	ND(300)
Acenaphthene	20	5000	50000	ND(250)	ND(250)	ND(100)	ND(250)
Bis(2-ethylhexyl)phthalate	9	30	100000	;	ND(500)	:	ND(500)
Fluorene	300	3000	30000	ND(250)	ND(250)	(001)CIN	ND(250)
Naphthalene	20	6000	60000	5600	4200	1900	4000
Phenol	4000	30000	100000	ł	ND(350)	;	ND(350)
Metals (ug/l)							
Arsenic	50	400	4000	ND(5)	ND(10)	ND(5)	ND(10)
Barium	2000	30000	100000	110	30	40	30
Fluoride	N/A	N/A	N/A	:	;	;	1
Cyanide (ug/l)			6				
Cyanide, Total	10	10	N/A	328	778	923	845
Cyanide, Physiologically Available	N/A	N/A	2000	32	120	190	128

Notes:

1. ND (5): Compound not detected above detection limit; detection limit

ŝ

is shown in parentheses.

VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;
 Table first detected community only compounds that uses not detected

Table fists detected compounds only; compounds that were not detected are not listed.

 Surrogates were not recovered fur samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samplet.

Haley & Aldrich, Inc. G:\06558\634 Phase III\Phase III Tables,xls

Page 7 of 18

-

-

.....

-

-

-

1. . .

---

0 H 2

-

-

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN

FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS 97B-B618-OW 97B-B618-OW ND(0.75) ND(0.5) 5/7/2002 VD(2.5) ND(2.5) ND(2.5) ND(0.5) ND(0.5) ND(0.5) ND(2.5) ND(0.5) ND(0.5) ND(0.5) ND(0.5) ND(0.5) ND(0.75) VD(0.75) ND(5) ND(5) ND(0.5) ND(2.5) 2.4 2.9 0.8 4.8 21 97B-B618-OW 02B-B618-OW 3/11/2002 ND(0.75) ND(0.75) ND(5) ND(2.5) ND(0.5) ND(0.5) ND(0.5) ND(0.5) ND(0.75) ND(0.5) ND(0.5) VD(2.5) ND(2.5) ND(2.5) ND(0.5) ND(0.5) ND(0.5) ND(0.5) ND(2.5) ND(0.5) ND(5) 3.2 3.7 5.3 21 97B-B617-OW 97B-B617-OW ND(380) ND(250) ND(2500) 8/1/2002 VD(1200) ND(250) ND(250) ND(250) ND(250) ND(250) ND(1200) VD(1200) ND(1200) VD(2500) ND(380) ND(250) ND(250) ND(250) VD(250) 7400 3000 10000 3400 20000 37000 VD(250) 97B-B617-OW 97B-B617-OW 3/12/2002 ND(2500) ND(250) ND(250) ND(1200) VD(1200) ND(250) ND(250) VD(1200) ND(250) VD(2500) ND(250) ND(380) ND(380) ND(250) VD(1200) 16000 ND(250) ND(250) 26000 6800 1200 3200 9200 290 400 97B-B617-OW WO-719-879 ND(250) 5/8/2002 VD(1200) ND(1200) ND(1200) VD(2500) ND(380) ND(250) VD(2500) ND(250) ND(1200) ND(250) ND(250) VD(250) ND(250) ND(250) ND(250) ND(380) ND(250) 11000 21000 33000 2600 13000 4500 590 METHOD 1 00000 00000 00000 00000 00000 00000 00000 0000 60000 00000 70000 50000 MCP N/A UCL (I/gu) N/A NIA N/A N/A NIA N/A NIA N/A NIA METHOD I 000001 GW-2 (I/gu) 00000 30000 50000 10000 0000 50000 00001 MCP 4000 6000 0001 2000 NIA 6000 6000 N/A 500 N/A NIA 80 0009 3000 300 METHOD 1 (I/gn) GW-1 MCP 10000 0000 N/A 20 N/A N/A N/A N/A N/A N/A 1000 5 5 700 700 700 8 400 s so trans-1, 3-Dichloropropene trans-1,2-Dichloroethene 1,3,5-Trimethylbenzene .2.4-Trimethylbenzene WELL/LOCATION ID cis-1,2-Dichloroethene SAMPLING NOTES 4-Isopropyltoluene Tetrachloroethene SAMPLE DATE Isopropylbenzene sec-Butylbenzene tert-Butylbenzene 2-Chlorotoluene n-Propylbenzene **Trichloroethene** n-Butylbenzene Chlorobenzene Ethylbenzene SAMPLEID VUCS (ugil) Naphthalene m,p-Xylene Chloroform 2-Butanone o-Xylene Benzene Acetone Styrene Foluene

Page 8 of 18

TABLE IV SUMMARY OF SUPPLEMENTAL, GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS MALDEN, MASSACHUSETTS

1

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	97B-B617-OW	97B-B617-OW	97B-B617-OW	97B-B618-OW	97B-B618-OW
SAMPLE ID	MCP	MCP	MCP	97B-617-OW	97B-B617-OW	97B-B617-OW	02B-B618-OW	97B-B618-OW
SAMPLE DATE	GW-1	GW-2	NCL	5/8/2002	3/12/2002	8/1/2002	3/11/2002	5/7/2002
SAMPLING NOTES	(l/gn)	(l/gn)	(I/gn)					
SVOCs (ugil)								
1-Methylnaphthalene	N/A	N/A	N/A	200	ND(260)	230	ND(5.2)	ND(5)
2-Methylnaphthalene	10	3000	100000	160	ND(410)	ND(200)	ND(8.3)	ND(8)
2-Methylphenol	5000	50000	N/A	ND(120)	ND(310)	180	ND(6.2)	ND(6)
Acenaphthene	20	5000	50000	ND(100)	ND(260)	ND(120)	ND(5.2)	ND(5)
Bis(2-ethythexyl)phthalate	9	30	100000	ND(200)	ND(520)	ND(250)	ND(10)	ND(10)
Fluorene	300	3000	30000	(001)QN	ND(260)	ND(120)	ND(5.2)	ND(5)
Naphthalene	20	0009	60000	2000	1600	1800	ND(5.2)	ND(5)
Phenol	4000	30000	10000	ND(140)	ND(360)	ND(180)	ND(7.3)	(L)(D)
Metals (ug/l)								
Arsenic	50	400	4000	ND(10)	10	6	ND(10)	ND(5)
Barium	2000	30000	100000	40	50	40	20	10
Fluoride	N/A	N/A	N/N	;	;	:	;	8
Cyanide (ug/l)								
Cyanide, Total	10	10	NIA	16	14	24	06	66
Cyanide, Physiologically Available	N/A	N/A	2000	9	ND(5)	ND(5)	ND(5)	34

## Notes:

v.6

1. ND (5): Compound not detected above detection limit; detection limit

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

Table lists detected compounds only; compounds that were not detected are not listed.  Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples.

Page 10 of 18

1

]

ł

l

Į

l

I

1

l

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	97B-B627-OW	97B-B627-OW	97B-B627-OW	97B-B628-OW	97B-B628-OW
SAMPLE ID	MCP	MCP	MCP	97B-B627-OW	97B-B627-OW	97B-B627-OW	97B-B628-OW	97B-B628-OW
SAMPLE DATE	GW-1	GW-2	UCL,	3/12/2002	5/7/2002	8/1/2002	3/12/2002	5/7/2002
SAMPLING NOTES	(I/gn)	(I/gn)	(l/gn)					
570CS (ag/i)								
1-Methylnaphthalene	NIA	N/A	N/A	ND(5.1)	ND(5)	ND(5)	ND(52)	ND(100)
2-Methylnaphthalene	10	3000	100000	ND(8.2)	ND(8)	ND(8)	ND(82)	(091)(IN
2-Methylphenol	5000	50000	N/A	ND(6.1)	ND(6)	ND(6)	ND(62)	ND(120)
Accnaphthene	20	5000	50000	ND(5.1)	ND(5)	ND(5)	ND(52)	(001)(IN)
Bis(2-ethylhexyl)phthalate	9	30	100000	ND(10)	ND(10)	ND(10)	ND(100)	ND(200)
Fluorene	300	3000	30000	ND(5.1)	ND(5)	ND(5)	ND(52)	ND(100)
Naphthalene	20	6000	60000	ND(5.1)	ND(5)	ND(5)	390	390
Phenol	4000	30000	100000	(I.7)dN	(L)(I)	(L)(D)	ND(72)	ND(140)
Metals (ug/l)								
Arsenic	50	400	4000	(01)(IN)	ND(5)	ND(5)	ND(10)	ND(5)
Barium	2000	30000	100000	120	100	I40	20	30
Fluoride	N/A	N/A	N/A	ND(200)	ND(200)	:	220	ND(200)
Cyanide (ug/l)	9	U1	NIA	122	270	745	1210	1650
Cyanuce, Lotat	1	2	1000			} (	000	
Cyanide, Physiologically Available	N/A	N/A	2000	51	80	10	805	383

# Notes:

0.0

1. ND (5): Compound not detected above detection limit; detection limit

ŀ

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

Table lists detected compounds only; compounds that were not detected to the second lists.

are not listed.

 Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples.

Haley & Aldrich, Inc. G:\06558\634 Phase III\Phase III Tables.xls



T.

Page 12 of 18

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	97B-B628-OW	GP98-103-OW	GP98-103-OW	GP98-103-OW	I-WM
SAMPLE ID	MCP	MCP	MCP	97B-B628-OW	GP98-103-OW	GP98-103-OW	GP98-103-OW	I-WM
SAMPLE DATE	GW-1	GW-2	UCL	8/1/2002	3/12/2002	5/8/2002	8/2/2002	3/11/2002
SAMPLING NOTES	(l/gn)	(I/gn)	(I/gn)					
SVOCs (ug/l)								
1-Methylnaphthalene	N/A	N/A	N/A	ND(10)	ND(5)	ND(5)	ND(5)	ND(5.2)
2-Methylnaphthalene	10	3000	100000	ND(16)	ND(8)	ND(8)	ND(8)	ND(8.3)
2-Methylphenol	5000	50000	N/A	ND(12)	ND(6)	ND(6)	1	ND(6.2)
Acenaphthene	20	5000	50000	ND(10)	ND(5)	ND(5)	ND(5)	ND(5.2)
Bis(2-ethylhexyl)phthalate	9	30	100000	ND(20)	(01)QN	ND(10)	:	ND(10)
Fluorene	300	3000	30000	ND(10)	ND(5)	ND(5)	ND(5)	ND(5.2)
Naphthalene	20	0009	60000	ND(10)	ND(5)	ND(5)	ND(5)	5.4
Phenol	4000	30000	100000	ND(14)	(L) (U)	ND(7)	:	ND(7.3)
Metals (ug/l) Arconic	9	400	4000	ND(S)	NDODON	ND(5)	ND(5)	NDODO
Barium	2000	30000	100000	40	20	20	30	30
Fluoride	N/A	N/A	N/A	:	;	;	:	:
Cyanide (ug/l) Cvanide Total	10	10	N/A	7000	ND(5)	ND(5)	Ŷ	16
Cyanide, Physiologically Available	N/A	NIA	2000	1020		ND(5)	ND(5)	ND(5)

## Notes:

1. ND (5): Compound not detected above detection limit; detection limit

1

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

3. Table lists detected compounds only; compounds that were not detected

are not listed.

 Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., raphthalene) in the samples.

Page 13 of 18

]

1

]

]

]

]

1

ł

J

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD I	I-WM	I-WM	MW-5	MW-5	NC-1	NC-1
SAMPLEID	MCP	MCP	MCP	I-WM	I-WM	MW-5	MW-5	I-DN	NC-1
SAMPLE DATE	GW-1	GW-2	UCL,	5/7/2002	7/31/2002	5/8/2002	7/31/2002	3/11/2002	5/7/2002
SAMPLING NOTES	(I/gn)	(I/gu)	(I/gn)						
VOCs (ug/i)									
1.2.4-Trimethylbenzene	10000	100000	N/A	ND(2.5)	ND(2.5)	ND(5)	ND(5)	ND(2.5)	ND(2.5)
1.3.5-Trimethylbenzene	100	0001	N/A	ND(2.5)	ND(2.5)	ND(5)	ND(5)	ND(2.5)	ND(2.5)
2-Chlorotoluene	N/A	NIA	N/A	ND(2.5)	ND(2.5)	ND(5)	ND(5)	ND(2.5)	ND(2.5)
4-Isopropyltoluene	1000	10000	N/A	ND(0.5)	ND(0.5)	(I)(IN	(I)(IN	ND(0.5)	ND(0.5)
Acetone	3000	50000	100000	ND(5)	ND(5)	(01)QN	ND(10)	ND(5)	ND(5)
Benzene	5	2000	70000	ND(0.5)	0.77	33	250	2.8	2.5
Chlorobenzene	100	500	10000	ND(0.5)	ND(0.5)	(I)(I)	(I)QN	2.6	2.1
Chloroform	5	400	100000	ND(0.75)	ND(0.75)	ND(1.5)	ND(1.5)	ND(0.75)	ND(0.75)
cis-1.2-Dichloroethene	R	30000	100000	ND(0.5)	ND(0.5)	(I)(I)(I)(I)(I)(I)(I)(I)(I)(I)(I)(I)(I)(	(I)(IN	0.54	ND(0.5)
rans-1.2-Dichloroethene	100	20000	100000	ND(0.75)	ND(0.75)	ND(1.5)	ND(1.5)	ND(0.75)	ND(0.75)
rans-1.3-Dichloropropene	N/A	N/A	N/A	ND(0.5)	ND(0.5)	(I)(I)	(I)QN	ND(0.5)	ND(0.5)
Ethylbenzene	700	4000	100000	ND(0.5)	ND(0.5)	45	180	ND(0.5)	ND(0.5)
sonronvibenzene	10000	100000	N/A	ND(0.5)	ND(0.5)	(I)(IN)	20	32	36
2-Butanone	400	50000	100000	ND(5)	ND(5)	(01)QN	ND(10)	ND(5)	ND(5)
Vanhthalene	20	0009	60000	ND(2.5)	ND(2.5)	ND(5)	9.3	ND(2.5)	ND(2.5)
-Butvlbenzene	NIA	N/A	N/A	ND(0.5)	ND(0.5)	(1)QN	(1)(IN	9.5	8.7
-Pronvlbenzene	1000	10000	N/A	ND(0.5)	ND(0.5)	(I) <b>U</b> N	7.9	33	46
n.p-Xvlene	6000	6000	NIA	ND(0.5)	ND(0.5)	(I)(I)	(I)QN	ND(0.5)	ND(0.5)
-Xvlene	6000	6000	N/A	ND(0.5)	ND(0.5)	(I)QN	7.2	1.8	0.82
sec-Butvlhenzene	N/A	N/A	N/A	ND(0.5)	ND(0.5)	(1)QN	(I)QN	19	20
Styrene	100	006	100000	ND(0.5)	ND(0.5)	(I)(IN	(I)QN	ND(0.5)	ND(0.5)
ert-Butvibenzene	1000	10000	N/A	ND(2.5)	ND(2.5)	ND(5)	ND(5)	9.1	9.6
Toluene	1000	6000	100000	ND(0.75)	ND(0.75)	ND(1.5)	ND(1.5)	ND(0.75)	ND(0.75)
Tetrachloroethene	5	3000	50000	ND(0.5)	ND(0.5)	(I)(I)	(I)QN	ND(0.5)	ND(0.5)
Trichloroethene	5	300	100000	ND(0.5)	ND(0.5)	(1)QN	(1)QN	ND(0.5)	ND(0.5)

÷

6/27/2003

Page 14 of 18

]

]

-

]

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

ų,

WELL/LOCATION ID	METHOD 1	METHOD I	METHOD 1	I-WW	I-WM	MW-5	MW-5	NC-1	NC-1
SAMPLE ID	MCP	MCP	MCP	I-WM	I-WM	MW-5	<b>MW-5</b>	NC-1	NC-1
SAMPLE DATE	GW-1	GW-2	UCL	5/7/2002	7/31/2002	5/8/2002	7/31/2002	3/11/2002	5/7/2002
SAMPLING NOTES	(I/gu)	(l/gn)	(l/gn)		1				
570CS (ug/l)									
-Methylnaphthalene	NIA	N/A	NIA	ND(5)	ND(5.3)	ND(5)	13	ND(26)	ND(5)
2-Methylnaphthalene	10	3000	100000	ND(8)	ND(8.5)	ND(8)	ND(8.3)	ND(41)	ND(8)
2-Methylphenol	5000	50000	N/A	(9)QN	:	ND(6)	:	ND(31)	ND(6)
Acenaphthene	20	5000	50000	ND(5)	ND(5.3)	ND(5)	ND(5.2)	ND(26)	ND(5)
3is(2-ethylhexyl)phthalate	9	30	100000	ND(10)	1	ND(10)	;	ND(52)	10
luorene	300	3000	30000	ND(5)	ND(5.3)	ND(5)	ND(5.2)	ND(26)	ND(5)
Vaphthalene	20	6000	60000	ND(5)	ND(5.3)	ND(5)	ND(5.2)	ND(26)	ND(5)
henol	4000	30000	100000	ND(7)	:	ND(7)	ł	ND(36)	ND(7)
detats (ug/l)									
Arsenic	50	400	4000	ND(5)	ND(5)	33	36	32	24
3arium	2000	30000	100000	30	30	99	20	011	001
luoride	N/N	N/A	N/A	;	E	:	;	:	;
Syanide (ug/l)	of t	ş	NIA	10	ŧ	5	8	5	35
yanide, Lotal	2	2	N/A	2	70	50	R	1	66
Cyanide, Physiologically Available	N/A	N/A	2000	s	11	17	80	9	1

Notes:

1. ND (5): Compound not detected above detection limit; detection limit

1

is shown in parentheses.

VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;
 Table lists detected compounds only; compounds that were not detected

Table lists detected compounds only; compounds that were not detected are not listed.

 Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of rative compounds (i.e., naphthalene) in the samples.

Page 15 of 13

ı.

3

]

١

J

Į

ļ

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

WELL/LOCATION ID	METHOD I	METHOD I	METHOD 1	NC-1	NC-2	NC-2	NC-2	NC-3	NC-3
SAMPLEID	MCP	MCP	MCP	NC-1	NC-2	NC-2	NC-2	BOZ-1 DUP	DUP-B
SAMPLE DATE	GW-1	GW-2	ncr	7/31/2002	3/11/2002	5/8/2002	8/2/2002	8/1/2002	5/7/2002
SAMPLING NOTES	(l/gn)	(I/gn)	(I/gn)						Dup. Of NC-3
VUCS (ug/l)									
1.2.4-Trimethylbenzene	10000	100000	N/A	ND(5)	ND(2.5)	ND(2.5)	ND(2.5)	ND(500)	ND(1000)
1,3,5-Trimethylbenzene	100	1000	N/A	ND(5)	ND(2.5)	ND(2.5)	ND(2.5)	ND(500)	ND(1000)
2-Chiorotoluene	NIA	NIA	N/A	29	ND(2.5)	ND(2.5)	ND(2.5)	ND(500)	ND(1000)
4-Isopropyltoluene	1000	10000	N/A	(I)(I)	ND(0.5)	1	ND(0.5)	(001)QN	ND(200)
Acetone	3000	50000	100000	ND(10)	ND(5)	ND(5)	ND(5)	ND(1000)	ND(2000)
Benzene	\$	2000	70000	3	26	27	160	11000	26000
Chlorobenzene	100	500	10000	3.3	ND(0.5)	ND(0.5)	ND(0.5)	ND(100)	ND(200)
Chloroform	5	400	100000	ND(1.5)	ND(0.75)	ND(0.75)	ND(0.75)	ND(150)	ND(300)
cis-1.2-Dichloroethene	2	30000	100000	9.1	ND(0.5)	ND(0.5)	ND(0.5)	ND(100)	ND(200)
trans-1,2-Dichloroethene	100	20000	100000	2.3	ND(0.75)	ND(0.75)	ND(0.75)	ND(150)	ND(300)
trans-1.3-Dichloropropene	NIA	N/A	N/A	(I)(I)	ND(0.5)	ND(0.5)	ND(0.5)	ND(100)	ND(200)
Ethylbenzene	200	4000	10000	(I)(I)	ND(0.5)	ND(0.5)	7.3	1200	2700
Isopropylbenzene	10000	100000	N/A	20	ND(0.5)	1.1	1.9	ND(100)	ND(200)
2-Butanone	400	50000	100000	ND(10)	ND(5)	ND(5)	ND(5)	ND(1000)	ND(2000)
Naphthalene	20	6000	60000	ND(5)	ND(2.5)	2.8	4.2	ND(500)	1300
n-Butylbenzene	N/A	N/A	N/A	10	ND(0.5)	ND(0.5)	ND(0.5)	ND(100)	ND(200)
n-Propylbenzene	1000	10000	N/A	31	ND(0.5)	ND(0.5)	0.54	ND(100)	ND(200)
n.p-Xylene	0009	6000	NIA	(I)(I)	ND(0.5)	ND(0.5)	0.61	110	570
o-Xvlene	0009	6000	N/A	(I)(I)	ND(0.5)	ND(0.5)	0.68	ND(100)	270
sec-Butylbenzene	N/A	N/A	N/A	17	0.78	1.8	1.7	ND(100)	ND(200)
Styrene	100	006	100000	(I)QN	ND(0.5)	ND(0.5)	0.56	ND(100)	ND(200)
ert-Butylbenzene	1000	10000	N/A	8.6	ND(2.5)	ND(2.5)	ND(2.5)	ND(500)	ND(1000)
Toluene	1000	0009	100000	ND(1.5)	ND(0.75)	ND(0.75)	2.8	210	320
Tetrachloroethene	s	3000	50000	(I)QN	ND(0.5)	ND(0.5)	ND(0.5)	ND(100)	ND(200)
Trichloroethene	5	300	100000	(1)QN	ND(0.5)	ND(0.5)	ND(0.5)	(001)CIN	ND(200)
	10.00								

Page 16 of 18

]

]

]

]

]

SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS 129 COMMERCIAL STREET TABLE IV

1

WELL/LOCATION ID	METHOD 1	METHOD 1	METHOD 1	NC-1	NC-2	NC-2	NC-2	NC-3	NC-3
SAMPLE ID	MCP	MCP	MCP	NC-1	NC-2	NC-2	NC-2	BOZ-1 DUP	DUP-B
SAMPLE DATE	GW-1	GW-2	NCL	7/31/2002	3/11/2002	5/8/2002	8/2/2002	8/1/2002	5/7/2002
SAMPLING NOTES	(l/gn)	(I/gn)	(l/gn)						Dup. Of NC-3
SVOUS (ug/1)									
1-Methylnaphthalene	N/A	N/A	N/A	ND(5.1)	ND(5.2)	ND(5)	ND(5)	34	49
2-Methylnaphthalene	10	3000	100000	ND(8.2)	ND(8.2)	ND(8)	ND(8)	ND(16)	ND(40)
2-Methylphenol	5000	50000	NIA	ŀ	ND(6.2)	ND(6)	;	ND(12)	ND(30)
Acenaphthene	20	5000	50000	ND(5.1)	ND(5.2)	ND(5)	ND(5)	ND(10)	ND(25)
Bis(2-ethylhexyl)phthalate	9	30	10000	:	ND(10)	ND(10)	;	ND(20)	ND(50)
Fluorenc	300	3000	30000	ND(5.1)	ND(5.2)	ND(5)	ND(5)	ND(10)	ND(25)
Naphthalene	20	6000	60000	ND(5.1)	ND(5.2)	ND(5)	ND(5)	120	370
Phenol	4000	30000	100000	;	ND(7.2)	ND(7)	:	ND(14)	ND(35)
Metals (ug/l)									
Arsenic	20	400	4000	31	14	ND(10)	12	22	16
Barium	2000	30000	100000	80	20	20	20	110	110
Fluoride	NIA	N/N	N/A	ł	:	;	:	:	;
Cyanide (ug/l)	10								
Cyanide, Total	01	10	N/A	18	28	25	26	164	217
Cyanide, Physiologically Available	N/A	N/A	2000	ND(5)	6	9	ND(5)	8	17

## Notes:

1. ND (5): Compound not detected above detection limit; detection limit

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

3. Table lists detected compounds only; compounds that were not detected are not listed.

Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples. 4

Haley & Aldrich, Inc. G:\06558\634 Phase III\Phase III Tables.tls

Page 17 of 18

1

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS MALDEN, MASSACHUSETTS

х

SAMPLE ID MCP SAMPLE DATE GW-1 SAMPLE DATE GW-1 SAMPLING NOTES (ug/l) VOCs (ug/l) 1,2,4-Trimethylbenzene 1000 1,3,5-Trimethylbenzene 1000 1,3,5-Trimethylbenzene 1000 A-lsopropyltoluene 5 A-lsopropyltoluene 5 Chlorobenzene 5 Chlorobenzene 5 cis-1,2-Dichloroethene 70	MCP GW-2 (ug/l) 10000 1000	MCP	NC-3	NC-3	NC-3	NC-DUP
SAMPLE DATE GW-1 SAMPLING NOTES (ug/l) VOCs (ug/l) 1,2,4-Trimethylbenzene 1000 1,3,5-Trimethylbenzene 1000 1,3,5-Trimethylbenzene 2-Chlorotuene 2-Chlorotuene 1000 Actone 5 Benzene 5 Chlorobenzene 5 cis-1,2-Dichloroethene 70	GW-2 (ug/l) 100000 1000 N/A	IDII	C DRIE NAMED IN	and a state of the	a da da da a	
SAMPLING NOTES     (ug/l)       VOCs (ug/l)     1,2,4-Trimethylbenzene     10000       1,3,5-Trimethylbenzene     1000       2.Chlorototuene     100       A-Isopropyltoluene     1000       Acetone     3000       Benzene     5       Chloroform     5       Chloroform     5       cis-I,2-Dichloroethene     70	(l/gu) 100000 N/A	200	3/12/2002	5/7/2002	8/1/2002	3/12/2002
VOCs (ug/l)     1,2,4-Trimethylbenzene     10000       1,3,5-Trimethylbenzene     1000       2.Chlorotoluene     100       4-Isopropyltoluene     1000       Acetone     3000       Benzene     5       Chlorobenzene     5       Chloroform     5       Chloroform     5       Chloroform     5	100000 1000 N/A	(l/gn)	and the second se			Dup. Of NC-3
1,2,4-Trimethylbenzene     10000       1,3,5-Trimethylbenzene     100       2.Chlorotoluene     N/A       4-Isopropyltoluene     3000       Acetone     5       Benzene     100       Chlorobenzene     5       Chloroform     5       cis-I,2-Dichloroethene     70	10000 1000 N/A	*				
1,3,5-Trimethytbenzene     100       2-Chlorototuene     N/A       2-Chlorototuene     1000       4-Isopropyltoluene     3000       Acetone     5       Benzene     5       Chlorobenzene     5       Chloroform     5       cis-1,2-Dichloroethene     70	1000 N/N	N/A	ND(500)	ND(500)	ND(500)	ND(500)
2-Chlorototuene N/A 4-Isopropyltoluene 1000 Acetone 3000 Benzene 5 Chlorobenzene 5 cis-1,2-Dichloroethene 70	N/N	N/A	ND(500)	ND(500)	ND(500)	ND(500)
4-Isopropyltoluene     1000       Acetone     3000       Benzene     5       Chlorobenzene     5       Chloroform     5       cis-1,2-Dichloroethene     70		N/A	ND(500)	ND(500)	ND(500)	ND(500)
Acetone 3000 Benzene 5 Chlorobenzene 100 Chloroform 5 cis-1,2-Dichloroethene 70	10000	N/A	ND(100)	ND(100)	ND(100)	(001)ON
Benzene 5 Chlorobenzene 100 Chloroform 5 cis-1,2-Dichloroethene 70	50000	100000	ND(1000)	ND(1000)	ND(1000)	ND(1000)
Chlorobenzene 100 Chloroform 5 cis-1,2-Dichloroethene 70	2000	70000	10000	20000	12000	10000
Chloroform 5 cis-1,2-Dichloroethene 70	500	10000	ND(100)	ND(100)	(001)GN	(001)QN
cis-1,2-Dichloroethene 70	400	100000	ND(150)	ND(150)	ND(150)	ND(150)
	30000	100000	ND(100)	ND(100)	ND(100)	ND(100)
trans-1,2-Dichloroethene 100	20000	100000	ND(150)	ND(150)	ND(150)	ND(150)
trans-1, 3-Dichloropropene N/A	N/A	N/A	ND(100)	ND(100)	ND(100)	ND(100)
Ethythenzene 700	4000	100000	1100	2200	1300	066
Isopropylbenzene 10000	100000	N/A	ND(100)	(001)QN	ND(100)	ND(100)
2-Butanone 400	50000	100000	ND(1000)	ND(1000)	ND(1000)	ND(1000)
Naphthalene 20	0009	60000	ND(500)	720	ND(500)	ND(500)
n-Butylbenzene N/A	N/A	N/A	ND(100)	(001)QN	ND(100)	(001)QN
n-Propylbenzene 1000	10000	N/A	ND(100)	ND(100)	(001)QN	ND(100)
m.p-Xylene 6000	0009	N/A	ND(100)	420	190	ND(100)
o-Xvlene 6000	6000	NIA	130	220	ND(100)	110
sec-Butylbenzene N/A	NIA	N/N	ND(100)	(001)QN	ND(100)	(001)(IN
Styrene 100	006	100000	ND(100)	(001)CIN	(001)QN	(001)(IN
tert-Butylbenzene 1000	10000	N/A	ND(500)	ND(500)	ND(500)	ND(500)
Toluene 1000	0009	100000	ND(150)	061	250	ND(150)
Tetrachloroethene 5	3000	50000	ND(100)	ND(100)	ND(100)	ND(100)
Trichloroethene 5	300	100000	(001)QN	ND(100)	ND(100)	ND(100)

-

6/27/2003

Page 18 of 18

TABLE IV SUMMARY OF SUPPLEMENTAL GROUNDWATER SAMPLING RESULTS 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

WELL/LOCATION ID	METHOD 1	METHOD I	METHOD 1	NC-3	NC-3	NC-3	NC-3
SAMPLE ID	MCP	MCP	MCP	NC-3	NC-3	NC-3	NC-DUP
SAMPLE DATE	GW-1	GW-2	ncr	3/12/2002	5/7/2002	8/1/2002	3/12/2002
SAMPLING NOTES	(I/Bn)	(l/gn)	(l/gn)				Dup. Of NC-3
SVOCs (ug/l)							
I-Methylnaphthalene	N/A	N/A	N/A	ND(130)	43	25	ND(53)
2-Methylnaphthalene	10	3000	100000	ND(210)	ND(16)	(91)(IN)	ND(84)
2-Methylphenol	5000	50000	N/A	ND(160)	ND(12)	ND(12)	ND(63)
Acenaphthene	20	5000	50000	ND(130)	(01)(IN)	ND(10)	ND(53)
Bis(2-ethylhexyl)phthalate	6	30	100000	ND(260)	ND(20)	ND(20)	ND(100)
Fluorene	300	3000	30000	ND(130)	(01)QN	ND(10)	ND(53)
Naphthalene	20	6000	60000	180	270	72	280
Phenol	4000	30000	100000	ND(180)	ND(14)	ND(14)	ND(74)
Metals (ug/l)							
Arsenic	50	400	4000	29	23	22	30
Barium	2000	30000	100000	120	110	110	120
Fluoride	N/A	N/A	N/A	;	:	;	ţ
Cyanide (ug/l)							
Cyanide, Total	10	10	N/A	210	221	157	217
Cyanide, Physiologically Available	N/A	N/A	2000	23	47	10	53

Notes:

1. ND (5): Compound not detected above detection limit; detection limit

+

is shown in parentheses.

2. VOCs: Volatile Organic Compounds; SVOCs: Semivolatile Organic Compounds;

Table lists detected compounds only; compounds that were not detected

are not listed.

 Surrogates were not recovered for samples 02B-B922-OW and 02B-B923-OW collected on 27 July 2002, due to high concentrations of native compounds (i.e., naphthalene) in the samples.

#### TABLE V

FORMER MALDEN MGP SITE

MALDEN, MASSACHUSETTS

			Quarterly Sa	npling Events			MADEP
	10/11/2001	1/10/2002	4/10/2002	6/25/2002	10/8/2002	1/24/2003	Background
VOCs (ppbV)							5 52
Site 2		neon S					
Benzene	-	0.71	ND(0.31)	ND(0.31)	ND(0.71)	0.6	6.6
Toluene		1.4	1.1	2.5	1.2	1.1	7.6
Ethylbenzene	-	ND(0.23)	0.52	ND(0.23)	ND(0.53)	ND(0.35)	2.2
m-&p-xylenes		0.74	ND(0.23)	0.64	0.36	0.57	9
Styrene	0.000	ND(0.23)	ND(0.23)	ND(0.23)	ND(0.54)	ND(0.36)	
o-xylenes	077	ND(0.23)	ND(0.23)	ND(0.23)	ND(0.53)	0.35	9
Naphthalene	077	ND(0.19)	ND(0.19)	ND(0.19)	ND(0.44)	ND(0.29)	1
Total Detected VOCs		2.85	1.62		1.6	2.6	
Site 4							
Benzene	ND(0.31)	ND(0.31)	ND(0.31)	0.44	ND(0.37)	ND(0.48)	6.6
Toluene	5.5	5.2	5.1	9.2	6.5	0.63	7.6
Ethylbenzene	0.44	ND(0.23)	ND(0.23)	0.71	ND(0.27)	ND(0.35)	2.2
m-&p-xylenes	1.3	1.0	0.56	2	0.59	ND(0.35)	9
Styrene	ND(0.23)	ND(0.23)	ND(0.23)	ND(0.23)	ND(0.27)	ND(0.36)	
o-xylenes	0.45	ND(0.23)	ND(0.23)	0.51	ND(0.27)	ND(0.35)	0
Naphthalene	ND(0.19)	ND(0.19)	ND(0.19)	ND(0.19)	ND(0.22)	ND(0.29)	i
Total Detected VOCs	7.7	6.2	5.7		7.1	0.6	
Site 5				2			
Benzene	ND(0.63)	ND(0.63)	ND(0.31)	ND(0.31)		ND(0.4)	6.6
Toluene	4.6	5.1	3.8	13		0.77	7.6
Ethylbenzene	ND(0.46)	ND(0.46)	0.31	0.57		ND(0.29)	2.2
m-&p-xylenes	0.81	1.1	1.0	1.5	224	0.36	0
Styrene	ND(0.47)	ND(0.47)	ND(0.23)	ND(0.23)		ND(0.30)	
o-xylenes	ND(0.46)	ND(0.46)	ND(0.23)	ND(0.23)		ND(0.29)	9
Naphthalene	ND(0.38)	ND(0.38)	ND(0.19)	ND(0.19)		ND(0.24)	1
Total Detected VOCs	5.4	6.2	5.1		22	1.1	5X
Site 6			1	60			
Benzene	3.4	12	ND(6.3)	19	ND(0.66)	ND(0.51)	6.6
Toluene	8.3	10	ND(5.3)	11	12	0.64	7.6
Ethylhenzene	ND(0.23)	ND(0.92)	ND(4.6)	ND(0.22)	ND/0 401	ND(0.27)	7.0
m-&p-xylenes	0.64	ND(0.92)	ND(4.6)	0.6	0.49	0.57	2.2
Styrene	ND(0.23)	ND(0.94)	ND(4.7)	ND(0.22)	ND(0.50)	ND/0 281	9
o-xylenes	ND(0.23)	ND(0.92)	ND(4.0)	NID(0.23)	ND(0.50)	ND(0.38)	
Naphthalene	ND(0.19)	ND(0.76)	ND(2.0)	ND(0.23)	ND(0.49)	ND(0.37)	9
Total Detected VOCs	12.3	22.0	ND(5.0)	ND(0.19)	10(0.40)	ND(0.31)	1
For Detected FOCS	12.5	22.0	ND		12.5	1.2	

NOTES AND ABBREVIATIONS:

 MADEP Background: Indoor air background levels obtained from Users Guide. Risk Assessment Shortform Residential Exposure Scenario. Version 1.6. October 1992. Policy #WSC/ORS-142-92. Adapted from Table 6-4, p. 33. Massachusetts Department of Environmental Protection. Office of Research and Standards and Bureau of Waste Site Cleanup. Boston, MA.

2. VOC (ppb): volatile organic compounds with values in parts per billion by volume; analyzed by EPA Method T014.

3. ND: compound not detected above quantitation limit, number in parentheses is the quantitation limit.

4. TR: compound detected below the quantitation limit, number in parentheses is the quantitaion limit;

5. Total Detected VOCs: Sum of detected concentrations of target analytes.

6. Bold values identify test results which exceed MADEP indoor air background levels.

1.

#### TABLE V

## SUMMARY OF SUPPLEMENTAL INDOOR AIR QUALITY DATA - OCTOBER 2001 THROUGH JUNE 2002 129 COMMERCIAL STREET PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

			Quarterly Sa	npling Events			MADEP Indoor Air
	10/11/2001	1/10/2002	4/10/2002	6/25/2002	10/8/2002	1/24/2003	Backgroun
VOCs (ppbV)			100	· · · · · · · · · · · · · · · · · · ·			Section Control State
Site 7	1						1000000
Benzene	3.6	14	1.4	1.8	ND(0.62)	ND(0.70)	6.6
Toluene	2.9	3.5	3.0	5	2.3	0.6	7.6
Ethylbenzene	ND(0.23)	ND(0.92)	ND(0.46)	ND(0.23)	ND(0.46)	ND(0.51)	2.2
n-&p-xylenes	0.58	ND(0.92)	ND(0.46)	0.68	ND(0.46)	ND(0.51)	9
Styrene	ND(0.23)	ND(0.94)	ND(0.47)	ND(0.23)	ND(0.47)	ND(0.52)	•
o-xylenes	ND(0.23)	ND(0.92)	ND(0.45)	ND(0.23)	ND(0.46)	ND(0.51)	9
Naphthalene	ND(0.19)	ND(0.76)	ND(0.38)	ND(0.19)	ND(0.38)	ND(0.42)	1
Total Detected VOCs	7.08	17.5	4.4		2.3	0.6	
Site 8							
Benzene	3.9	15	ND(6.3)	0.97	ND(0.62)	ND(0.57)	6.6
Toluene	2.1	2.8	ND(5.3)	2.3	1.8	ND(0.48)	7.6
Ethylbenzene	ND(0.46)	ND(1.8)	ND(4.6)	ND(0.46)	ND(0.46)	ND(0.42)	2.2
m-&p-xylenes	0.68	ND(1.8)	ND(4.6)	0.68	ND(0.46)	ND(0.42)	9
Styrene	ND(0.47)	ND(1.9)	ND(4.7)	ND(0.47)	ND(0.47)	ND(0.43)	
o-xylenes	ND(0.46)	ND(1.8)	ND(4.6)	ND(0.46)	ND(0.46)	ND(0.42)	9
Naphthalene	ND(0.38)	ND(1.5)	ND(3.8)	ND(0.38)	ND(0.38)	ND(0.35)	1
Total Detected VOCs	6.68	17.8	ND	~	1.8	ND	225
Site 10	10						
Benzene	2.4	10	ND(6.3)	1.7	0.6	ND(0.55)	6.6
Toluene	2.2	2.9	ND(5.3)	2.2	2.5	0.54	7.6
Ethylbenzene	0.74	ND(1.8)	ND(4.6)	0.71	0.82	ND(0.40)	2.2
m-&p-xylenes	1.2	ND(1.8)	ND(4.6)	1.3	1.4	0.58	9
Styrene	3.7	2.1	ND(4.7)	4.7	5.5	1	-
o-xylenes	0.51	ND(1.8)	ND(4.6)	0.51	0.57	ND(0.90)	9
Naphthalene	ND(0.19)	ND(1.5)	ND(3.8)	ND(0.19)	ND(0.32)	ND(0.33)	1
<b>Total Detected VOCs</b>	10.75	15	ND	0.001/000-0400-040	11.4	2.1	

## NOTES AND ABBREVIATIONS:

 MADEP Background: Indoor air background levels obtained from Users Guide. Risk Assessment Shortform Residential Exposure Scenario. Version 1.6. October 1992. Policy #WSC/ORS-142-92. Adapted from Table 6-4, p. 33. Massachusetts Department of Environmental Protection, Office of Research and Standards and Bureau of Waste Site Cleanup. Boston, MA.

2. VOC (ppb): volatile organic compounds with values in parts per billion by volume; analyzed by EPA Method T014.

3. ND: compound not detected above quantitation limit, number in parentheses is the quantitation limit.

4. TR: compound detected below the quantitation limit, number in parentheses is the quantitaion limit;

5. Total Detected VOCs: Sum of detected concentrations of target analytes.

6. Bold values identify test results which exceed MADEP indoor air background levels.

TABLE XI INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 1: NORTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

Process Option     Description     DNAPL Extraction using Recovery Wells: Installation of large-     Is the technology reasonably li     diameter DNAPL extraction wells, equipped with a submersible pump to Are experts available to imple
Extraction using Recovery Wells: Installation of large- r DNAPL extraction wells, equipped with a submersible pump to DNAPL from the subsurface. Extracted DNAPL stored in tanks mall buildings on the property and transported off-site for
Installation of large-       Is the technology reasonably list         sing Recovery Wells: Installation of large-       Is the technology reasonably list         action wells, equipped with a submersible pump to       Are experts available to implete         he subsurface. Extracted DNAPL stored in tanks       Comments/Concerns/Explanat         on the property and transported off-site for       existing DNAPL extraction we
Installation of large-       Is the technology reasonably ling         I with a submersible pump to cited DNAPL stored in tanks       Are experts available to impletee to impletee to comments/Concerns/Explanatee to comments/Explanatee to comments/Concerns/Explanateee to comm
Is the technology reasonably li np to Are experts available to imple anks Comments/Concerns/Explanat existing DNAPL extraction we exist available to imple nd Are experts available to imple very Comments/Concerns/Explanat vould odors, space for excavation, a
nology reasonably li is available to imple i/Concerns/Explanat NAPL extraction we nology reasonably li s available to imple concerns/Explanat
ilability of Experti manent or Tempora Yes. plicable to both sha ting successfully or manent or Tempora Yes. Yes. Yes. ignificantly more d

	Initial Screet Outcome
reasonably likely to be feasible based on E (Area 2).	Retain
nstall, due to subsurface obstructions,	Retain
Area 1, and potential remains for e of subsurface utilities.	Retain
d from the subsurface,	Retain
sive methods or belt-skimmers. ction system is not currently justified.	Eliminate
on which gradual LNAPL removal is erefore the need for an LNAPL extraction	Eliminate
has been attempted on the Site, however recent monitoring; therefore the need	Eliminate

Haley & Aldrich, Inc. G:\06558\634 Phase IINPh\_III\_Initial\_Screen\_Tables.xls

Taroet Media	Technology	Process Option Description	Feasibility of Achieving a Permanent or Temporary Seasibility of Expertise, and Other Comments
Shallow and Deep DNAPL/ LNAPL	Monitoring	<b>DNAPL or LNAPL Monitoring:</b> Conduct semi-annual monitoring of NAPL thickness in monitoring wells using oil/water interface probes. Place monitoring wells strategically to detect off-site migration.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to imprement the technology? Yes. Comments/Concerns/Explanation: NAPL monitoring may be an applicable component for a Tem LNAPL or deep DNAPL are not present.
Soil/TSM	No Action	Natural Attenuation: Allow natural processes such as biodegradation and volatilization to reduce concentrations or toxicity of contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This approach is not applicable for remediation of soil contain
Soil/TSM	Physical removal	Excavation and off-site treatment: Excavate contaminated soil and DNAPL/LNAPL, where present, and transport off-site for treatment.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Excavation of TSM would likely be difficult due to the general buildings, disposal of excavated soil, etc.
Soil/TSM	Capping	Installation of Engineered Barrier: RCRA-type engineered barrier consistent with 310 CMR <sup>-40</sup> . "Excavate 3 feet of soil, install barrier; backfill soil and re-vegetate on top of barrier.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes Comments/Concerns/Explanation: Installation of an engineered barrier may limit future developm
Soil/TSM	Capping	Installation/maintenance of a direct contact barrier: Prevent potential exposure to shallow soils by maintaining 3 feet of clean soil or asphalt cover.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Maintenance of a direct contact barrier would likely be require exposure to contaminated soils.
Soil/TSM	Ex-situ treatment	Incineration: Treat excavated material off-site using incineration.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes, Are experts available to implement the technology? Yes. Commenis/Concerns/Explanation: Would be appropriate for treatment of extracted LNAPL and E hermal desorption treatment.
Soil/TSM	Ex-situ treatment	Thermal desorption: Treat excavated soil material using thermal desorption with an afterburner. May be performed on-site using a mobile thermal desorption unit, or material may be sent off-site for treatment.	is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: On-site thermal desorption unlikely to be reasonably feasible d missions. Off-site thermal desorption is reasonably feasible.
Soil/TSM	Ex-situ treatment	Soil washing: Separate silt and clay fraction from sand fraction, and desorb contaminants into the aqueous phase.	is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is unlikely to be effective at this site; TSM is usahing.
Soil/TSM	Ex-silu treatment	Stabilization: Reduce leachability of organic contaminants through physical encapsulation and/or cement stabilization. Treated soil may be re-used as backfill on the site or sent to appropriate receiving facilities	s the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not reasonably feasible due to the anticipated volume of soil to

1

TABLE XI INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 1: NORTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN

	ated and the limited space available on
Eliminate	y to be removed from soil through
Retain	imited available space, odors, and air
Retain	, or soils that are not suitable for
Retain	art of an AUL to restrict potential
· · · Retain -	tions for the Area.
Retain	odors, support of excavations and
Eliminate	h PAH concentrations or TSM.
Retain	or Permanent Solution, if recoverable
Initial Screeni Outcome	

ľ

12

9. .

.

9 (gi)

1

85 (H

Haley & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xls

Target Medi	Technology a Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary Se Availability of Expertise, and Other Comments
Soil/TSM	Ex-situ treatment	Asphalt batching: Treat excavated soil by mixing with asphalt emulsion to create asphaltic roadbase material. Process may be conducted on- or off-site, and asphalt product may be re-used on-site or off-site.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: May be appropriate treatment method for some impacted soil. considered feasible for MGP waste, as it is not compatible with hazardous wastes or soil containing selected TSM samples have failed TCLP due to elevated benzene concentrations.
Soil/TSM	Ex-situ treatment	Landfarming: Excavated material is spread on the ground surface and is periodically tilled and aerated to promote biological removal of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not feasible due to urban setting of the site.
Soil/TSM	Ex-situ treatment	Bioslurry Reactors: Excavated soil material is slurried with water and nutrients and is placed in a stirred, aerated reactor to promote biodegradation of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: High concentrations of contaminants limit the feasibility of this
Soil/TSM	Ex-situ treatment	MGP-REM Process: Chemically treat recalcitrant organics with Fenton's reagent (ferrous iron and hydrogen peroxide) to chemically oxidize and biodegrade organic contaminants. May be conducted as either a landfarming or bioreactor process.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This treatment method is not feasible for Area 1 due to the rela the potential for emissions that may impact the public.
Soil/TSM	In-situ treatment	In-situ thermal desorption: Application of heat and vacuum to in-situ soil to volatilize, destroy and extract organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible due to the presence of subsurfac Additionally, the application of this technology in the presence of DNAPL is unreliable.
Soil/TSM	In-situ ่น caiurciti	In-situ stabilization: Mix soil in-situ with stabilizing agents using large- diameter hollow-stem auger drilling .	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible on the scale required at the Site, limit the feasibility of this technology.
Soil/TSM	In-situ treatment	Bioventing (unsaturated zone soils): Inject air into vadose zone soils to stimulate aerobic degradation of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible for high concentrations of PAH
Soil/TSM	In-situ treatment	In-situ chemical oxidation: Injection of hydrogen peroxide mixture into in-situ soil using geoprobe equipment or wells to chemically break down organic contaminants to innocuous compounds.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Feasible remedial technology for MGP and petroleum-impacted preferable due to the public location of Area 1. Injection of chemical oxidant into the soil beneath generate excessive heat and wapors - ventilation may be required.
Soil/TSM	Restrictions	Activity and Use Limitations (AULs): Place restrictions on future property use to prevent exposure through pathways for which a condition of No Significant Risk could not be established.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be implemented to prevent potential future exposu

TABLE XI INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA I: NORTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN

6/27/2003

Retain	In-situ treatment method may be	soil
Eliminate	amination.	conta
Eliminate	subsurface obstructions on Area 1 also	and
Eliminate	lifies and the urban setting of the Site.	
Eliminate	y small area available for treatment and	ively
Eliminate	nology.	techn
Eliminate		
Eliminate	vever, this process is generally not ee product or cyanide. TCLP analyses on	Hown g free
Initial Screenin Outcome	Ţ	ution

7 20

TABLE XI INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 1: NORTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

Target Medi	Technology a Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary So Availability of Expertise, and Other Comments
Soil/TSM	Restrictions	Fencing: Limit site access using fencing.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Although Area 1 is located in a relatively public commercial ar property is currently limited with a fence surrounding a portion of the property. Additional restric feasible.
Soil/TSM	Disposal	Off-site Disposal: Dispose non-hazardous soils off-site in a landfill.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This option is considered less favorable than thermal treatment disposal. Because off-site thermal desorption treatment is retained, off-site disposal is not necessal Solution.

Haley & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xls

ution, Initial Screening Outcome a, public access to the 100 Commercial Street tions to public access are not reasonably pptions, which treat contaminated soil prior to y to achieve a Temporary or Permanent	100	
ution, Initial Screening Outcome a, public access to the 100 Commercial Street tions to public access are not reasonably Eliminate	Eliminate	ptions, which treat contaminated soil prior to y to achieve a Temporary or Permanent
ution, Initial Screening Outcome	Eliminate	a, public access to the 100 Commercial Street tions to public access are not reasonably
	nitial Screening Outcome	ution,

6/27/2003

 $\mathbf{\hat{x}}$ 

TABLE XIII DETAILED EVALUATION OF REMEDIAL ACTION ALTERNATIVES AREA 1 - NORTHERN PORTION OF PARCEL E PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

REMEDIAL ACTION ALTERNATIVE	DESCRIPTION OF REMEDIAL ACTION ALTERNATIVE	EFFECTIVENESS	RELIABILITY (SHORT-TERM/LONG-TERM)	IMPLEMENTABILITY	COST (NPV)	RISKS	
Alternative 1-1:	Excavation of impacted soil to remove	High. Excavation of soil is likely	Short term: High, excavation of	Difficult.	Selected Components:	High. Risks to workers and /	Achie
- Excavation of	shallow DNAPL, LNAPL, TSM, and	to be an effective method for	contaminated soil has a high	- Large excavation area	TSM Excavation: \$1.3M	passersby due to inhalation of F	erma
DNAPL, LNAPL,	petroleum impacts. Transport	removal of the sources of LNAPL,	probability of removing impacted	(approximately 5100 sy; 20,250 cu	Excavation of petroleum-	vapors during excavation; r	elativ
ISM and petroleum-	contaminated soil off-site for thermal	DNAPL and contaminated soul	soil and shallow DNAPL.	yd)	Impacied Soll: 32.0M	would require using an	Tanic
- Deen DNAPL	desorption treatment, and backfill with	Likely to achieve a Permanent Solution for the Area	Long-term: High, likely to be	<ul> <li>High degree of disruption to Malden Center Area (construction</li> </ul>	migration control: \$50K	and Level B protection	
monitoring	clean soil. Monitor LNAPL and deep		source removal. Vertical barriers	noise, potential odors, truck traffic)	Engineering/Design:	measures (i.e., supplied air)	
- Shallow DNAPL	DNAPL, control shallow DNAPL		to DNAPL migration onto the	- Limited access to property during	\$260K	for workers.	
migration control	migration onto the Area from adjacent		Area wili prevent future re-	remediation			
- AUL	Areas if necessary with a vertical		contamination of soil.	- Level B respiratory protection	NPV: \$4.9M		
	subsurface barrier (i.e., sheet piling or			may be required for workers			
	Waterloo Barrier) located on Area 1.			- Construction of a temporary			
				enclosure to prevent escape of odors			
				Bycavarian of sail from beneath			
				the maintenance garage would			
				require demolition of the building.			
				- Dewatering of the excavation and			
				water treatment would be required	-		
Alternative 1-2:	Excavation to remove shallow DNAPL,	High. Excavation of soil is likely	Short term: High, both	Difficult.	Selected Components:	High. Risks to workers and A	Ichie
- Excavation of	LNAPL, and TSM contamination.	to be an effective method for	excavation and in-situ chemical	<ul> <li>Moderate excavation size</li> </ul>	TSM Excavation: \$1.3M	passersby due to inhalation of P	erma
shallow DNAPL,	Fransport contaminated soil material	removal of the sources of LNAPL,	oxidation have been used	(approximately 1900 sy, 7450 cu.	In-situ chemical	vapors during excavation; n	elativ
LNAPL and TSM soil of	off-site for thermal desorption	DNAPL and contaminated soil.	effectively at other MGP and	yd)	oxidation of petroleum-	would require using an fi	rame.
- In-situ chemical t	reatment, and backfill excavation with	Chemical oxidation has been	petroleum contamination sites to	- Highly disruptive to Malden	impacted soil: \$320K	enclosure during excavation. in	njecti
oxidation of petroleum-	acceptable excavated soil or imported	successfully at other sites for in-	remediate soil.	Center area (construction noise,	Shallow DNAPL	and Level B protection o	ontan
impacted soil	clean soil. Monitor LNAPL and deep	situ reduction of concentrations of	Long term: High, due to removal	potential odors, truck traffic)	migration control: \$50K	measures (i.e., supplied air) in	ntrusi
- Deep DNAPL	DNAPL, inject chemical oxidant into	petroleum constituents in soil.	of DNAPL source material, and	- Use of the property would be	Engineering/Design:	for workers; risks to workers N	fay n
monitoring	etroleum-impacted soils to promote in-	Likely to achieve a Permanent	use of an accepted method of in-	limited during excavation.	3100K	nangling oxidant during	OIUTIC
Shallow Divini	nile (Assume 3 intertions required)	been used at other slice as an	situ treatment. Vertical barriers to	- Level B respiratory protection	NPV: \$2.3M	excavation, risk to subsurface to structures (e.g., maintenance to	) less
- AUL	Control shallow DNAPL migration (if	effective method for limiting	will prevent future re-	- Would require an enclosure		garage foundation) due to	
2	necessary) onto the Area with a	exposure to contaminated soil	contamination of soil.	- Would require dewatering of		vapor buildup during oxidant	
5	ubsurface barrier located on Area 1.	remaining in place.		excavation and water treatment		injection.	
	An AUL involving the maintenance of a	53 88		- Injection of chemical oxidant			
6	lirect contact barrier (i.e., pavement)			would be readily implementable,			
-	nay be used to restrict future exposure			although injection beneath the			
	o contaminated soils.			maintenance garage may be difficult.			

Haley & Aldrich, Inc. G./06556/834 Phase III/Ph\_III\_Det\_Eval\_Tables

TABLE XIII DETAILED EVALUATION OF REMEDIAL ACTION ALTERNATIVES AREA 1 - NORTHERN PORTION OF PARCEL E PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

REMEDIAL ACTION	ACTION ALTERNATIVE	EFFECTIVENESS	RELIABILITY (SHORT-TERM/LONG-TERM)	IMPLEMENTABILITY	(NPV)	Nono
Alternative 1-3: - Installation of an Engineered Barrier - Shallow DNAPL recovery wells - LNAPL monitoring - Provisional shallow DNAPL migration control - AUL - AUL	Excavation of shallow surface soil, and construction of Engineered Barrier over TSM and petroleum-impacted soil. Installation of 5 shallow DNAPL recovery wells, monitoring of LNAPL and deep DNAPL and, if necessary, a vertical subsurface barrier located on Area 1 to prevent DNAPL migration onto the property. A DNAPL storage building would be required on the property to house DNAPL storage. An AUL involving the maintenance of the engineered barrier may be used to restrict future exposure to contaminated soils.	Moderate. Engineered Barrier eliminates potential for exposure to contaminated soils. DNAPL recovery wells are likely to be effective for DNAPL extraction. Would achieve a Temporary Solution in the near term, and would eventually result in a Permanent Solution, once DNAPL removal is complete. AULs have been used at other sites as an effective method for limiting exposure to contaminated soil remaining in place.	Short term: Moderate. Engineered Barrier eliminates potential soil exposure pathways, but does not eliminate NAPL- related UCLs. DNAPL recovery wells provide reliable but slow DNAPL recovery; A DNAPL extraction well has functioned reliably on Area 2 since October 2001. Long term: Moderate, Would likely result in a Permanent Solution over the longer term (estimated 7 years), once DNAPL remedial goals have bet met. Vertical barriers to the flow of DNAPL onto the Area will prevent future re-contamination of soil.	Moderately Difficult. Installation of an Engineered Barrier requires excavation to a depth of 3 feet over a large portion of the property, which would be highly disruptive. Installation would likely require several months to complete, during which time use of the property would be severely limited. Would also likely require clean ccrridors for utility access, and may involve re- installation of utility lines across the property.	Selected Components: Engineered Barrier Installation: \$1.1M Shallow DNAPL Recovery System Installation: \$100K Shallow DNAPL Imigration control: \$20K Engineering/Design: \$20K O & M: \$180K NPV: \$2.1M	Low. Contaminants (TSM, DNAPL, LNAPL) are located deeper than 3 feet, therefore contaminated material during construction is anticipated to be low. Risks associated with installation of extraction wells anticipated to be low. anticipated to be low.
Alternative 1-4: - Shallow DNAPL recovery wells - Monitoring of LNAPL and deep DNAPL - Provisional Shallow DNAPL migration control - In-situ chemical oxidation of TSM and petroleum - impacted soil - AUL	This alternative involves the installation of 5 shallow DNAPL extraction wells, construction of a DNAPL storage building and monitoring of LNAPL and deep DNAPL. When appropriate (i.e., TSM soil is accessible and/or DNAPL has been reduced to less than 1/2 inch in wells) in-situ chemical oxidation would be conducted in areas of contaminated soil. If necessary. DNAPL migration onto the property would be prevented using a subsurface harrier located on Area 1. An AUL involving the maintenance of a direct contact barrier (i.e., pavement) may be used to restrict future exposure to contaminated soils.	Moderate. May eventually achieve Permanent Solution. DNAPL recovery has been demonstrated to be effective at one location on Area 2, and chemical oxidation has been used effectively at other MGP and petroleum contamination sites to reduce concentrations of VOCs and PAHs in soit. Chemical ovi <u>dation would</u> not be effective in TSM areas until the quantity of DNAPL has been reduced. AULs have been used at other sites as an effective method for limiting exposure to contaminated soil remaining in place.	Short term: Moderate, DNAPL extraction well has functioned reliably on Area 2 since October 2001. Extraction system will require maintenance to continue to operate. Unlikely to accomplish remedial goals in short term. Long-term: Moderate, DNAPL extraction system will require long term maintenance and replacement of parts to continue to operate.	Readily Implementable. This alternative requires relatively little intrusion into current facility operations. If required, injection of chemical oxidants beneath the maintenance garage may require either injection through the garage floor slab, or demolition of the garage structure.	Selected Components: Shallow DNAPL Recovery System Installation: \$100K Chemical Oxidation of Soil: \$430K Shallow DNAPL migration control: \$50K Engineering/Design: \$150K O & M: \$180K NPV: \$1.1M	Moderate. Relatively little risk associated with the installation of extraction wells. Risks to workers during injection of oxidants for chemical oxidation. Some risk of damage to the WEB Culvert, the matintenance garage foundation, and subsurface utilities during injection of chemical oxidants due to potential for heat and vapor buildup during injection.

Haley & Aldrich, Inc. G:/06558/634 Phase HJ/Ph\_III\_Det\_Eval\_Tables

BENETTS         TIMELINESS         EFFECT ON NON- PECUNIARY INTERESTS           salt in Permanent nonce DNAPL         Requires Log.term Implementation.         Moderatt. Installation of the Engineered Barrier on the stalled.           Interactions to operations to end maintain. An this Area. In-situ chemical ovidation of soils would require a additional 1 to 2         Low. Would require a this alternative.           Nover require a al market soils.         Therefore, the estimated time to attain the remedial goals is 6 to 9 years.         Low. Barrier is this alternative.			
salt in a Permanent         Requires Long-term         Moderate.         Installation of the Engineered Barrier on the Engineered Barrier on the Engineered Barrier on the Engineered Barrier on the completion is approximately be approximately 7 6 to 9 years.         Moderate.         Installation of the Engineered Barrier on the property would require several months to complete; site use would be severely limited during this time. Much less than in alternatives 1-1 and 1- 2; less excavation of heavily impacted material, less muisance odor, less truck traffic. Options for future is mainiarined.           ternative is mainiarined.         Requires Long-term Implementation.         Low. Would require relatively impacted material, less muisance odor, less truck traffic. Options for future development are limited once the barrier is installed.           ternative is mainiarined of rounding Malden areea) and trequires terrapion of intermedial goals with respect intermy to require a approximately 5 to Tyears is using extraction wells in operations to install and require a matition of solls would intact solls.         Low. Would require relatively intermative. Therefore, the estimated inte a of using extraction wells in goals is 6 to 9 years.	BENEFITS	TIMELINESS	EFFECT ON NON- PECUNIARY INTERESTS
ternative isRequires Long-termLow. Would require relativelyly unobrusive (to operations and to rounding Malden area) and requiresImplementation. remedial goals with respect to DNAPL thickness is likely to require approximately 5 to 7 years this Area. In-situ chemical oxidation of souls would require an additional 1 to 2 years to implement. Therefore, the estimated goals is 6 to 9 years.Low. Would require relatively little intrusion to facility operations to install and maintain. Does require a DNAPL storage shed for storage of extracted DNAPL, Also relatively little disturbance to surrounding this Area. In-situ chemical oxidation of souls would inated soils.Low. Would require relatively operations to implement. Therefore, the estimated time to attain the remedial goals is 6 to 9 years.Low. Would require relatively intervent.	sult in a Permanent on once DNAPL al is complete, gh this is estimated ine approximately 7 Does not involve tion of highly inated soil and ore is less disruptive Iternatives 1-1 and Iternatives 1-1 and Iter	Requires Long-term Implementation. Estimated time for completion is approximately 6 to 9 years. 6 to 9 years.	Moderate. Installation of the Engineered Barrier on the property would require several months to complete; site use would be severely limited during this time. Much less impact on the surrounding area than in alternatives 1-1 and 1- 2; less excavation of heavily impacted material, less nuisance odor, less truck traffic. Options for future development are limited once the barrier is installed.
	ternative is ely unobtrusive (to operations and to rounding Malden area) and requires tterruption of operations to operations to and maintain. An would limit the ai for exposure to ninated soils.	Requires Long-term Implementation. Accomplishment of remedial goals with respect to DNAPL thickness is likely to require approximately 5 to 7 years using extraction wells in this Area. In-situ chemical oxidation of soils would require an additional 1 to 2 years to implement. Therefore, the estimated time to attain the remedial goals is 6 to 9 years.	Low. Would require relatively little intrusion to facility operations to install and maintain. Does require a DNAPL storage shed for storage of extracted DNAPL, Also relatively little disturbance to surrounding community of Malden under this alternative.

6/27/2003

• •

TABLE XIII DETAILED EVALUATION OF REMEDIAL ACTION ALTERNATIVES AREA 1 - NORTHERN PORTION OF PARCEL E PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

÷,

×.

Alternative 1-6:       This alternative involves c         - LNAPL and deep       monitoring of subsurface (         DNAPL monitoring       ensure that DNAPL locate         - AUL       off-site. An AUL involvin maintenance of a direct co         (i.e., pavement) may be us       first contenance of a direct co	Alternative 1-5:       This Alternative is less in covery         - Shallow DNAPL recovery       Alternatives 1-1 through involves the installation on DNAPL monitoring deep DNAPL recovery wells, 1         - Provisional Shallow       require the construction of DNAPL migration of store extracted DNAPL migration of necessary, migration of D the property could be preventhe installation of a subsurie on Area 1. An AUL invol maintenance of a direct co (i.e., pavement) may be u future exposure to contamination.	REMEDIAL ACTION DESCRIPTION OF R ALTERNATIVE ACTION ALTERN
ontinued conditions to not migrate not migrate ug the mact barrier nated soils.	rusive than 4. It F 5 shallow .NAPL and and would f a small shed and would f a small shed race barrier ving the ntact barrier ving the ntact barrier sed to restrict inated soils.	EMEDIAL
Low. Would not result in reduction of NAPL thickness or reduction in contaminant concentrations in soil. Therefore would not result in a Permanent Solution. Would achieve a Temporary Solution in he short term. AULs have been used at other sites as an effective nethod for limiting exposure to contaminated soil remaining in	Low. Would result in reduction of NAPL thickness over approximately 7 years. However, would not result in a Permanent Solution because soil containing concentrations of contaminants greater than the UCLs would remain. Would achieve a Temporary Solution in the short term. AULs have been used at other sites as an effective method for limiting exposure to contaminated soil remaining in place.	EFFECTIVENESS
Not Applicable: No remedy is installed under this alternative.	Short term: Moderate, DNAP extraction well has functioned reliably on Area 2 since October 2001. Extraction system will require maintenance to continue to operate. Unlikely to accomplish remedial goals in short term. Long-term: Low, Extraction system will require long-term maintenance and replacement of parts to continue to operate. Likely to accomplish remedial goals with respect to reduction in NAPL thickness; however this alternative will not reduce contaminant concentrations in soil.	RELIABILITY (SHORT-TERM/LONG-TERM
Readily Implementable. This alternative requires the least intrusion into facility operations to implement.	L Readily Implementable. This alternative requires the least intrusion into facility operations to implement among alternatives 1-1 through 1-4. p	IMPLEMENT'A BILITY
NPV: \$110K	Selected Components: n Shallow DNAPL Recovery System Installation: \$100K Shallow DNAPL migration control: \$50K Engineering/Design: \$90K O & M: \$130K NPV: \$540K	COST (NPV)
Low. Few risks associated with this alternative.	Low. Few risks associated with this alternative. Some level of risk associated with handling of extracted DNAPL. DNAPL.	RISKS
Requires 1 of facility	Requires of facility accomplis DNAPL s from the r Removes presence to the long to would lim for exposi- contamina	B

Cost estimates include contractor mobilizations, contingencies and other additional components in addition to selected components listed; Refer to Cost Estimate Tables in Appendix D for detailed breakdown of cost components.
 Cost estimates represent an opinion of probable cost for comparison purposes.

Haley & Aldrich, Inc. G/0655&634 Phase III/Ph\_III\_Det\_Eval\_Tables

ENEFITS	TIMELINESS	EFFECT ON NON- PECUNIARY INTERESTS
little interruption y operations, and shes removal of source material subsurface. UCLs due to of NAPL over cerm. An AUL nit the potential ure to arted soils.	a Requires Long-term Implementation. Expected to require approximately 5 to 7 years to accomplish DNAPL reduction remedial goals. Not expected to reduce soil contaminant concentrations to less than UCLs.	Low. This alternative results in little disruption to facility operations. Minimal impacts include the need for a DNAPL storage shed on the property, and maintenance work on the DNAPL extraction wells and pumps on the property. As with Alternative 1-4, this alternative results in minimal disruption to the surrounding neighborhood.
operations.	Requires Long-term Implementation. Monitoring would continue for the foresecable future until a Permanent Solution can be implemented.	Low. This alternative results in little disruption to facility operations. As with Alternatives 1-4 and 1-5, this alternative results in minimal disruption to the surrounding neighborhood.

Haley & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xis

	8	ALTER AND ADDRESS AND ADDRESS	
Target Media	Technology 1 Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary Availability of Expertise, and Other Commen
DNAPL (Shallow and Deep)	Extraction	<b>DNAPL Extraction using Recovery Wells:</b> Installation of large- diameter DNAPL extraction wells, equipped with a submersible pump to extract DNAPL from the subsurface. Extracted DNAPL stored in tanks inside small buildings on the property and transported off-site for disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Technology is applicable to both shallow and deep DNAPL. existing DNAPL extraction well that has been operating successfully on the southern portion of
DNAPL	Extraction/ Migration contro	DNAPL Extraction/Migration Control using Trenches: Excavation of ltrenches that are keyed into the low-permeability organic deposit, and backfilled with gravel for collection of DNAPL. One or more recovery wells equipped with a submersible pump for extraction of DNAPL would be installed in the trench.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes, Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be significantly more difficult than extraction v odors, space for excavation, and disposal of excavated soil material.
DNAPL	Migration control	Vertical Subsurface Barrier: Install barrier to DNAPL flow. May be slurry wall, sheet piling, or Waterloo <sup>TM</sup> barrier, dependent upon Phase IV design. Would also involve injection of grout through the base of the WEB Culvert to prevent DNAPL migration beneath the culvert.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be required in the event that DNAPL is removi migration from an adjacent area.
Shallow & Deep DNAPL/ LNAPL	Disposal/ Incineration	Disposal or Incineration: Off-site transport of extracted NAPL for incineration or disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be required in the event that DNAPL or LNAPL are
LNAPL	Extraction	Multi-phase extraction: Extraction of groundwater, soil vapor and LNAPL through the application of a vacuum to specially designed extraction wells. Requires separation of LNAPL from groundwater and ex-situ treatment of both soil vapor and groundwater.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Appropriate for LNAPL removal, and removal of volatile co
LNAPL	Extraction	Belt-skimmer: Extraction of LNAPL using belt-skimmers, consisting of two pulleys that drive a hydrophobic belt through the water table in a monitoring well, bring LNAPL oil to the surface, and skim the oil into a collection container. System is driven by an electric motor.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Appropriate for LNAPL removal in areas of high public acce

Retain	ed from the subsurface.
Retain	n Area 2 and potential remains for DNAPL
Retain	o install, due to subsurface obstructions,
Retain	onably likely to be feasible based on an I E (Area 2).
Dutcome	on,

Hatey & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xls

Target Media	Technology Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary Availability of Expertise, and Other Commen
LNAPL	Extraction	Passive LNAPL extraction methods: LNAPL extraction methods that rely on siphons, absorbent pads, or other passive methods for LNAPL collection.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Passive methods of LNAPL removal (absorbent pads, siphon the past, however rates of £NAPL extraction have not been consistent with remedial goals.
DNAPL/ LNAPL	Monitoring	DNAPL or LNAPL Monitoring: Conduct semi-annual monitoring of NAPL thickness in monitoring wells using oil/water interface probes. Place monitoring wells strategically to detect off-site migration.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: NAPL monitoring may be an applicable component for a Ter
Soil/TSM	No Action	Natural Attenuation: Allow natural processes such as biodegradation and volatilization to reduce concentrations or toxicity of contaminants	Is the technology reasonabily likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This approach is not applicable for remediation of soil contai
Soil/TSM	Physical removal	Excavation and off-site treatment: Excavate contaminated soil and DNAPL/LNAPL, where present, and transport off-site for treatment.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Excavation of TSM would likely be difficult due to the gener buildings, disposal of excavated soil, etc.
Soil/TSM	Capping	Installation of Engineered Barrier: RCRA-type engineered barrier consistent with 310 CMR 40. Excavate 3 feet of soil, install barrier, backfill soil and re-vegetate on top of barrier.	Is the technology reasonabiy likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Installation of an engineered barrier would eliminate potential this technology would not attain a Permanent Solution due to the presence of the 100 Commercia
Soil/TSM	Capping	Installation/maintenance of a direct contact barrier: Prevent potential exposure to shallow soils by maintaining 3 feet of clean soil or asphalt cover.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Maintenance of a direct contact barrier would likely be require exposure to contaminated soils.
Soil/TSM	Ex-situ treatment	Incineration: Treat excavated material off-site using incineration	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be appropriate for treatment of extracted LNAPL and thermal desorption treatment.
Soil/TSM	Ex-situ treatment	Thermal desorption: Treat excavated soil using thermal desorption with an afterburner. May be performed on-site using a mobile thermal desorption unit, or material may be sent off-site for treatment.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: On-site thermal desorption unlikely to be reasonably feasible emissions. Off-site thermal desorption is reasonably feasible.

TABLE XIV INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 2: SOUTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

6/27/2003

4

y Solution,	Initial Screen Outcome
ons without a pump) have been tried on the Site in	Eliminate
emporary or Permanent Solution.	Eliminate
aining high concentrations of PAH compounds.	Eliminate
eration of odors, support of excavations and	Retain
ial for exposure to contaminated soil. However, cial Street building on Area 2.	Retain
aired as part of an AUL to restrict potential	Retain
d DNAPL, or soils that are not suitable for	Retain
a due to limited available ename odore and air	Retain

75

a 2

Haley & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xts

Target Medi	Technology ia Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary ( Availability of Expertise, and Other Comment
Soil/TSM	Ex-situ treatment	Soil washing: Separate silt and clay fraction from sand fraction, and desorb contaminants into the aqueous phase.	is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is unlikely to be effective at this site; TSM is washing.
Soil/TSM	Ex-situ treatment	Stabilization: Reduce leachability of organic contaminants through physical encapsulation and/or cement stabilization. Treated soil may be re-used as backfill on the site or sent to appropriate receiving facilities for disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not reasonably feasible due to the anticipated volume of soil the Area.
Soil/TSM	Ex-situ treatment	Asphalt batching: Treat excavated soil by mixing with asphalt emulsion to create asphaltic roadbase material. Process may be conducted on- or off-site, and asphalt product may be re-used on-site or off-site.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: May be appropriate treatment method for some impacted soil considered feasible for MGP waste, as it is not compatible with hazardous wastes or soil contain selected TSM samples have failed TCLP due to elevated benzene concentrations.
Soil/TSM	Ex-situ treatment	Landfarming: Excavated material is spread on the ground surface and is periodically tilled and aerated to promote biological removal of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not feasible due to urban setting of the site.
Soil/TSM	Ex-situ treatment	Bioslurry Reactors: Excavated soil material is slurried with water and nutrients and is placed in a stirred, aerated reactor to promote biodegradation of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: High concentrations of contaminants limit the feasibility of th
Soil/TSM	Ex-situ treatment	MGP-REM Process: Chemically treat recalcitrant organics with Fenton's reagent (ferrous iron and hydrogen peroxide) to chemically oxidize and biodegrade organic contaminants. May be conducted as either a landfarming or bioreactor process.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This treatment method is not feasible for Area 2 due to the re the potential for emissions that may impact the public.
Soil/TSM	In-situ treatment	In-situ thermal desorption: Application of heat and vacuum to in-situ soil to volatilize, destroy and extract organic contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible due to the presence of subsurfa Additionally, the application of this technology in the presence of DNAPL is unreliable.
Soil/TSM	In-situ treatment	In-situ stabilization: Mix soil in-situ with stabilizing agents using large- diameter hollow-stem auger drilling rigs	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible on the scale required at the Site imit the feasibility of this technology.

TABLE XIV INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 2: SOUTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN

2 also Eliminate	
	nd subsurface obstructions on Area
Site. Eliminate	utilities and the urban setting of the
nt and Eliminate	vely small area available for treatme
Eliminate	echnology.
Eliminate	
ot ilyses on Eliminate	Idwever, this process is generally no free product or cyanide. TCLP ana
able on Eliminate	be treated and the limited space avail
ugh Eliminate	nlikely to be removed from soil throu
Initial Screenin Outcome	ution,

.

(2)

TABLE XIV INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 2: SOUTHERN PORTION OF 100 COMMERCIAL STREET PROPERTY PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS

H	20	2	8	2	
arget Media	oil/TSM	oil/TSM	oil/TSM	oil/TSM	oil/TSM
Technology Type	In-situ treatment	In-situ treatment	Restrictions	Restrictions	Disposal
Process Option Description	Bioventing (unsaturated zone softs): Inject air into vadose zone soils to stimulate aerobic degradation of organic contaminants	In-situ chemical oxidation: Injection of hydrogen peroxide mixture into in-situ soil using geoprobe equipment or wells to chemically break down organic contaminants to innocuous compounds.	Activity and Use Limitations (AULs): Place restrictions on future property use to prevent exposure through pathways for which a condition of No Significant Risk could not be established.	Fencing: Limit site access using fencing	Off-site Disposal: Dispose non-hazardous soils off-site in a landfill
Feasibility of Achieving a Permanent or Temporary Availability of Expertise, and Other Comme	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible for high concentrations of P	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Feasible remedial technology for MGP-impacted soil. In-sipublic location of Area 2. Injection of chemical oxidant into the soil beneath asphalt parking and vapors - ventilation may be required.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to in plement the technology? Yes. Comments/Concerns/Explanation: Would likely be implemented to prevent potential future ex	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Although Area 2 is located in a relatively public commercia property is currently limited with a fence surrounding a portion of the property. Additional re feasible.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This option is considered less favorable than thermal treatm

Haley & Aldrich, Inc. G:\06558\634 Phase III\Ph\_III\_Initial\_Screen\_Tables.xls

Solution,	Initial Screening Outcome
AH contamination.	Eliminate
tu treatment method may be preferable due t treas and buildings may generate excessive h	the Retain at
posures to contaminated soils.	Retain
l area, public access to the 100 Commercial strictions to public access are not reasonably	treet Eliminate
ent options, which treat contaminated soil pr essary to achieve a Temporary or Permanent	or to Eliminate
	6/27/2003

and deep DNAPL monitoring - AUL	Alternative 2-3:	Alternative 2-2: - Shallow and deep DNAPL recovery using extraction wells - LNAPL extraction using belt-skimmers - Provisional shallow DNAPL migration control - AUL	oxidation of accessible TSM and LNAPL- impacted soils - Provisional shallow DNAPL migration control - AUL - AUL	Alternative 2-1: - Shallow and deep DNAPL recovery using extraction wells - LNAPL extraction using Multi-phase extraction (MPE) - In-situ chemical	REMEDIAL ACTION ALTERNATIVE	TABLE XVI DETAILED EVALUAT AREA 2 - SOUTHERN PHASE III REMEDIAL FORMER MALDEN M MALDEN, MASSACH
located in the subsurface on Area 2 does not migrate off-site. An AUL involving the maintenance of a direct contact barrier (i.e., pavement) may be used to restrict future exposure to contaminated soils.	This alternative involves continued monitoring	Install 13 shallow and deep DNAPL recovery wells, and 6 wells equipped with belt- skimmers to extract LNAPL. Construct NAPL storage shed to store extracted LNAPL and DNAPL. A vertical subsurface barrier to shallow DNAPL flow (e.g., sheet piles, shallow DNAPL flow (e.g., sheet piles, waterloo Barrier) may be installed on Area 2 if it is necessary to prevent DNAPL flow from adjacent areas onto Area 2. An AUL involving the maintenance of a direct contact barrier (i.e., pavement) may be used to restric future exposure to contaminated soils.	practical, install in-situ chemical oxidation system to remove residual contaminants from soil. A vertical subsurface barrier to shallow DNAPL flow (e.g., sheet piling, Waterloo Barrier) may be installed on Area 2, if it is necessary to prevent DNAPL flow from adjacent areas onto Area 2. An AUL involving the maintenance of a direct contact barrier (i.e., pavement) may be used to restric future exposure to contaminated soils.	Install shallow and deep DNAPL recovery wells, and MPE wells to extract LNAPL. Install a groundwater treatment facility on site to separate LNAPL from groundwater and treat extracted groundwater. Construct NAPI storage shed to store extracted LNAPL and DNAPL. Once both LNAPL and DNAPL thicknesses have been reduced to the extent	DESCRIPTION OF REMEDIAL ACTION ALTERNATIVE	TION OF REMEDIAL ACTION ALTERNAT PORTION OF PARCEL E . ACTION PLAN GP SITE USETTS
Solution, and does not extract DNAPL or LNAPL from the subsurface. An AUL is a relatively effective method for restricting future exposure to contaminated soils. This alternative meets the requirements of a Temporary Solution.	Low.	Moderate. This alternative is unlikely to achieve a Permanent Solution, as DNAPL and TSM would likely remain beneath the 100 Commercial Street building. However, this alternative would achieve a Temporary Solution.	of DNAPL and access limitations. Therefore, because DNAPL and TSM (i.e., soil exceeding UCLs) remains on the Area, a Permanent Solution is unlikely to result in this alternative. However, this alternative would achieve a Temporary Solution.	Moderate. This alternative is unlikely to reach a Permanent Solution, as DNAPL and TSM would remain beneath th 100 Commercial Street building. Cannot inject chemical oxidant beneath 100 Commercial Street building, both due to the presence	EFFECTIVENESS	IVES
	Not Applicable: No remedy is	Short term: Moderate, Belt- skimmer and DNAPL extraction wells are reliable systems, but will require maintenance to continue operation. A DNAPL extraction well has bee functioning reliably on Area 2 since October 2001. Long-term: Moderate, Belt- skimmer and DNAPL extraction wells are reliable in the long term. however will likely continue to operate for the foreseeable future.	Long-term: Moderate, MPE and DNAPL extraction wells are reliable in the long term, however would likely continue to operate for the foresceable future.	Short term: Moderate, MPE and h DNAPL extraction wells are reliable systems in the short term, chowever would require maintenance to continue operation A DNAPL extraction well has bee functioning reliably on Area 2 since October 2001	RELIABILITY (SHORT-TERM/LONG-TERM)	
program ob the Area would be relatively straightforward, as monitoring wells currently exist on Area 2.	Readily Implementable.	Readily Implementable. Installation of DNAPL and LNAPL belt-skimmer wells is relatively straightforward. This alternative requires construction of a NAPL storage shed on Area 2. Access to soils beneath the 100 Commercial Street building for DNAPL removal is limited.	facility, and a NAPL storage shed be located on Area 2, which may interfere with the owner's use of the Area. Soil beneath the 100 Commercial Street building in inaccessible for DNAPL removal or for chemical oxidation treatment.	Moderately Isplementable. Installation of DNAPL extraction and LNAPL MPE system is relatively straightforward, and chemical oxidation system can be designed and implemented. This alternative requires that an MPE system, a groundwater treatment	IMPLEMENTABILITY	21 B
	NPV: \$1.3M NPV: \$110K	Selected Components: DNAPL Recovery System Installation: \$200K LNAPL Recovery System Installation: \$130K Shallow DNAPL Shallow DNAPL migration control: \$110K Engineering: \$180K O & M: \$280K	TSM and LNAPL- impacted soil: \$560K Shallow DNAPL migration control: \$110K Engineering: \$270K O & M: \$280K NPV: \$2.4M	Selected Components: DNAPL Recovery System Installation: \$200K MPE System Installation: \$270K Chemical Oxidation of	COST (NPV)	
exposure to hazardous materials by workers in this alternative.	Low.	Low. Risks associated with this alternative include risks to workers handling extracted NAPL.	injection of chemical oxidant beneath pavement or structures, (such as the building or WEB Culvert) due to the generation of heat and gasses during injection.	Moderate. Risks associated with implementation of this alternative include risks to workers handling extracted NAPL and chemical oxidants. Some risk associated with	RISKS	
impact on property use Little potential for exposure to contaminants under current/foresetable future property uses	This alternative	This alternative involves significantly less impact on property use than Alternative 2 1, as it does not requir an MPE system or a groundwater treatment system.		Removal of LNAPL and DNAPL, and remediation of contaminated soil (ove time) in accessible areas	BENEFITS	
. alternative does not extract LNAPL or DNAPL from the subsurface: therefore monitoring will continue under a Temporary Solution until a change in Area conditions allows remedial measures that result in a Permanent Solution.	Requires Long-term	Requires Long-term Implementation. y Extraction of estimated - quanities of LNAPL and on Area 2 may require an t extended period of time (i.e., 4 to 6 years and 7 to 10 years, respectively) in this alternative.	DNAPL from the subsurface on Area 2 is likely to require an extended period of time (i.e., 7 to 10 years) to accomplish. LNAPL extraction using MPE is likely to achieve more rapid LNAPL extraction rates than other methods of LNAPL removal.	Requires Long-term Implementation. Extraction of estimated quantities of LNAPL using MPE would likely be relatively timely, requiring approximately 2 years. However - extraction of	TIMELINESS	
little impact on property use.	Low.	Low. This alternative involves little impact on property use, other than construction of a small NAPL storage shed on the Area.	the nature of the current use of the Area, it is not anticipated that the presence of these facilities will have a significan effect on these interests.	Moderate. The presence of a NAPL storage shed, MPE system, an groundwater treatment system will occupy space on the property and will affect the current use of the Area somewhat However based or	EFFECT ON NON- PECUNIARY INTERESTS	r egy

-

2

1

Haley & Aldrich, Inc. G;/06558/634 Phase III/Ph\_III\_Det\_Eval\_Tables

5	
ω	
2	
۳.	
-	
ឮ	
_	

.

•

G:/06558/634 F	Haley & Aldrict
phase	1, Inc.
III/Ph_III	
Initial	
Screen_Tab	
les.xls	

Target Medi	Technology a Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary Solution Availability of Expertise, and Other Comments
DNAPL	Extraction	DNAPL Extraction using Recovery Wells: Installation of large-diameter DNAPL extraction wells, equipped with a submersible pump to extract DNAPL from the subsurface. Extracted DNAPL stored in small buildings on the property and transported off-site for disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Technology is applicable to both shallow and deep DNAPL. Reas existing DNAPL extraction well that has been operating successfully on the southern portion of Parce
DNAPL	Extraction/ . Migration control	DNAPL Extraction/Migration Control using Trenches: Excavation of trenches that are keyed into the low- permeability organic deposit, and backfilled with gravel for collection of DNAPL. One or more recovery wells equipped with a submersible pump for extraction of DNAPL would be installed in the trench.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be significantly more difficult than extraction wells to obstructions, odors, space for excavation, and disposal of excavated soil material.
DNAPL	Migration control	Vertical Subsurface Barrier: Install barrier to DNAPL flow. May be slurry wall, sheet piling, or Waterloo <sup>TM</sup> barrier, dependent upon Phase IV design. Would also involve injection of grout through the base of the WEB Culvert to prevent DNAPL migration beneath the culvert.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be required in the event that DNAPL is removed from DNAPL migration from an adjacent area.
DNAPL	Disposal/ Incineration	Disposal or Incineration: Off-site transport of extracted NAPL for incineration or disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be required in the event that DNAPL or LNAPL are remov
DNAPL	Monitoring	DNAPL Monitoring: Conduct semi-annual monitoring of DNAPL thickness in monitoring wells using oil/water interface probes. Place monitoring wells strategically to detec off-site migration.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: NAPL monitoring may be an applicable component for a Tempora NAPL extraction is not feasible.
Soil/TSM	No Action	Natural Attenuation: Allow natural processes such as biodegradation and volatilization to reduce concentrations or toxicity of contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not applicable for remediation of soil containing
Soil/TSM	Physical remova	Excavation and off-site treatment: Excavate contaminated soil and DNAPL/LNAPL, where present, and transport off- site for treatment. Would require support of excavations and building foundations during excavation, and may require demolition of buildings on the property.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Excavation of TSM would likely be difficult due to the generation buildings, disposal of excavated soil, etc.
Soil/TSM	Capping	Installation of Engineered Barrier: RCRA-type engineered barrier consistent with 310 CMR 40. Excavate 3 feet of soil, install barrier, backfill soil and re-vegetate on top of barrier.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Installation of an engineered barrier would eliminate potential for e However, this technology is unlikely to attain a Permanent Solution due to the presence of buildings of

TABLE XVII INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 3: NORTHERN PORTION OF PARCEL A PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE

Initial Screer Outcome Coutcome coully i s for Retain Retain Retain Retain Retain	to contaminated soil. ties in the Area.	support of excavatio	H concentrations or	manent Solution, esp	the subsurface.	and potential remain	due to subsurface	kely to be feasible ba a 2).		
	Retain	ns and Retain	TSM. Eliminate	ecially if Eliminate	Retain	s for Retain	Retain	sed on an Retain	Initial Screen Outcome	

G:/06558/634 Phase IIVPn\_III\_Initia I dole i \_ lias 3

TABLE XVII INITIAL SCRE AREA 3: NOR PHASE III REM FORMER MAL MALDEN, MA	ENING OF REME THERN PORTION AEDIAL ACTION DEN MGP SITE SSACHUSETTS	EDIAL TECHNOLOGIES I OF PARCEL A PLAN			2
Target Media	Technology	Process Option Description	Feasibility of Achieving a Permanent or Temporary Solution, Availability of Expertise, and Other Comments		Initial Screening
Soil/TSM	Capping	Installation/maintenance of a direct contact barrier: Prevent potential exposure to shallow soils by maintaining 3 feet of clean soil or asphalt cover.	s the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Maintenance of a direct contact barrier would likely be required as part of a xposure to contaminated soils.	n AUL to restrict potential	Retain
Soil/TSM	Ex-situ treatment	Incineration: Treat excavated material off-site using incineration	s the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be appropriate for treatment of extracted DNAPL or soils that are no esorption treatment.	t suitable for thermal	Retain
Soil/TSM	Ex-situ treatment	Thermal desorption: Treat excavated soil material using thermal desorption with an afterburner. May be performed on site using a mobile thermal desorption unit, or material may be sent off-site for treatment.	s the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: On-site thermal desorption unlikely to be reasonably feasible due to limited ir emissions. Off-site thermal desorption is reasonably feasible.	available space, odors, and	Retain
Soil/TSM	Ex-situ treatment	Soil washing: Separate silt and clay fraction from sand fraction, and desorb contaminants into the aqueous phase.	s the technology reasonably likely to achieve a Permanent or Temporary Solution? No. tre experts available to implement the technology? Yes. comments/Concerns/Explanation: This technology is unlikely to be effective at this site; TSM is unlikely to be rashing.	removed from soil through	Eliminate
Soii/TSM	Ex-situ treatment	Stabilization: Reduce leachability of organic contaminants through physical encapsulation and/or cement stabilization. Treated soil may be re-used as backfill on the site or sent to appropriate receiving facilities for disposal.	s the technology reasonably likely to achieve a Permanent or Temporary Solution? No. tre experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not reasonably feasible due to the anticipated volume of soil to be treated an n the Area.	d the limited space available	Eliminate
Soii/TSM	Ex-situ treatment	Asphalt batching: Treat excavated soil by mixing with asphalt emulsion to create asphaltic roadbase material. Process may be conducted on- or off-site, and asphalt product ( may be re-used on-site or off-site.	the technology reasonably likely to achieve a Permanent or Temporary Solution? No. The experts available to implement the technology? Yes. Comments/Concerns/Explanation: May be appropriate treatment method for some impacted soil. However, this considered feasible for MGP waste, as it is not compatible with hazardous wastes or soil containing free productions analyses on selected TSM samples have failed TCLP due to elevated benzene concentrations.	s process is generally not t or cyanide. TCLP	Eliminate
Soil/TSM	Ex-situ treatment	Landfarming: Excavated material is spread on the ground surface and is periodically tilled and aerated to promote biological removal of organic contaminants.	the technology reasonably likely to achieve a Permanent or Temporary Solution? No. re experts available to implement the technology? Yes. omments/Concerns/Explanation: Not feasible due to urban setting of the site.		Eliminate
Soil/TSM	Ex-situ treatment	Bioslurry Reactors: Excavated soil material is slurried with 1 water and nutrients and is placed in a stirred, aerated reactor 4 to promote biodegradation of organic contaminants.	the technology reasonably likely to achieve a Permanent or Temporary Solution? No. re experts available to implement the technology? Yes. omments/Concerns/Explanation: High concentrations of contaminants limit the feasibility of this technology.	222.972	Eliminate
Soil/TSM	Ex-situ treatment	MGP-REM Process: Chemically treat recalcitrant organics 1 with Fenton's reagent (ferrous iron and hydrogen peroxide) to / chemically oxidize and biodegrade organic contaminants. May be conducted as either a landfarming or bioreactor process.	the technology reasonably likely to achieve a Permanent or Temporary Solution? No. re experts available to implement the technology? Yes. omments/Concerns/Explanation: This treatment method is not feasible for Area 3 due to the relatively small a nd the potential for emissions that may impact the public.	rca available for treatment	Eliminate
Haley & Aldrich,	Inc.	tial Screen Tables vis			69

1

# 6/27/2003

ю. <sup>34</sup>

 $\hat{\mathbf{x}}_{i}$ 

age 2 of 3

Haley & Aldrich, Inc. G:/06558/634 Phase III/Ph\_III\_Initial\_Screen\_Tables.xls

INITIAL SCREE AREA 3: NORT PHASE III REM FORMER MALI MALDEN, MAS	ENING OF REME HERN PORTION EDIAL ACTION DEN MGP SITE SSACHUSETTS	DIAL TECHNOLOGIES OF PARCEL A PLAN	
Target Media	Technology Type	Process Option Description	Feasibility of Achieving a Permanent or Temporary Solution, Availability of Expertise, and Other Comments
Soil/TSM	In-situ treatment	In-situ thermal desorption: Application of heat and vacuum to in-situ soil to volatilize, destroy and extract organic contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible due to the presence of subsurface utili Additionally, the application of this technology in the presence of DNAPL is unreliable.
Soil/TSM	In-situ treatment	In-situ stabilization: Mix soil in-situ with stabilizing agents using large-diameter hollow-stem auger drilling rigs	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible on the scale required at the Site, and s also limit the feasibility of this technology.
Soil/TSM	In-situ treatment	Bioventing (unsaturated zone soils): Inject air into vadose zone soils to stimulate aerobic degradation of organic contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible for high concentrations of PAH contar
Soil/TSM	In-situ treatment	In-situ chemical oxidation: Injection of hydrogen peroxide mixture or other oxidant into in-situ soil using geoprobe equipment or wells to chemically break down organic contaminants to innocuous compounds.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Feasible remedial technology for MGP and petroleum-impacted soil. preferable due to the public location of Area 3. Injection of chemical oxidant into the soil beneath aspha generate excessive heat and vapors - ventilation may be required.
Soil/TSM	Restrictions	Activity and Use Limitations (AULs): Place restrictions on future property use to prevent exposure through pathways for which a condition of No Significant Risk could not be established.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be implemented to prevent potential future exposures to
Soil/TSM	Restrictions	Fencing: Limit site access using fencing	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? No. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Area 3 contains several commercial businesses that must remain acces access to these properties cannot be restricted.
Soil/TSM	Disposal	Off-site Disposal: Dispose non-hazardous soils off-site in a landfill	Is the technology reasonably likely to achieve a Permanent or Temporary Solution? Yes. Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This option is considered less favorable than thermal treatment-option prior to disposal. Because off-site thermal desorption treatment is retained, off-site disposal is not necess Permanent Solution.

ge 3 of 3

÷

	Initial Screenin Outcome
ilities and the urban setting of the Site.	Eliminate
1 subsurface obstructions on Area 3	Eliminate
ramination.	Eliminate
<ol> <li>In-situ treatment method may be halt parking areas and buildings may</li> </ol>	Retain
to contaminated soil and indoor air.	Retain
cessible to the public; therefore public	Eliminate
ons, which treat contaminated soil -	Eliminate

TABLE XIX DETAILED EVALUAT AREA 3 - NORTHERN PHASE III REMEDIAL FORMER MALDEN N MALDEN, MASSACH	TION OF REMEDIAL ACTION ALT I PORTION OF PARCEL A . ACTION PLAN IGP SITE USETTS	ERNATIVES		,		ы;	
REMEDIAL ACTION ALTERNATIVE	DESCRIPTION OF REMEDIAL ACTION ALTERNATIVE	EFFECTIVENESS	RELIABILITY (SHORT-TERM/LONG-TERM)	IMPLEMENTABILITY	COST (NPV)	RISKS	
Alternative 3-1: - Shallow DNAPL recovery wells	This alternative involves the installation of shallow DNAPL extraction wells on properties located	Moderate. Implementation of this alternative is likely to meet remedial goals with respect to	Short term: Moderate, DNAPL extraction wells are reasonably reliable; one DNAPL extraction	Moderately Implementable. Installation of a DNAPL extraction system will impact	Selected Components: DNAPL recovery system installation: \$180K	Moderate. Relatively little risk associated with the	Remove both DN contami
- Provisional Shallow DNAPL migration control	on Area 3. A DNAPL storage sneu would be constructed on a centrally located property to store extracted	DNAPL removal and reduction of contaminant concentrations in soil on portions of the Area without buildings.	well on Area 2 nas runctioned reliably since October 2001. DNAPL extraction system will	property use; current retail uses of properties within Area 3 involve high (egree of public	Chemical oxidation of TSM soil: \$550K Shallow DNAPL	wells, although measures would be required to ensure	term in portions
- In-situ chemical	DNAPL. When appropriate (i.e.,	DNAPL recovery has been demonstrated	require maintenance to continue to	access. Injection of chemical	Migration Control:	that pubic access to DNAPL	however
TSM - impacted soil	TSM soil is accessible and DNAPL has been reduced to less than 1/2	to be effective at one location on Area 2, and chemical oxidation has been used	operate; unlikely to accomplish remedial goals in short term.	conducted beneath the	SZ20K Engineering: \$190K	storage facility is prevented. Some risk exists to workers	remain b
- AUL	inch), chemical oxidation of TSM soils will be conducted. A vertical	effectively at other MGP sites to reduce concentrations of VOCs and PAHs in soil.	Long-term: Moderate, Systems	buildings located on Area 3, due to inaccessibility of the	0 & M: \$140K	during injection of oxidants for chemical oxidation. Some	building
	DNAPL flow (e.g., sheet piling, Waterloo Barrier) may be installed	soils beneath the buildings on Area 3 is limited, and DNAPL and TSM would	maintenance, however is likely to continue to function reliably over	adverse affects of chemical oxidation reactions beneath	IN A STORE	culvert, building foundations, and subsurface utilities due to	
	on Area 3, if necessary. An AUL involving the maintenance of a direct	likely remain beneath the buildings. Therefore, a Permanent Solution is unlikely to result from implementation of	the long term.	buildings.		heat and vapor generation during chemical oxidation.	
	be used to restrict future exposure to contaminated soils.	this Alternative.			-		
Alternative 3-2: - Shallow DNAPL	This alternative is similar to Alternative 3-1, however it does not	Moderate. Similar to Alternative 3-1, this alternative	Short term: Moderate, DNAPL extraction wells are reasonably	Readily Implementable. This alternative requires some	Selected Components: DNAPL recovery system	Low. Relatively little risk	Requires
- Provisional shallow	soils, based on the assumption that	is reasonably likely to meet DNAPL remedial goals on portions of Area 3	reliable; one DNAPL extraction well on Area 2 has functioned	operations of businesses on	Installation: \$180K Shallow DNAPL	associated with the installation of extraction	operation
DNAPL migration	DNAPL extraction goals are not met	without buildings. However, access to	reliably since October 2001.	Area 3. Construction of a	Migration Control:	wells, although measures	DNAPL
- AUL	chemical oxidation is not feasible due	limited, and DNAPL is likely to remain	require maintenance.	shared use among the	Findineering: \$120K	that pubic access to DNAPL	Removes
	to the presence of the buildings on Area 3. If necessary, vertical	beneath buildings. This alternative would not result in a Permanent Solution because	Long-term: Moderate, Systems	properties in Area 3) is likely to be implementable on one of	O & M: \$140K	storage facility is prevented.	ver the
	barriers to shallow DNAPL flow may be installed on Area 3 to prevent	DNAPL would likely remain beneath buildings and soil containing	maintenauce, however is likely to	the Area 3 properties.	NPV: \$870K		trea 3; h
	migration of shallow DNAPL onto	concentrations of contaminants greater	continue to function reliably over				esidual u
	maintenance of a direct contact	Would meet the requirements of a	The sould not set up.				uildings
	barrier (i.e., pavement) may be used	Temporary Solution.					
Alternative 3-3:	This is the least intrusive of the	Low.	Not Applicable; a remedy is not	Readily Implementable.	NPV: \$110K	Low.	lequires
- DNAPL monitoring	alternatives for Area 3, as it only	Would not result in reduction of NAPL	installed in this Alternative.	This alternative requires the		Few risks associated with this it	nterrupti
- AUL	involves continued monitoring of	thickness or significant reductions in		least intrusion into facility		alternative.	peration
	conditions to ensure that off-site	result in a Permanent Solution because		operations to imprement.			
	An AUL involving the maintenance	DNAPL and TSM would remain in the					
	of a direct contact barrier (i.e.,	subsurface and soil containing					
	future exposure to contaminated	than the UCLs would remain. Would					
	soils.	meet the requirements of a Temporary					
		Solution					
Notes: 1. Cost estimates includ 2. Cost estimates representation of the statement of the	te contractor mobilizations, contingenci sent an opinion of probable cost for con	es and other additional components in addit nparison purposes.	ion to selected components listed; Re	fer to Cost Estimate Tables in A	ppendix D for detailed brea	akdown of cost components.	

\_\_\_\_\_

Haley & Aldrich, Inc. G./06558/634 Phase III/Ph\_III\_Det\_Eval\_Tables

ī .

D.	
8	
8	
0	
-	

SENEF	TS	TIMELINESS	EFFECT ON NON- PECUNIARY INTERESTS
es UCL inated so over th accessit s of Are r, residu ination v beneath beneath	i due to nd sil e long ole a 3; a 1 vould	Requires Long-term Implementation. Accomplishment of remedial goals with respect to DNAPL thickness is likely to require approximately 3 to 5 years using extraction wells in this Area.	Moderate. Would impact current property uses to install and maintain, and requires a DNAPL storage shed for storage of extracted DNAPL Operation of DNAPL extraction and in-situ chemical oxidation system may be undesirable on some properties due to negative perceptions associated with contamination on the property.
s little tion of t subsure subsure subsure subsure subsure to DN. s UCLs s UCLs s UCLs s c of DN. s or the to portion the port	actility moval of material face. due to due to due to APL rn in ms of r, r, r, nns of nns of nns of	Requires Long-term Implementation. Accomplishment of remedial goals with respect to DNAPL thickness is likely to require approximately 3 to 5 years using extraction wells in this Area.	Moderate. Would require some intrusion to current property uses to install and maintain, and requires a DNAPL storage shed for storage of extracted DNAPL. Operation of DNAPL extraction system may be undesirable on some properties due to negative perceptions associated with contamination on the property.
s little	acility	Requires Long-term Implementation. Monitoring would continue indefinitely, until installation of a Permanent Solution is feasible.	Low. This alternative results in little disruption to facility operations. Minimal impacts include access to the properties on the Area for monitoring of DNAPL conditions.

Target	Technology	Process Option Description	Feasibility of Achieving a Permanent or Tempora Availability of Expertise, and Other Comm
LNAPL	Extraction	Multi-phase extraction: Extraction of groundwater, soil vapor and LNAPL through the application of a vacuum to specially designed extraction wells. Requires separation of LNAPL from groundwater and ex-situ treatment of both soil vapor and groundwater.	Is the technology reasonably likely to achieve a Permanent or Temporary Soluti Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Appropriate for LNAPL removal, and remov BTEX) from groundwater.
LNAPL	Extraction	Belt-skimmer: Extraction of LNAPL using belt-skimmers, consisting of two pulleys that drive a hydrophobic belt through the water table in a monitoring well, bring LNAPL oil to the surface, and skim the oil into a collection container. System is driven by an electric motor.	Is the technology reasonably likely to achieve a Permanent or Temporary Soluti Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Appropriate for LNAPL removal in areas of LNAPL is consistent with remedial goals
LNAPL	Extraction	Passive LNAPL extraction methods: LNAPL extraction methods that rely on siphons, absorbent pads, or other passive methods for LNAPL collection.	Is the technology reasonably likely to achieve a Permanent or Temporary Soluti Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Passive methods of LNAPL removal (absorb been tried on the Site in the past, however rates of LNAPL extraction have not l
LNAPL	Monitoring	LNAPL Monitoring: Conduct semi-annual monitoring of LNAPL thickness in monitoring wells using oil/water interface probes. Place monitoring wells strategically to detect off-site migration.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: LNAPL monitoring may be feasible componer Solution.
LNAPL	Disposal/ Incineration	Disposal or Incineration: Off-site transport of extracted NAPL for incineration or disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be required in the event that DNAPL subsurface.
Soil/TSM	No Action	Natural Attenuation: Allow natural processes such as biodegradation and volatilization to reduce concentrations or toxicity of contaminants	is the technology reasonably likely to achieve a Permanent or Temporary Solutio Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not applicable for remediations of PAH compounds.
Soil/TSM	Physical removal	Excavation and off-site treatment: Excavate contaminated soil and DNAPL/LNAPL, where present, and transport off-site for treatment. Would require support of excavations and building foundations during excavation, and may require demolition of buildings.	Is the technology reasonably likely to achieve a Permanent or Temporary Solutic Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Excavation of contaminated soil and TSM wo generation of odors, support of excavations and buildings, disposal of excavated
Soil/TSM	Capping	Installation of Engineered Barrier: RCRA-type engineered barrier consistent with 310 CMR 40. Excavate 3 feet of soil, install barrier, backfill soil and re-vegetate on top of barrier.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes, Comments/Concerns/Explanation: Installation of an engineered barrier would elicontaminated soil. However, this technology would not attain a Permanent Solution Area 4.

TABLE XX INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 4: SOUTHERN PORTION OF PARCEL A PHASE III REMEDIAL ACTION PLAN

Page 1 of 3

y Solution,	Initial Screenin
m? Yes. al of volatile compounds (such as	Retain
n? Yes. high public access, on which gradual	Retain
n? No. nt pads, siphons without a pump) have een consistent with remedial goals.	Eliminate
n? Yes. at for a Temporary or Permanent	Retain
n? Yes. r LNAPL are removed from the	Retain
n? No.	Eliminate
n? Yes. and likely be difficult due to the soil, ejc.	Retain
n? Yes.	Retain

1

1

Haley & Aldrich, Inc. G:/06558/634 Phase III/Ph\_III\_Initial\_Screen\_Tables.xls

Target	Technology Type	Process Option Description	Feasibility of Achieving a Permanent or Tempo Availability of Expertise, and Other Co
Soil/TSM	Capping	Installation/maintenance of a direct contact barrier: Prevent potential exposure to shallow soils by maintaining 3 feet of clean soil or asphalt cover.	Is the lechnology reasonably likely to achieve a Permanent or Temporary So Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Maintenance of a direct contact barrier w to restrict potential exposure to contaminated soils.
Soil/TSM	Ex-situ treatment	Incineration: Treat excavated material off-site using incineration	Is the technology reasonably likely to achieve a Permanent or Temporary So Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would be appropriate for treatment of exi suitable for thermal desorption treatment.
Soil/TSM	Ex-situ treatment	Thermal desorption: Treat excavated soil material using thermal desorption with an afterburner. May be performed on-site using a mobile thermal desorption unit, or material may be sent off-site for treatment.	Is the technology reasonably likely to achieve a Permanent or Temporary Sc Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: On-site thermal desorption unlikely to be space, odors, and air emissions. Off-site thermal desorption is reasonably for
Soil/TSM	Ex-situ treatment	Soil washing: Separate silt and clay fraction from sand fraction, and desorb contaminants into the aqueous phase.	Is the technology reasonably likely to achieve a Permanent or Temporary Sc Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is unlikely to be effectiv unlikely to be removed from soil through washing.
Soil/TSM	Ex-situ treatment	Stabilization: Reduce leachability of organic contaminants through physical encapsulation and/or cement stabilization. Treated soil may be re-used as backfill on the site or sent to appropriate receiving facilities for disposal.	Is the technology reasonably likely to achieve a Permanent or Temporary Sc Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not reasonably feasible due to the anticip limited space available on the Area.
Soil/TSM	Ex-situ treatment	Asphalt batching: Treat excavated soil by mixing with asphalt emulsion to create asphaltic roadbase material. Process may be conducted on- or off-site, and asphalt product may be re-used on-site or off-site.	Is the technology reasonably likely to achieve a Fermanent or Temporary So Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: May be appropriate treatment method for process is generally not considered feasible for MGP waste, as it is not comp containing free product or cyanide. TCLP analyses on selected TSM sample benzene concentrations.
Soil/TSM	Ex-situ treatment	Landfarming: Excavated material is spread on the ground surface and is periodically tilled and aerated to promote biological removal of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary So Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Not feasible due to urban setting of the si
Soil/TSM	Ex-situ treatment	Bioslurry Reactors: Excavated soil material is slurried with water and nutrients and is placed in a stirred, aerated reactor to promote biodegradation of organic contaminants.	Is the technology reasonably likely to achieve a Permanent or Temporary So Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: High concentrations of contaminants limit

TABLE XX INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 4: SOUTHERN PORTION OF PARCEL A PHASE III REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE

6/27/2003

1

olution,	Initial Screening Outcome
Yes.	
tely be required as part of an AUL	Retain
Yes	Refain
bly feasible due to limited available	Retain
No.	
site; MGP oils and TSM are	Eliminate
No.	
ume of soil to be treated and the	Eliminate
No	
npacted soil. However, this vith hazardous wastes or soil	Eliminate
failed TCLP due to elevated	
No	Eliminate
No	
	Eliminate

11 -

Page 2 of 3

Hatey & Aldrich, Inc. G:/06558/634 Phase III/Ph\_III\_Initial\_Screen\_Tables.xls

Target	Technology	Process Option Description	Feasibility of Achieving a Permanent or Tempora Availability of Expertise, and Other Comm
Soil/TSM	Ex-situ treatment	MGP-REM Process: Chemically treat recalcitrant organics with Fenton's reagent (ferrous iron and hydrogen peroxide) to chemically oxidize and biodegrade organic contaminants. May be conducted as either a landfarming or bioreactor process.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This treatment method is not feasible for Are available for treatment and the potential for emissions that may impact the public for the solution.
Soil/TSM	In-situ treatment	In-situ thermal desorption: Application of heat and vacuum to in-situ soil to volatilize, destroy and extract organic contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible due to the presurban setting of the Site.
Soil/TSM	In-situ treatment	In-situ stabilization: Mix soil in-situ with stabilizing agents using large-diameter hollow- stem auger drilling rigs	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible on the scale repostructions on Area 4 also limit the feasibility of this technology.
Soil/TSM	In-situ treatment	Bioventing (unsaturated zone soils): Inject air into vadose zone soils to stimulate aerobic degradation of organic contaminants	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This technology is not feasible for high concerned to the second
Soil/TSM	In-situ treatment	<i>In-situ</i> chemical oxidation: Injection of hydrogen peroxide mixture or other oxidizing agent into in-situ soil using geoprobe equipment or wells to chemically break down organic contaminants to innocuous compounds.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Feasible remedial technology for MGP and put reatment method may be preferable due to the public location of Area 4. Inject beneath asplialt parking areas and buildings may generate excessive heat and variable beneath asplialt parking areas and buildings may generate excessive heat and variable beneath asplication.
Soil/TSM	Restrictions	Activity and Use Limitations (AULs): Place resultions on future property use to prevent exposure through pathways for which a condition of No Significant Risk could not be established.	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: Would likely be implemented to prevent potensoils.
Soil/TSM	Restrictions	Fencing: Limit site access using fencing	Is the technology reasonably likely to achieve a Permanent or Temporary Solution Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: The properties on Area 4 are currently common public access; therefore access to Area 4 cannot be limited.
Soil/TSM	Disposal	Off-site Dispose non-hazardous soils off-site in a landfill	Is the technology reasonably likely to achieve a Permanent or Temporary Solutio Are experts available to implement the technology? Yes. Comments/Concerns/Explanation: This option is considered less favorable than the contaminated soil prior to disposal. Because off-site thermal desorption treatment necessary to achieve a Temporary or Permanent Solution.

TABLE XX INITIAL SCREENING OF REMEDIAL TECHNOLOGIES AREA 4: SOUTHERN PORTION OF PARCEL A PHASE III REMEDIAL ACTION PLAN

e 3 of 3

Solution,	Initial Screenin Outcome
1? No. 4 due to the relatively small area	Eliminate
1? No. nce of subsurface utilities and the	Eliminate
1? No. uired at the Site, and subsurface	Eliminate
17 No. trations of PAH contamination.	Eliminate
1? Yes. roleum-impacted soil. In-situ or of chemical oxidant into the soil rs - ventilation may be required.	Retain
1? Yes. ial future exposures to contaminated	Retain
17 No.	Eliminate
? Yes. ermal treatment options, which treat is retained, off-site disposal is not	Eliminate
Haley & Aldrich, Inc. G:/06558/634 Phase III/Ph\_III\_Det\_Eval\_Tables

2. Cost estimates represent an opinion of probable cost for comparison purposes.

		achieves a Temporary Solution, and an AUL is a relatively effective means of restricting exposure to contaminated soil.				
Alternative 4-2 - LNAPL extraction using belt-skimmers - AUL	Install LNAPL extraction wells equipped with belt-skimmers to extract LNAPL, and construct NAPL storage shed to store extracted LNAPL. This alternative does not include chemical oxidation of TSM and LNAPL-impacted soils, based on the assumption that LNAPL extraction goals are not met in a reasonable time frame, and that chemical oxidation is not feasible due to the presence of the buildings on Area 4. An AUL involving maintenance of a direct contact barrier (i.e., pavement) may be used to restrict future exposure to contaminated soils.	Moderate. This alternative is unlikely to achieve a Permanent Solution, as LNAPL and TSM will likely remain beneath the buildings on the Area. However, this alternative does reduce the volume of LNAPL on the Area and achieves a Temporary Solution, and an AUL is a relatively effective means of restricting exposure to contaminated soil.	Short term: Moderate, Belt- skimmers are reliable method of LNAPL extraction; would require maintenance to continue operation. Long-term: Moderate, Belt- skimmer extraction system would likely continue to operate for the foreseeable future.	Moderately Implementable. Installation of LNAPL belt-skimmer wells is relatively straightforward. This alternative requires construction of a NAPL storage shed on Area 4, which would require approval of the property owner. Access to soils beneath buildings on the Area for LNAPL removal is limited.	Selected Components: LNAPL extraction system installation: \$160K Engineering: \$90K O & M: \$140K NPV: \$510K	Low. Risks associated with this alternative include risks to workers handling extracted LNAPL.
Alternative 4-3: - LNAPL monitoring - AUL	This alternative involves continued monitoring of subsurface conditions to ensure that LNAPL located in the subsurface on Area 4 does not migrate off-site. An AUL involving maintenance of a direct contact barrier (i.e., pavement) may he used to restrict future exposure to contaminated soils.	Low. Does not achieve a Permanent Solution, and does not extract LNAPL from the subsurface. This alternative meets the requirements of a Temporary Solution, and an AUL is a relatively effective means of restricting exposure to contaminated soil.	Not Applicable: No remedy is installed under this alternative.	Readily Implementable. Implementation of a monitoring program on the Area would be relatively straightforward, as monitoring wells currently exist on Area 4. Would require access from property owner.	NPV: SIOOK	Low. Very low risk for exposure to hazardous materials by workers in this alternative.
Notes: 1. Cost estimates inclu	de contractor mobilizations, contingencies	and other additional components in addit	ion to selected components listed; R	efer to Cost Estimate Tables in Appendi	ix D for detailed breakdow	m of cost components.

PHASE III REMEDIAL ACTION PLAN DETAILED EVALUATION OF REMEDIAL ACTION ALTERNATIVES AREA 4 - SOUTHERN PORTION OF PARCEL A TABLE XXII MALDEN, MASSACHUSETTS FORMER MALDEN MGP SITE

impacted soils

- AUL

residual contaminants from TSM and chemical oxidation system to remove the extent practical, install in-situ

LNAPL - impacted soils. An AUL

(i.e., soil exceeding UCLs) may remain

Therefore, because LNAPL, and TSM of LNAPL and access limitations.

unlikely to result in this alternative. on the Area, a Permanent Solution is

oxidation treatment.

injection.

due to the generation of

neat and vapors during

adjacent to structures, beneath pavement or

for LNAPL removal or for chemical

Soil beneath buildings is inaccessible

LNAPL volume on the Area and However, this alternative does reduce involving maintenance of a direct

ontact barrier (i.e., pavement) may be

contaminated soils.

used to restrict future exposure to

**FSM and LNAPL-**

- In-situ chemical oxidation of accessible

groundwater. Construct NAPL storage shed to store extracted LNAPL. Once

portion of the LNAPL that may be located beneath buildings on the Area.

the buildings, both due to the presence Cannot inject chemical oxidant beneath

system would likely continue to

Long-term: Moderate, MPE

continue operation.

would require maintenance to reliable LNAPL removal. System

chemical oxidation system can be designed and implemented. This

relatively straightforward, and Moderately Implementable. Installation of MPE system is

\$370K

Selected Components:

(NPV) COST

MPE System Installation

**Risks** associated with

Moderate.

RISKS

operate for the foreseeable future.

located on a property on Area 4,

which would require an access agreement with the property owner.

NPV: \$1.3M

of chemical oxidant

associated with injection

facility, and a NAPL storage shed be O & M: \$280K

system, a groundwater treatment alternative requires that an MPE

Engineering: \$210K

LNAPL, Some risk oxidants and extracted

\$220K

LNAPL and TSM soil: Chemical oxidation of

handling chemical

risks to workers this alternative include

LNAPL thickness has been reduced to

groundwater and treat extracted

site to separate LNAPL from

extraction (MPE) using Multi-phase Alternative 4-1:

- LNAPL extraction

and a groundwater treatment facility on

LNAPL extraction using MPE is likely

MPE combines groundwater and

(SHORT-TERM/LONG-TERM) Short term: Moderate. Because

RELIABILITY

IMPLEMENTABILITY

LNAPL extraction, it results in

to be more effective than belt-skimmers in Alternative 4-2, and may remove a

Install MPE system to extract LNAPL,

Moderate.

REMEDIAL ACTION

DESCRIPTION OF REMEDIAL ACTION ALTERNATIVE

EFFECTIVENESS

ALTERNATIVE

Page 1 of 1

This alternative involves very little impact on property use	This alternative involves significantly less intrusion intoproperty use than Alternative 4-1 as it does not require an MPE system or a groundwater treatment system.	Relatively rapid remova of LNAPL frem subsurface, reduction of concentrations in soil in accessible portions of Area 4; however, residual contamination would remain beneath buildings.
Requires Long-term Implementation. This alternative does not extract LNAPL from the subsurface: therefore monitoring will continue under a Temporary Solution until a change in Area conditions allows installation of a Permanent Solution.	Requires Long-term Implementation. Extraction of estimated quantities of LNAPL in the subsurface on Area 4 may require an extended period of time (i.e., 3 to 5 years).	Relatively Timely. Extraction of estimated quantities of LNAPL in the subsurface on Area 4 using MPE technology is likely to require up to approximately 2 years. LNAPL extraction using MPE is likely to achieve more rapid LNAPL extraction rates than other methods of LNAPL removal However, LNAPL may remain in inaccessible portions of the Area (i.e., underneath buildings)
Low. This alternative involves very little impact on property use.	Low. This alternative involves little impact on property use, other than construction of a small LNAPL storage shed on the Area.	PECUNIARY Moderate. The presence of a NAPL storage shed, MPE system, and groundwater treatment system will occupy space on the property and will affect current use of properties on the Area; would require access agreement with the propert, owner.

6/27/2003

Page 1 of 1

SUMMARY OF COSTS FOR THE SELECTED REMEDIAL ALTERNATIVE PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS TABLE XXIV

ALIENNALIVE AKEA	AREA DESCRIPTION	REMEDIAL ALTERNATIVE DESCRIPTION	TO COMPLETE ALTERNATIVE	VAL	PRESENT UE (NPV <sup>(1)</sup> )
Area 1	Northern portion of the 100 Commercial Street property	Alternative 1-4: DNAPL extraction, provisional DNAPL migration control, in-situ chemical oxidation of 'ISM and petroleum-impacted soils, and monitoring of LNAPL and deep DNAPL.	6 to 9 years	s	1,100,000
Area 2	Southern portion of the 100 Commercial Street property	Alternative 2-2: DNAPL extraction, LNAPL extraction using belt-skimmers, and provisional DNAPL migration control.	7 to 10 years	\$	1,300,000
Area 3	Northern portion of Parcel A	Alternative 3-2: Shallow DNAPL extraction and provisional shallow DNAPL migration control.	3.10.5.years	\$	870,000
Area 4	Southern portion of Parcel A	Alternative 4-2: LNAPL extraction using belt-skimmers	3 to 5 years	÷	510,000
Area 5	Parcel B	Brown and Caldwell Alt-3: Biosparging and SVE	6 to 7 years	\$	970,000
Groundwater Monitoring		Implementation of a site-wide annual groundwater monitoring program	30 years	∽.	200,000
SUM OF ESTIMATED N	PV FOR FORMER MAI	DEN MGP SITE <sup>(2)</sup> :		s	4,950,000

Cost estimates represent an opinion of probable cost for comparison purposes.
 Refer to the Focused Feasibility Study prepared by Brown and Caldwell (Appendix E) for Area 5 estimated costs.

Ŀ





£

UNDERGROUND ENGINEERING & ENVIRONMENTAL SOLUTIONS	HALFY	3. P2 MOI		
SCALE: AS SHOWN	PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MOP SITE MALDEN, MASSACHUSETTS PARCEL DESIGNATIONS	SE BASE PLAN SOURCE: TOPOGRAPHIC WORKSHEET ( THE MANUFACTURED CAS PLANT, MALDEN, MA. FR MASSACHUSETTS ELECTRIC COMPANY, WESTBORDO, NH SCALE 1"-40", CONTOUR INTERNAL: 2", PHOTO DA 23 MAY 1995, COMPLATION DATE: 22 JUNE 1996 STIE CONOTIONS ON PARCEL D ARE FROM "AS STIE PLAN OF MILLIAM R. CALLAHAM MEMORIAL PA MALDEN, MASS." BY MEDFORD ENGINEERING & SU 10 OCTOBER 1996. THE EXTENT OF PARCEL BOUNDARES IS APPROXIMATE. BOUNDARIES OF DEP SITES ARE APPROXIMATE.	ND: WAVER COMPLETION STATEMENT OR RAO FILED WITH WADEP RTN 3-0362	Contraction Contraction
UNE 2003				





						6	
FLOURE 3	WINDOWS       PHASE III - REMEDIAL ACTION PLAN         PHASE III - REMEDIAL ACTION PLAN         FORMARIA MAXOEN MSP SITE         MADEN, MASACHUSETIS         DISPOSAL SITE BOUNDARY         SUBLITIONS         SUBLITIONS         SUBLITIONS	2. ELEVATIONS ARE IN FEET AND RELATIVE TO NOVD	NOTES 1. BASE PLAN ADAPTED FROM "TOPOGRAPHIC WORKSHEET OF THE MANUFACTURED FARM TOPOGRAPHIC WORKSHEET FOR MASSACHUSETTS ELECTRIC COMPANY BY EASTERN TOPOGRAPHICS, JUNE 1995, AND CITY OF MALDEN ASSESSOR'S PLAN SHEETS 1 AND 2, JUNE 1995, AND SHEET NO 53 BY FAY, SPOFFORD & THORNDIKE, INC.	A X X X X X X X X X X X X X X X X X	FORMER MALDEN MGP DISPOSAL SITE BOUNDARY PILE-SUPPORTED CONCRETE CULVERTS	LEGEND	



G:/06558/616 Ph II/ArcView/Phase\_III\_Arc\_Map\_Figures/Fig5.mxd\_ARCGIS



ATURES ATURES	UNDERGROUND ECONFERENCIAL ENVIRONS SOLUTIONS SOLUTIONS UNDER SOLUTIONS SOLUTIONS UNDER SOLUTIONS SOLUTIONS UNDER SOLUTIONS SOLUTIONS SOLUTIONS UNDER SOLUTIONS SOLUTIO	200 100 0 200 Feet	AND SHEET NO. 53 BY FAY, SPOFFORD & THORNDIKE, INC., UPDATED JUNE 1976 AND REVISED 30 JULY 1979. 2. HISTORIC RIVER LIMITS AND MGP STRUCTURES ESTIMATED BASED ON HISTORIC SANBORN PLANS, "MALDEN AND MELROSE GAS LIGHT CO., LÄND FLÄN", 1907, AND "MALDEN MASS., PLAN OF WORKS", 1916.	NOTES: 1. BASE PLAN ADAPTED FROM "TOPOGRAPHIC WORKSHEET OF THE MANUFACTURED GAS PLANT, MALDEN, MA" FOR MASSACHUSETTS ELECTRIC COMPANY BY EASTERN TOPOGRAPHICS, JUNE 1995, AND CITY OF MALDEN ASSESSOR'S PLAN SHEETS 1 AND 2, JUNE 1995,	EXISTING PILE-SUPPORTED CONCRETE CULVERTS (INSTALLED CIRCA 1970)	PRESENT-DAY BASEMAP FEATURES	APPROXIMATE LIMITS OF HISTORIC WATER FEATURES, 1907 AND 1916	
--	---	-----------------------	---	--	---	------------------------------	---	--



















•





	1	1	41			
HALEY & PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS TOP OF ORGANIC DEPOS CONTOUR PLAN ENGINERIDANS ENGINERIDANS ENGINERIDANS SCALE AS SHOWN	200 100 0 200	<ul> <li>NOTES:</li> <li>BASE PLAN ADAPTED FROM "TOPOGRAPHIC OF THE MANUFACTURED GAS PLANT, MALD FOR MASSACHUSETTS ELECTRIC COMPAN TOPOGRAPHICS, JUNE 1995, AND CITY OF A ASSESSORY: FLAN CITCTS 1 AND 2, JUNE AND SHEET NO. 53 BY FAY, SPOFFORD &amp; TH UPDATED JUNE 1976 AND REVISED 30 JULY</li> <li>ELEVATIONS ARE IN FEET AND RELATIVE TO</li> </ul>		Ho     GROUND SURFACE CONTOUR (10-9     GROUND SURFACE CONTOURS OF THE TO     ORGANIC DEPOSIT (2-FOOT INTER)     APPROXIMATE LIMITS OF THE ORG     EXISTING CONDITIONS     PILE_SUPPORTED CONCRETE CULV	LEGEND: $\oplus$ ELEVATION OF THE TOP OF THE OF 4.64 BASED ON TEST BORING OBSERVA - & - ELEVATION OF THE TOP OF THE OF 4.64 BASED ON TEST PIT OBSERVATION ORGANIC DEPOSIT NOT DETECTED AT THIS LOCATION	
SIT JUNE 2003	feet	C WORKSHEET DEN, MA" Y BY EASTERN MALDEN 1995, HORNDIKE, INC., 1979.		POOT INTERVAL) OP OF THE VAL) IANIC DEPOSIT	RGANIC DEPOSIT ATIONS RGANIC DEPOSIT	



LEGEND:					
DESIGNAT DRILLED E OBSERVAT	ON AND BY GEOLO TION OF H	APPROX GIC, INC IALEY &	IMATE ., UND : ALDR	LOCAT ER TE ICH, IN	
-	98E-B801	DRILLED	0 19 N 3 JULY	OVEME - 25	BE 5
-	97A-B601	DRILLED	8-26	3 JULI	r
4	B403	DRILLED	) 5-7	APRIL	
	B301L	DRILLED	) 18 J	ANUAF	۲Y
-	B201	DRILLED	27-2	29 JUL	.Y
-	B101	DRILLED	2-7	MAY	19
-	<sup>B1</sup>	DRILLED	) 23-2 BER 19	25 NO 87	VE
DESIGNATI DRILLED E OBSERVAT	ON AND IY GUILD ION OF H	APPROXI DRILLING ALEY &	MATE G CO., ALDRI	LOCAT INC. U ICH, IN	
4	B501	DRILL	ED 25-	-30 J	U
-	97B-701	DRILL	ED 13-	-15 F	EB
-\$	028-8917	DRILL AND	ED 18- 28 JUL	-21 FE	EB
DESIGNATI DRILLED B OBSERVAT	ON AND Y TECHNI ION OF H	APPROXI CAL DRI ALEY &	MATE LLING ALDRI	LOCAT CO., II CH, IN	
-6	GP98-101	DR	ILLED 2	24 AP	RI
-	98B-GP1	DR	ILLED 🤇	3 FEBI	RL
4	99B-GP20	1 DR	ILLED 2	2—4 A	U
ф <sup>NO.1</sup>	DESIGNA BORING I ON 29 S	TION AN DRILLED EPTEMB	D APP BY CA ER-2	ROXIM RR-D OCTOE	A
OW	INDICATE WAS INS	S GROUI TALLED	NDWAT IN THE	ER MC	PL
TP202	DESIGNA MACHINE SONS, IN 2 SEPTEI	TION AN -EXCAV C., EVER MBER 19	D APP ATED RETT, 1 998.	ROXIM TEST F MASSA	
TP102	DESIGNA MACHINE SONS, IN 18 TO 21	TION AN -EXCAV C., EVEI I MARCH	D APP ATED RETT, 1 H 1995	ROXIM TEST F MASSA	
M TP8	DESIGNA MACHINE CONSTRU ON 6 TO	TION AN -EXCAV ICTION ( 14 AU)	D APP ATED CO., IN GUST 1	ROXIM TEST F C., ME 988.	A 7 21 7
SS-101	DESIGNA SURFACE ON 9 JU	NON AN SOIL S NE 2002	D APP AMPLE 2	roxim S by	AT H
	DESIGNAT OF SUBS	ION, AP	PROXIN	ATE	LC

NOTES:

- 1. BASE PLAN ADAPTED FROM "TOPOGRAPHIC WORKSHEET OF THE MANUFACTURED GAS PLANT, MALDEN, MA" FOR MASSACHUSETTS ELECTRIC COMPANY, WESTBOROUGH, MA, BY EASTERN TOPOGRAPHICS, WOLFEBORO, NH, SHEETS 1 AND 2, AT A SCALE OF 1 IN. EQUALS 40 FT., JUNE 1995, AND CITY OF MALDEN ASSESSOR'S PLAN SHEET NO. 53, BY FAY, SPOFFORD & THORNDIKE, INC., BOSTON, MA, AT A SCALE OF 1 IN. EQUALS 40 FT., UPDATED JUNE 1976 AND REVISED 30 JULY 1979.
- 2. LOCATION OF TEST BORINGS AND TEST PITS WERE DETERMINED BY HALEY & ALDRICH, INC.

e e	1,2411	i seetses i ii	5 • 500000 5 5 *		HALEY & ALDRICH	PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS
						SUBSURFACE EXPLORAT
0	60 S	120 CALE IN FE	180 ET	240	UNDERGROUND ENGINEERING & ENVIRONMENTAL SOLUTIONS	SCALE: AS SHOWN





ar.

HALEY & HALDRICH ENGINEERGEUND ENGINEERGEUND ENGINEERGEUND ENGINEERGEUND	NOTES: 1. BASE PLA OF THE PLA FOR MASS TOPOGING ASSESSIO AND SHEE UPDATED 2. ELEVATIO	LEGEND:	
PHASE III - REMEDIAL ACTIO FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS SHALLOW GROUN ELEVATION CONTO DECEMBER 2000	N ADAPTED FROM 'TOPO ANUFACTURED GAS PLAN SACHUSETTS ELECTRIC O SACHUS, JUNE 1995, AND C PS PLAN SHEETS 1 AND C PS PLAN SHEETS 1 AND C TI NO. 53 BY FAY, SPOFFC JUNE 1976 AND REVISED JUNE 1976 AND REVISED	DESIGNATION AND APPF OF MONITORING WELL A ELEVATION USED IN SHA CONTOURING, DECEMBE GROUNDWATER ELEVAT DECEMBER 2000 APPROXIMATE GROUND EXISTING CONDITIONS FILE-SUPPORTED CONCI	
200 Feet DWATER OURS	GRAPHIC WORKSHEET T, MALDEN, MA* COMPANY BY EASTERN TITY OF MALDEN 2, JUNE 1995, 30 JULY 1979, 30 JULY 1979, ATIVE TO NGVD.	AD GROUNDWATER ND GROUNDWATER R 2000 TON CONTOURS WATER FLOW DIRECTION	



NALL AND AND AND AND AND AND A A A A

UNDERGROUND ENGINEERING & ENVIRONMENTAL	HALEY &	200	NOTES: 1. BASE PLA OF THE M FOR MAS TOPOGRA ASSESSO AND SHEE UPUA LEU UPUA LEU 2. ELEVATIC	LEGEND:	
SCALE AS SHOWN	PHASE III - REMEDIAL, ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS DEEP GROUNDWATER I AUGUST 2001	100 0 200	AN ADAPTED FROM "TOPOGRAPH AANUFACTURED GAS PLANT, MAL SACHUSETTS ELECTRIC COMPA APHICS, JUNE 1995, AND CITY OF DR'S PLAN SHEETS 1 AND 2, JUNE ET NO. 53 BY FAY, SPOFFORD &T DI'NE 1975 AND REVISED COULU JUNE 1975 AND REVISED COULU	DESIGNATION-AND-APPROXIMA OF DEEP MONITORING WELL AN HYDRAULIC HEAD, AUGUST 2001 DEEP GROUNDWATER HYDRAU CONTOURS, AUGUST 2001 APPROXIMATE FLOW DIRECTIO DEEP GROUNDWATER EXISTING CONDITIONS PILE-SUPPORTED CONCRETE C	
JUNE 2000	ELEVATIONS	0 Feet	IIC WORKSHEET DEN, MA' DEN, MA' NALDEN 1995, HORNDIKE, INC., Y 1970.	TE LOCATION D GROUNDWATER LIC HEAD	















ENVIRONMEN		HALE			2. ELEVATI	1. BASE PL OF THE MASSA FOR MASSA TOPOGRAF TOPOGRAF ASSESSOR ASSESSOR SHEET NO.	NOTES:	0	0	+ OW	₿502-OW	RW-1	e109A-OW	I	-4-					00	•		+	LEGEND.
	LOCATIONS OF PROPOSED REMEDIAL COMPONENTS AREA 2	CH PHASE III - REMEDIAL ACTION PLAN FORMER MALDEN MOP SITE MALDEN, MASSACHUSETTS	Feet	60 30 0 60	ONS ARE IN FEET AND RELATIVE TO NGVD.	AN ADAPTED FROM "TOPOGRAPHIC WORKSHEET NUFACTURED GAS PLANT, MALDEN, MA" ACHUSETTS ELECTRIC COMPANY BY EASTERN "SPLAN SHEETS 1 AND CITY OF MALDEN 53 BY FAY, SPOFFORD & THORNDIKE, INC., UNE 1976 AND REVISED 30-JULY 1979.		LIMITS OF AREA IN WHICH TSM IMPACTS HAVE BEEN OBSERVED	LIMITS OF AREA IN WHICH LNAPL IMPACTS HAVE BEEN OBSERVED	DESIGNATION AND APPROXIMATE LOCATION OF MONITORING WELL AT WHICH DEEP DNAPL HAS BEEN OBSERVED	DESIGNATION AND APPROXIMATE LOCATION OF MONITORING WELL AT WHICH SHALLOW DNAPL HAS BEEN OBSERVED	DESIGNATION AND APPROXIMATE LOCATION OF EXISTING EXTRACTION WELL	DESIGNATION AND APPROXIMATE LOCATION AT WHICH LNAPL HAS BEEN OBSERVED	APPROXIMATE LIMITS OF THE ORGANIC DEPOSIT	ELEVATION CONTOURS OF THE TOP OF THE ORGANIC DEPOSIT (2-FOOT INTERVAL)	SELECTED HISTORIC MGP FEATURES	APPROXIMATE LIMITS OF REMEDIAL ALTERNATIVE AREA 2	GROUT INJECTED INTO CRUSHED STONE BASE BENEATH THE WEB CULVERT	APPROXIMATE LOCATION OF PROPOSED SHALLOW DNAPL MIGRATION BARRIER	APPROXIMATE EXTENT OF PROPOSED IN-SITU CHEMICAL OXIDATION REMEDIATION (APPROXIMATE SPACING SHOWN)	APPROXIMATE LOCATION OF PROPOSED LNAPL MULTI-PHASE EXTRACTION (MPE) WELL	APPROXIMATE LOCATION OF PROPOSED LNAPL EXTRACTION WELL	APPROXIMATE LOCATION OF PROPOSED DNAPL EXTRACTION WELL	

1 -



HALE	2.ELEVAT	1. BASE PL OF THE MA FOR MASSA TOPOGRAF ASSESSOR SHEET NO.	NOTES:	0		-4-			₿502-OW					-	LEGEND:	
PHASE IN - REMEDIAL ACTION PLAN FORMER MALDEN MGP SITE MALDEN, MASSACHUSETTS LOCATIONS OF PROPOSED REMEDIAL COMPONENTS AREA 3	ONS ARE IN FEET AND RELATIVE TO NGVD	AN ADAPTED FROM "TOPOGRAPHIC WORKSHEET WUFACTURED GAS PLANT, MALDEN, MA" ACHUSETTS ELECTRIC COMPANY BY EASTERN PHICS, JUNE 1005, AND CITY OF MALDEN YS PLAN SHEETS 1 AND 2, JUNE 1995, AND .53 BY FAY, SPOFFORD & THORNDIKE, INC., JUNE 1976 AND HEVISED 30 JULY 1979.		LIMITS OF AREA IN WHICH TSM IMPACTS HAVE BEEN OBSERVED	APPROXIMATE LIMITS OF THE ORGANIC DEPOSIT	ELEVATION CONTOURS OF THE TOP OF THE ORGANIC DEPOSIT (2-FOOT INTERVAL)	SELECTED HISTORIC MGP FEATURES	BUILDINGS LOCATED WITHIN THE LIMITS OF AREA 3	DESIGNATION AND APPROXIMATE LOCATION OF MONITORING WELL AT WHICH SHALLOW DNAPL HAS BEEN OBSERVED	APPROXIMATE LIMITS OF REMEDIAL ALTERNATIVE AREA 3	GROUT INJECTED INTO CRUSHED STONE BASE BENEATH THE WEB CULVERT	APPROXIMATE LOCATION OF PROPOSED SHALLOW DNAPL MIGRATION BARRIER	APPROXIMATE EXTENT OF PROPOSED IN-SITU CHEMICAL OXIDATION REMEDIATION (APPROXIMATE SPACING SHOWN)	APPROXIMATE LOCATION OF PROPOSED DNAPL EXTRACTION WELL		



UNDERGROUN ENGINEERING		HALEY			2. ELEVATI	NOTES: 1. BASE PL OF THE MA FOR MASS; TOPOGRAF ASSESSOR SHEET NO. UPDATED J	C	0	erosena O	ļ	-4-				•	۲	LEGEND:	
1. A 10		Ê		50	ONS A	ACHUS SPLACHUS SS PLA 53 BY	IMP	IMP/	MON	APP	SEL THE	THE	APP	APP IN-S	APP	APP		
SCALE AS SI	LOCATIC	PHASE III - R FORMER MA		25	RE IN FEE	APTED FRO CTURED G. SETTS ELE JUNE 1000 N SHEETS FAY, SPOT FAY, SPOT	ACTS HAVE	ACTS HAVE	IGNATION UTORING V ERVED	ROXIMATE	ECTED HIS VATION CC ORGANIC	LIMITS OF	ROXIMATE	ROXIMATE ROXIMATE	ROXIMATE TI-PHASE	PL EXTRAC		
	DNS OF PROPO NENTS AREA 4	IEMEDIAL ACTION PLAN LIDEN MGP SITE ISSACHUSETTS		0	T AND RELATIVE TO	DM "TOPOGRAPHIC AS PLANT, MALDEN CTRIC COMPANY B , AND CITY OF WAI FORD & THORNDIN FORD & THORNDIN EVISED 30 JULY 19	BEEN OBSERVED	A IN WHICH LNAPL	AND APPROXIMATI VELL AT WHICH LN	LIMITS OF THE OF	TORIC MGP FEATU MUTOURS OF THE T DEPOSIT (2-FOOT )	CATED WITHIN AREA 4	AREA 4	EXTENT OF PROP CAL OXIDATION RE SPACING SHOWN	EXTRACTION OF PRO	TION WELL		
	SED REMEDIAL		Feet	50	O NGVD.	WORKSHEET 4. MA" BY EASTERN DEN 5. AND KE, INC.			E LOCATION OF APL HAS BEEN	IGANIC DEPOSIT	IRES TOP OF INTERVAL)		AL	OSED EMEDIATION I)	) WELL	OPOSED		





## APPENDIX A

Copy of Transmittal Form BWSC-108 and Public Notification of Availability Letters

27 June 2003

Massachusetts Department of Environmental Protection Northeast Regional Office One Winter Street Boston, Massachusetts: 02108

Attention: Site Management Branch

Subject: Phase III – Remedial Action Plan Former Malden Manufactured Gas Plant (MGP) Site – Upland Portion Malder, Massachusetts RTN 3-0362 and Linked RTNs 3-3757, 3-11581, 3-12448, 3-13310, 3-13345, 3-13753, and 3-13754 Tier IB Permit Number 7378

1

4.

Dear Ladies and Gentlemen:

Massachusetts Electric Company (MEC) is pleased to submit this report entitled "Phase III – Remedial Action Plan, Former Manufactured Gas Plant (MGP) Site - Upland Portion, Malden, Massachusetts, RTN 3-0362, Tier IB Permit 7378," prepared by Haley & Aldrich, Inc. This report is designed to meet the Phase III requirements under the Massachusetts Contingency Plan (310 CMR 40.0000) for the upland portions of the former Malden MGP Site. The Malden River portion of the Site will be addressed separately, as discussed below.

As MEC discussed in a meeting with Massachusetts Department of Environmental Protection (MADEP) officials on April 3, 2003, MEC intends to address the sediments in the Malden River within the Site boundary as a separate operable unit. MGP-related impacts attributable to the former MGP Site were identified during the Phase II assessment in Malden River sediments from the Malden River culvert outfall to a point approximately 1,400 feet downstream (just north of the Medford Street bridge). The Mystic Valley Development Commission (MVDC), through the TeleCom City partnership, has formed a group of parties with interest in Malden River sediment remediation from the culvert outfall to the Amelia Earhart Dam. The Telecom City partnership formed due to the development of a telecommunications research and development park on 200 acres of land situated in Malden, Medford and Everett. This area is located along the Malden River downstream of the Site boundary. The MVDC has partnered with the Army Corps of Engineers to conduct a study of the nature and extent of sediment impacts and to identify potential remedial measures that may be undertaken in the area. It is also our understanding that MADEP is participating in this work. MEC is contributing technical and financial support to this project and intends to participate in this study, which has been designated the Malden River Ecosystem Restoration Study. MEC also intends to participate in discussions regarding the remediation and restoration of the Malden River. In light of these recent developments, remedial measures related to Malden River sediments associated with the MGP Site are best conducted in

> 55 Bearfoot Road Northborough, MA 01532-1555 508.421.7000

conjunction with the efforts along the larger portion of the River to ensure consistency and coordination. Representatives of MADEP endorsed this approach at our April 3 meeting and during conversations thereafter.

An original signed copy of transmittal form BWSC-108 is submitted, unbound, along with this report. A copy of the signed form and this cover letter are provided in Appendix A of this report along with copies of Notification of Availability letters to appropriate Malden municipal officials.

Please contact me at 508-421-7564 with any questions or comments regarding this matter.

Sincerely,

Michel Viteone Gow

Michele V. Leone Senior Environmental Engineer

cc: Gregg Hunt, DEP (without enclosure) Rick Standish, H&A File

D E P	Massachusetts Departme Bureau of Waste Site Clea COMPREHENSIVE RESP FORM & PHASE I COMPL Pursuant to 310 CMR 40.0484 (Subp	ONSE ACTION TRANSMITT ETION STATEMENT Part D) and 40.0800 (Subpart H)	AL Release Tracking Numbe
A. SITE LOCATION: Site Name: (optional)	Former Malder MGP Site		
Street: 100 Comm	ercial Street	Location Aid: Commeci	al & Charles Streets
CityTown: Malden		ZIP Code: 02148-551	0
Felated Release Trackir	ng Numbers that this Form Addresses: 3-	-3757.3-11581.3-12448.3-13310.	.3-13345.3-13753.3-13754
Tier Classification: (che	ack one of the following)	Tier IB Tier IC	Tier II Not Tier Classified
If a Tier I Permit ha	is been issued, state the Permit Number.	ermit No. 7378, effective c	iate 12/28/1999
. THIS FORM IS BI	EING USED TO: (check all that apply)		
Submit a Phase I	Completion Statement, pursuant to 310 CMR	40.0484 (complete Sections A, B, C, G, H, I a	nd J).
Submit a Phase II	Scope of Work, pursuant to 310 CMR 40.0834	4 (complete Sections A, B, C, G, H, I and J).	
Submit a final Pha (complete Sections	se II Comprehensive Site Report and Comp s A, B, C, D, G, H, I and J).	eletion Statement, pursuant to 310 CMR 40.0	0836
Submit a Phase III	Remedial Action Plan and Completion Stat	tement, pursuant to 310 CMR 40.0862 (comp	lete Sections A, B, C, G, H, I and J).
Submit a Phase N	Remedy Implementation Plan, pursuant to 3	310 CMR 40.0874 (complete Sections A, B, C,	, G, H, I and J).
Submit an As-Buil	t Construction Report, pursuant to 310 CMR	40.0875 (complete Sections A, B, C, G, H, I a	nd J).
Submit a Phase IV (complete Sections	Final Inspection Report and Completion S & A, B, C, E, G, H, I and J).	tatement, pursuant to 310 CMR 40.0878 and	40.0879
Submit a periodic I	Phase V Inspection & Monitoring Report, pu	ursuant to 310 CMR 40.0892 (complete Sectio	ons A, B, C, G, H, I and J).
Submit a final Pha	se V Inspection & Monitoring Report and Co	ompletion Statement, pursuant to 310 CMR	40.0893
(complete Sections Yo	; A, B, C, F, G, H, Fand J). u must attach all supporting documentation any Legal Notices and Notices to	n required for each use of form indicated, o Public Officials required by 310 CMR 40.	including copies of .1400.
. RESPONSE ACTI	ONS:		
Check here if any r interested in using	esponse action(s) that serves as the basis for th this information to create an Innovative Technol	ne Phase submittal(s) involves the use of Inno ogies Clearinghouse.)	vative Technologies. (DEP is
Describe Technolo	gles:		1. 1. 0. 000
PHASE II COMPL	ETION STATEMENT:	WATER - INTER - INTER - INTER- IN	
Specify the outcome of t	he Phase II Comprehensive Site Assessment:		
Additional Comprei	hensive Response Actions are necessary at thi	s Site, based on the results of the Phase II Co	omprehensive Site Assessment.
The requirements of be submitted to DE	of a Class A Response Action Outcome have be P.	een met and a completed Response Action C	Dutcome Statement (BWSC-104) will
The requirements of be submitted to DE	of a Class B Response Action Outcome have be P.	een met and a completed Response Action O	outcome Statement (BWSC-104) will
Rescoring of this S	ite using the Numerical Ranking System is neo	essary, based on the results of the final Phase	e II Report.
. PHASE IV COMPI	LETION STATEMENT:		
Specify the outcome of F	Phase IV activities:		54.6
Phase V operation, (This site will be su	, maintenance or moni oring of the Comprehens bject to a Phase V Operation, Maintenance and	sive Response Action is necessary to achieve d Monitoring Annual Compliance Fee.)	e a Response Action Outcome.
The requirements of ensure the integrity	of a Class A Response Action Outcome have be of the Response Action Outcome. A complete	een met. No additional operation, maintenan Id Response Action Outcome Statement (BW	ce or monitoring is necessary to VSC-104) will be submitted to DEP.
The requirements of	of a Class C Response Action Outcome have be	een met. No additional operation, maintenan	ice or monitoring is necessary to

.

0	Massachusetts Bureau of Wast	Department of e Site Cleanup	Environm	ental Protection	BWSC-1
dul	COMPREHENS	WE BESDONS	E ACTION	COP	Pelease Tracking Num
	FORM & PHAS	E I COMPLETIC	ON STATE!	MENT	
U E P	Pursuant to 310 CMR	40.0484 (Subpart D)	and 40.0800 (Sr	ubpart H)	3 - 302
E. PHASE IV COMP	LETION STATEMENT:	(continued)		5	
The requirements of necessary to ensure Outcome Statement	of a Class C Response Action re that conditions are in aintain of (BWSC-104) will be submit	Outcome have been met ed and that further progre	. Further operatio	n, maintenance or monitorir d a Permanent Solution. A d	ng of the remedial action is completed Response Action
Indicate whether th	e operation and maintenance	will be Active or Passive.	(Active Operatio	n and Maintenance is define	ed at 310 CMR 40.0006.):
Active Opera	tion and Maintenance		O Passive	Operation and Maintenance	i
(Active Operation :	and Maintenance makes the S	ite subject to a Post-RAC	Class C Active C	Diversion and Maintenance	Annual Compliance Fee.)
Metive Operation	ETION STATEMENT:				
F. PHASE V COMPL	Enon Statement.				
The requirements	of a Class A Response Action	Outcome have been met	and a completed	Response Action Outcome	Statement (BWSC-104) will
resolution to be		~			
the integrity of the F	It a Class C Response Action Response Action Outcome. A	completed Response Ac	tion Outcome Stat	tement (BWSC-104) will be	submitted to DEP.
The requirements of necessary to ensure Outcome Statement	If a Class C Response Action e that conditions are maintainent (BWSC-104) will be submitt	Dutcome have been met. ed and that further progre ed to DEP.	. Further operatio iss is made toward	n, maintenance or monitorin d a Permanent Solution. A	ng of the remedial action is completed Response Action
indicate whether th	e operation and maintenance	will be Active or Passive.	(Active Operation	n and Maintenance is define	ed at 310 CMR 40.0006.):
Active Operation	tion and Maintenance		O Passive	Operation and Maintenance	
(Active Operation a	ind Maintenance makes the S	ite subject to a Post-RAC	Class C Active C	Operation and Maintenance	Annual Compliance Fee.)
<ul> <li>If Section B indicates that is (are) the subject of</li> </ul>	that a Phase I, Phase II, Pha f this submittal (I) has (have) b	se III, Phase IV or Phase en developed and imple	se V Completion	Statement is being subm lance with the applicable pr	itted, the response action(s) ovisions of M.G.L. c. 21E and
210 CMR 40.0000, (ii) is M.G.L. c. 21E and 310 C	(are) appropriate and reasona MR 40.0000, and (iii) complie	able to accomplish the pu s(y) with the identified pr	urposes of such re ovisions of all orde	esponse action(s) as set for ers, permits, and approvals	th in the applicable provisions of identified in this submittal;
<ul> <li>if Section B indicates (are) the subject of this se (are) appropriate and rea CMR 40.0000, and (iii) or</li> </ul>	that a Phase II Scope of Wo ubmittal (i) has (have) t een de isonable to accomplish the pu omplies(y) with the identified p	rk or a Phase IV Remed veloped in accordance w rposes of such response wovisions of all orders, p	ly Implementation with the applicable action(s) as set file ermits, and appro-	on Plan is being submitted, provisions of M.G.L. c. 216 orth in the applicable provis vals identified in this submit	, the response action(s) that is E and 310 CMR 40.0000, (ii) is ions of M.G.L. c. 21E and 310 tal;
If Section B indicates action(s) that is (are) the <0.0000, (ii) is (are) appl 21E and 310 CMR 40.00	that an As-Built Constructio subject of this submittal (i) is ( opriate and reasonable to acc 100, and (iii) complies(y) with t	n Report or a Phase V I are) being implemented complish the purposes of he identified provisions of	Inspection and M in accordance wit I such response a of all orders, perm	fonitoring Report is being h the applicable provisions ction(s) as set forth in the approvals identified its, and approvals identified	submitted, the response of M.G.L. c. 21E and 310 CMR pplicable provisions of M.G.L. c in this submittal.
I am aware that significar	t penalties may result, includir fally incomplete.	ig, but not limited to, pos-	sible fines and im	prisonment, if I submit infor	mation which I know to be
false, inaccurate or mater		is chining is based it an	y, are (were) subje	provisions	nd/or approval(s) issued by
false, inaccurate or mater Check here if the Re DEP or EPA. If the	esponse Action(s) on which th box is checked, you MIJST att	ach a statement identifyi	ng ine applicable		0.0
false, inaccurate or mate Check here if the Ri DEP or EPA. If the LSP Name: <u>Richarc</u>	esponse Action(s) on which th box is checked, you MIJST at <u>P. Standish</u>	LSP #: 2242	_ Stamp:	3 State	THE
false, inaccurate or mate Check here if the Ri DEP or EPA. If the LSP Name: <u>Richarc</u> Telephone: <u>860-282</u>	esponse Action(s) on which th box is checked, you MIJST att <u>P. St.andish</u>	_ LSP #: <u>2242</u> Ext.: <u>3150</u>	_ Stamp:	RICHARI	
false, inaccurate or mate Check here if the Ri DEP or EPA. If the LSP Name: <u>Richarc</u> Telephone: <u>860-282</u> FAX: (optional) <u>860-2</u>	esponse Action(s) on which th box is checked, you MUJST att <u>1 P. Standish</u>	LSP #: <u>2242</u> Ext.: <u>3150</u>	_ Stamp:	RICHARI STANDIS No. 2242	
f2lse, inaccurate or mate Check here if the Ri- DEP or EPA. If the LSP Name: <u>Richarr</u> Telephone: <u>R60-282</u> FAX: (optional) <u>R60-2</u> Signature: <u>LSP</u>	esponse Action(s) on which th box is checked, you MIJST at 1 P. Standish 2-9400 282-950 282-950 282-950 282-950 282-950 282-950 282-950 282-950 282-950 282-950 282-950 282-950	LSP #: 2242	_ Stamp: -	RICHARI STANDIS No. 2242 STELES	
false, inaccurate or mate Check here if the Ri DEP or EPA. If the LSP Name: <u>Richard</u> Telephone: <u>R60-282</u> FAX: (optional) <u>660-</u> Signature: <u>Richard</u> Signature: <u>Richard</u>	esponse Action(s) on which the box is checked, you MIJST at 1 P. Standish 2-9400 282-950 202-9500 202-9500 202-950 202-950 200	LSP #: <u>2242</u> Ext.: <u>3150</u>	. Stamp: - -	RICHARI STANDIS No. 2242	

ē2

Massachusetts De Bureau of Waste Si	partment of E te Cleanup	nvironn	nental	Pro		BWSC-108
DEP COMPREHENSIVE FORM & PHASE I O Pursuant to 310 CMR 40.0	RESPONSE A COMPLETION 0484 (Subpart D) an	STATE	TRAN MENT		TTAL	Release Tracking Number
H. PERSON UNDERTAKING RESPONSE ACTIO	DN(S):				<u>a n naora</u>	
Name of Organization: Massachusetts Elect.	ric Company					
Name of Contact: Michele_V. Leons	14120 2	Title: _S	Senior	_Env	ironmer	tal_Engineer
Street: <u>55_Bearfoot_Road</u>	7.	-				
City, Town: Northborough		State: 1	MA	Z	IP Code: _0	1532-0000
Telephone: 508-421-7564	Ext.:	_ FAX: (or	ptional) _	508-8	90-4706	5
Check here if there has been a change ir the person	undertaking the Respo	mse Action.				
I. RELATIONSHIP TO SITE OF PERSON UNDER	TAKING RESPON	SE ACTIO	N(S):	(che	eck one)	
RP or PRP Specify: O Owner O Operator	⊖ Generator ⊖	Transporte	r Other	RP or F	RP: Par	ty of Interest
Fiduciary, Secured Lender or Municipality with Exem	pt Status (as defined b	y M.G.L. c. 2	21E, s. 2)			
Agency or Public Utility on a Right of Way (as define	d by M.G.L. c. 21E, s.	5())				
Any Other Person Undertaking Response Action	pecify Relationship:	0.520	0.0000			
J. CERTIFICATION OF PERSON UNDERTAKING	RESPONSE ACT	ION(S):	1211		<u>, 12</u>	
Michele V. Leone	attest under the pain	s and penalt	ies of per	iury (i) t	hat I have p	ersonally examined and am
of those individuals immediately responsible for obtaining to knowledge and belief, true, accurate and complete, and (iii this submittal. Uthe person or entity on whose behalf this s possible fines and imprisonment, for willfully submitting fail	he information, the mai ) that I am fully authoriz submittal is made am/is ise, inaccurate, or incor	terial informa zed to make aware that t mplete inform	ition cont this attes there are nation.	ained in tation ö signific	this submit n behalf of t ant penalties	tal is, to the best of my he entity legally responsible for s, including, but not limited to,
By MORIONE		Title: _£	Senior	Env	ironmer	tal Engineer
(signature) For: Massachusetts Electric (Company		Date:	06	25/	63	
(print name of person or entity recorded in Section H)						
Enter address of the person providing certification, if different	ent from address recor	ded in Section	on H:			
Street:						
City/Town:		State:		2	IP Code: _	
Telephone:	Ext.:	FAX: (or	otional) _	-		
YOU MUST COMPLETE ALL RELEVANT INCOMPLETE. IF YOU SUBMIT AN	SECTIONS OF TH NINCOMPLETE FO A REQUIRED D	IS FORM ( DRM, YOU EADLINE.	DR DEP MAY B	MAY E PEN	RETURN	THE DOCUMENT AS
Revised 3/30/95 Superset	des Forms BWSC-	010 (in par	t) and 0	13	-	Page 3 of 3

UNDERGROUND ENGINEERING & ENVIRONMENTAL SOLUTIONS

Haley & Aldrich, Inc. 465 Medford Street Suite 2200 Boston, MA 02129-1400 Tei: 617.886.7400 Fax: 617.886.7600 www.HaleyAldrich.com



27 June 2003 File No. 06558

City of Malden Office of the Mayor 200 Pleasant Street Malden, Massachusetts 02148

Attention:

on: Mayor Richard C. Howard

Subject:

Public Notification of Availability Under 310 CMR 40.1403
 Phase III - Remedial Action Plan
 Former Malden Manufactured Gas Plant (MGP) Site - Upland Portion
 Malden, Massachusetts
 RTN 3-0362
 Tier IB Permit No. 7378

Ladies and Gentlemen:

On behalf of the Massachusetts Electric Company (MEC) and in accordance with the Massachusetts Contingency Plan (MCP), 310 CMR 40.0000, Haley & Aldrich, Inc., is notifying you of the availability of the above document.

This letter serves as notice of availability to the Malden Chief Municipal Officer in accordance with the MCP under 310 CMR 40.1403(3)(e), of a report entitled "Report on Phase III – Remedial Action Plan, Former Malden MGP Site – Upland Portion, Malden, Massachusetts, RTN 3-0362, Tier IB Permit No. 7378," dated June 2003, and prepared by Haley & Alcrich, Inc.

A copy of this Phase III - Remedial Action Plan report is available for review at the Massachusetts Department of Environmental Protection, Northeast Regional Office.

If you have questions concerning this letter, please contact Ms. Michele V. Leone of MEC at 508-421-7564, or the undersigned at 860-290-3150.

Sincerely yours, HALEY & ALDRICH, INC.

C:

W.K.

Richard P. Standish, LSP-of-Record Vice President

1

Massachusetts Electric Company; Attn: Ms. Michele V. Leone KeySpan Energy Delivery New England; Attn: Mr. Alexander G. Taft City of Malden Board of Health; Attn: Mr. Walter F. Carlan

G:\06558\634 Phase III\Final Phase\_III\_Report\Phase III public notification mayor.doc

### OFFICES

Charles Towr. West Virginia

Cleveland Ohic

Denver Colorade

Detroit Michigan

Hartford Connecticut

Los Angeles California

Manchester New Hampshire

Newark New Jersey

Portland Maine

Rochester New York

Ser, Diegr California

Sar Francisco California

Weshington District of Cohimbia

UNDERGROUND ENGINEERING & ENVIRONMENTAL SOLUTIONS

Haley & Aldrich, Inc. 465 Medford Street Suite 2200 Boston, MA 02129-1400 Tel: 617.886.7400 Fax: 617.886.7600 www.Haley.Aldrich.com



27 June 2003 File No. 06558

City of Malden Board of Health 200 Pleasant Street Malden, Massachusetts 02148

Attention: Mr. Walter F. Carlan Director of Public Health

Subject:

ect: Public Notification of Availability Under 310 CMR 40.1403 Phase III - Remedial Action Plan Former Malden Manufactured Gas Plant (MGP) Site – Upland Portion Malden, Massachusetts RTN 3-0362 Tier IB Permit No. 7378

Ladies and Gentlemen:

On behalf of the Massachusetts Electric Company (MEC) and in accordance with the Massachusetts Contingency Plan (MCP), 310 CMR 40.0000, Haley & Aldrich, Inc., is notifying you of the availability of the above document.

This letter serves as notice of availability to the Malden Board of Health in accordance with the MCP under 310 CMR 40.1403(3)(e), of a report entitled "Report on Phase III - Remedial Action Plan, Former Malden MGP Site - Upland Portion, Malden, Massachusetts, RTN 3-0362, Tier IB Permit No. 7378," dated June 2003, and prepared by Haley & Aldrich, Inc.

A copy of this Phase III - Remedial Action Plan report is available for review at the Massachusetts Department of Environmental Protection, Northeast Regional Office.

If you have questions concerning this letter, please contact Ms. Michele V. Leone of MEC at 508-421-7564, or the undersigned at 860-290-3150.

Sincerely yours, HALEY & ALDRICH, INC.

C:

Richard P. Standish, LSP-of-Record Vice President

> Massachusetts Electric Company; Attn: Ms. Michele V. Leone KeySpan Energy Delivery New England; Attn: Mr. Alexander G. Taft City of Malden, Chief Municipal Officer; Attn: Mayor Richard C. Howard

G:\06558\634 Phase III\Final\_Phase\_III\_Report\Phase III public notification BOH.doc

#### OFFICES

Charles Towr. West Virginia

Cieveland Omc

Denver Colorado

Detroit Michigan

Hartford Connecticut

Los Angeles California

Manchester New Hampshire

Newark New lersev

Portland Meine

Rochester New York

San Diege California

San Francisco California

Weshington Durinct of Columbia

# APPENDIX B

1

1

### **Revised AMEC Risk Characterization**

.


Amendment to Method 3 Risk Characterization and Substantial Hazard Evaluation Portion of Former Manufactured Gas Plant Site Malden, Massachusetts

1

Submitted to:

Massachusetts Electric Company Northborough, Massachusetts

Submitted by:

AMEC Earth & Environmental Boston, Massachusetts

February, 2003

6-7037-0500

Massachusetts Electric Company Substantial Hazard Evaluation Malden, Massachusetts February, 2003

### amec<sup>®</sup>

t

### **1.0 INTRODUCTION**

A Method 3 Human Health Risk Characterization was performed in December 2001 as part of a Phase II Comprehensive Site Investigation (Haley and Aldrich, 2001) for portions of the Former Malden Manufactured Gas Plant (MGP) Site (referred to herein as "The Site"), including parcels at 100 Commercial Street, 129 Commercial Street, 99-109 Commercial Street, 89 Commercial Street, 77 Commercial Street, 65 Commercial Street, 51 Commercial Street, Charles Street and the Governor House located at the corner of Pearl and Charles Street. These parcels are collectively referred to as the Upper Site; the portion of the Site consisting of the Malden River from the Malden River culvert outfall to the Medford Street Bridge is not addressed in this evaluation.

The conclusions of that assessment indicated that a condition of No Significant Risk to human health under <u>current</u> Site conditions exists at 100 Commercial Street, 99-109 Commercial Street, 89 Commercial Street, 77 Commercial Street, 65 Commercial Street, Charles Street, and Governor House. At the remaining two parcels that comprise the Upper Site, the Phase II Risk Characterization concluded that a condition of No Significant Risk to human health did not exist under current conditions for the following identified current receptors and exposure pathways:

- At the 129 Commercial Street parcel: the excess risk was associated with potential exposures to a current commercial/industrial worker who is assumed to inhale contaminants of potential concern (CPCs) in indoor air. Estimated excess cancer risk estimates for the current commercial/industrial worker exceeded the risk management criterion based on an average concentration of benzene. However, the average concentration of benzene (16.5 ug/m<sup>3</sup>) in indoor air samples (representing December 2000, March 2001, June 2001, and October 2001) was less than MADEP's published indoor air background concentration (21 ug/m<sup>3</sup>).
- At the 51 Commercial Street parcel: the excess risk was associated with a current landscape worker who is assumed to have direct exposure to soils. No soil data from the limited landscaped area on this parcel were available when the Risk Characterization was conducted. Rather, a single surrogate soil sample from an adjacent paved area (representing a depth interval 1 to 3 feet below ground surface [fbgs]) was used to estimate potential risks.

Four indoor air sampling events have been conducted at 129 Commercial Street following submission of the Method 3 Human Health Risk Characterization in December, 2001. These occurred on January 15, April 12, June 26, and October 14, 2002 (Haley and Aldrich, 2002; 2003). Additionally, three soil borings were collected and analyzed from 51 Commercial Street in June, 2002 (Haley and Aldrich, 2002). These data were used herein to update the results of the human health risk characterization for these two parcels.

A Substantial Hazard Evaluation was also conducted to determine whether a temporary solution has been achieved at portions of the Site in support of a Class C Response Action Outcome (RAO) in accordance with the Massachusetts Contingency Plan (MCP) (310 CMR 40.0956, 310

- -

Page 1

Massachusetts Electric Company Substantial Hazard Evaluation Malden, Massachusetts February, 2003

amec®

who may be exposed to volatile compounds in indoor air, the results of this Risk Characterization utilizing recent indoor air data indicate that the Cumulative Receptor Cancer Risk and Non-cancer Risk are less than the MADEP risk management criteria. The risk estimates for the utility worker, the landscape worker, and the trespasser, which represent conclusions of the Method 3 Risk Characterization (AMEC 2001) at 129 Commercial Street, are also less than the MADEP risk management criteria. Therefore, a condition of No Significant Risk to Human Health exists at 129 Commercial Street.

### 2.2 51 Commercial Street

The parcel located at 51 Commercial Street is currently used as an office building (dentist office and chiropractor office). A single building composed of multiple offices currently exists on the property, with landscaped areas surrounding the sides abutting Commercial and Centre Streets. Under current conditions, office workers, utility workers, trespassers, and landscapers are assumed to exist at this parcel. The complete exposure pathways for these receptors include inhalation of indoor air, incidental ingestion of soil, dermal contact with soil, incidental ingestion of groundwater, dermal contact with groundwater, inhalation of constituents in ambient air (volatilized from scil), inhalation of constituents in ambient air (volatilized from groundwater), and inhalation of particulates from soil. The conclusion of the Phase II Risk Characterization for current Site conditions was that a condition of No Significant Risk exists for the current office worker, utility worker, and the current trespasser.

For current landscapers, the basis of the exposure point concentration in the landscaped area adjacent to the building at 51 Commercial Street (soil exposure point) in the Phase II Risk Characterization was surrogate soil analytical data. The soil analytical data consisted of a single surface (0 to 3 fbgs) soil sample obtained from beneath pavement located near the boundary with the parcel at 65 Commercial Street. However, this is not soil to which landscapers could actually be exposed. Therefore, in June 2002, three soil borings were installed in the landscaped area with soil samples collected from the 0.5 to 1 foot interval and 0 to 3 feet interval (Haley and Aldrich 2002; see Attachment B for locations). Soil samples collected from the 0.5 to 1 foot interval were analyzed for Volatile Petroleum Hydrocarbon fractions (VPH) and target analytes; soil samples from the 0 to 3 feet interval were analyzed for Extractable Petroleum Hydrocarbon fractions (EPH) and target analytes. A summary of the data is presented in Table 4.

The entire landscaped area is considered a single exposure point, with the three soil samples (0 to 3 fbgs) considered representative of the exposure point. The results for naphthalene from the VPH analysis representing the 0.5 to 1 foot depth interval is used to represent the 0 to 3 feet depth interval because the detected concentrations were higher relative to that from the EPH analysis. For each detected compound, the soil exposure point concentration is based on the arithmetic average concentration of the detected concentrations and values equal to one-half the laboratory reporting limit. Field duplicate samples and primary samples were averaged prior to incorporation into the exposure point concentration. The soil exposure point concentrations are presented in Table 5.

Massachusetts Electric Company Substantial Hazard Evaluation Malden, Massachusetts February, 2003

amer

### 3.0 HUMAN HEALTH SUBSTANTIAL HAZARD EVALUATION

Despite the fact that no significant risk was found to receptors at either 129 Commercial Street or 51 Commercial Street, a Substantial Hazard Evaluation was conducted for these two parcels. Each is described in the following sections.

### 3.1 129 Commercial Street

 $(\mathbf{r})$ 

As described in Section 2.1, the parcel located at 129 Commercial Street is currently used as a commercial bread bakery. Under current conditions, utility workers, trespassers, commercial (site) workers, and landscapers are assumed to exist at this parcel. The complete exposure pathways for these receptors include incidental ingestion of soil, dermal contact with soil, incidental ingestion of groundwater, dermal contact with groundwater, inhalation of constituents in ambient air (volatilized from soil), inhalation of constituents in ambient air (volatilized from soil), inhalation of particulates from soil. The conclusion of the Method 3 Risk Characterization for current Site conditions was that a condition of No Significant Risk exists for the current utility worker, the current landscape worker, and the current trespasser. Therefore, a condition of No Substantial Hazard to Human Health exists for these receptors at 129 Commercial Street. As such, only commercial workers are included in this Substantial Hazard Evaluation.

Commercial workers are assumed to be present at 129 Commercial Street for 8-hour shifts 24 hours per day. The identified exposure pathway for current workers to identified OHM at this Site is via inhalation of indoor air. For the current commercial worker, exposures to OHM in indoor air are evaluated in this Substantial Hazard Evaluation based on indoor air data representing current Site conditions (collected on December 12, 2000, March 16, 2001, June 29, 2001, October 17, 2001, January 15, 2002, April 12, 2002, June 26, 2002, and October 14, 2002 [Haley and Aldrich, 2002]). Exposure point concentrations for indoor air are shown in Table 1 and were derived as described in Section 2.1. This represents the only current potential exposure pathway for this receptor.

The ADD for this receptor was calculated using the equations presented in Section 2.1 above. With the exception of the exposure duration and the averaging period for non-cancer risk estimates, the algorithm represents the same equation and assumptions utilized in the Method 3 Risk Characterization (AMEC, 2001) for current commercial workers at 129 Commercial Street. For this Substantial Hazard Evaluation, the period of exposure is representing January 1989 (the date of "site notification") to September 2007 (five years from the date of this evaluation) (310 CMR 40.0956(1)(b)). The toxicity values (reference dose [RfD] and Cancer Slope Factor [CSF]) and relative absorption values used in this Substantial Hazard Evaluation (see Table 8) are the same values used in the Method 3 Risk Characterization (AMEC, 2001), and represent currently available information regarding potential toxicity.

In order to determine whether a condition of No Substantial Hazard exists at this parcel, a risk characterization was conducted. As stated in Section 1.0, a condition of No Substantial Hazard to Human Health exists if, for an appropriate Exposure Period, no Cumulative Receptor Cancer Risk and no Cumulative Receptor Non-cancer Risk is greater than the Cumulative Receptor Risk Limits, defined as one in one hundred thousand (10<sup>-5</sup>) for carcinogenic effects and 1 for non-cancer health effects (MADEP Risk Management Criteria) (310 CMR 40.0956; 310 CMR

---

Page 7

### TABLE 1 SUMMARY OF INDOOR AIR EXPOSURE POINT CONCENTRATIONS 129 COMMERCIAL STREET FORMER MALDEN MANUFACTURED GAS PLANT MALDEN, MASSACHUSETTS

٠

Constituent	No. of Detections	Total No. of Samples	Minimum Detection (mg/m <sup>3</sup> )	Maximum Detection (mg/m <sup>3</sup> )	Arithmetic Mean (mg/m <sup>3</sup> )
Benzene	31	46	0.0014	0.0480	0.0132
Ethylbenzene	43	46	0.0010	0.0110	0.0025
Naphthalene	18	46	0.0013	0.0423	0.0029
Styrene	36	46	0.0012	0.0258	0.0043
Toluene	12	46	0.0068	0.0988	0.0232
m-&p-xylenes	16	46	0.0021	0.0232	0.0053
o-xylenes	10	46	0.0010	0.0070	0.0023

Note:

(1) This table presents the summary of data collected during the eight most recent monitoring events

conducted on December, 2000; March 2001; June 2001; October 17, 2001, January 15, 2002, April 12, 2002, June 26, 2002, and October 14, 2002.

(2) Exposure Point concentration based on arithmetic mean concentration of detected concentrations and values equal to one-half reporting limit for non-detects.

Table1.xls

### Table 2

Potential Carcinogenic Risk Following Exposure via Inhalation of Indoor Air-129 Commercial Street Amendment to Method 3 Risk Characterization Former Malden MGP, Malden, Massachusetts

F.

Receptor:

Site Worker (Adult)

LADD (mg/kg-day) =

CA x AAF x IR x ET x EF x ED BW x AP

ELCR =

LADD (mg/kg-day) \* CSF ( 1/mg/kg-day)

Parameter	r (units)	Value
LADD =	Lifetime Average Daily Dose Due to Inhalation (mg/kg-day)	See Below
CA =	Compound Concentration in Air (mg/m <sup>3</sup> )	Chemical-Specific
AAF =	Absorption Adjustment Factor (unitless)	Chemical-Specific
IR =	Inhalation Rate (m <sup>3</sup> /hr)	1.2
ET=	Exposure Time (hr/day)	8
EF =	Exposure Frequency (days/year)	250
ED =	Exposure Duration (years)	27
BW =	Body Weight (kg)	7,0
AP =	Averaging Period (days) (75 years x 365 days/yr, cancer)	27375
CSF=	Cancer Slope Factor (1/mg/kg-day)	Chemical-Specific
ELCR=	Estimated Lifetime Cancer Risk	Calculated

Compound	Compound (mg/m <sup>3</sup> )	AAF (unitless)	ADD (mg/kg-day)	Inhalation CSF (1/mg/kg-day)	Inhalation (unitless)
Benzene	0.0132	1	4E-04	0.0273	1E-05
Ethylbenzene	0.0025	1 1	9E-05	Class D	NA
Naphthalene	0.0053	1 1	2E-04	Class D	NA
Styrene	0.0029	1 1	1E-04	NA	NA
Toluene	0.0023	1 1	8E-05	Class D	NA
m-&p-xylenes	0.0043	1	1E-04	NA	NA
o-xylenes	0.0232	1	8E-04	Class D	NA
ÉL	CR =				1.E-05

Table2.xlsTable2\_pg2

Former Malden Manufactured Gas Plant Malden, Massachusetts Summary of Soll Data 51 Commercial Street Table 4

of 3 e jo of 3 013 of 3 of 3 of 3 013 of 3 of 3 of 3 of 3 of 3 of 3 Frequency of Detect of 3 0-3 3 3 3 3 3 3 1 N N 3 e 3 3 ł N 3 3 N 3 3 3 N 3 m 3 3 of 3 e jo 0.5-1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 I I 1 I 1 ł ł I 1 ł ł 2 I 1 1 810.33 135.83 42.63 23.33 62.83 318.33 5.28 0.85 2.34 2.14 5.42 4.52 4.14 4.40 0.52 5.52 4.45 1.21 7.70 0.59 0.35 0.79 1.22 0.16 0.64 8.77 3.81 3.67 4.84 9.70 0-3 1.47 Concentration Average 0.5-1 1.215 0.16 1 1 ł 1 1 ł 1 1 1 ł 1 ł ł ł 1 1 ļ I 1 1 I I 1 I 1 ł ł ł 1 20 D mg/kg 6-3 SS-103 DUP 11.5 0.45 9.5 2.4 1.8 0.51 2.6 1.3 3.6 1.5 0.24 1.8 1.1 65 2.4 127 ÷ 330 1.7 2.1 -F 1.7 4 2 -4.4 ٦. 81 I 1 N mg/kg 0.5-1 0.86 0.17 ł I ł 1 1 ļ 1 I ł ł 1 I I 1 I i I 1 I ł i 1 1 I 1 ł 5 :> mg/kg 0-3 0.92 0.76 0.76 0.89 34.5 0.92 0.57 0.51 0.57 191 0.47 1.6 0.57 1.6 0.57 300 0.25 0.57 0.57 0.82 8.9 16 4.0 1.8 1.1 -0:71 62 1 1 I l Sample Id and Depth (fbgs) SS-103 0 mg/kg 0.5-1 0.89 0.18 1 1 i 1 1 1 ł ł 1 1 1 1 1 1 1 1 I i 1 I I i 1 ļ 1 1 1 l l mg/kg 5 1730 65.4 0.46 0.38 0.94 0.98 4.9 -4-1-2.6 9.5 4.9 1.9 6.5 9.5 2.4 4.2 4.6 151 6.3 0.8 340 6.2 1.3 SS-102 53 4 8.4 9 17 I I 1.7 12 mg/kg 0.5-1 1.9 0.3 1 1 ł 1 ł 1 1 1 1 1 I 1 I ł I. 1 1 1 1 ł 1 L 1 1 ł I ł I 1 mg/kg 0-3 0.85 542 102 1.8 7.6 6.6 6.8 155 0.87 --5.8 300 0.41 8.3 7.3 5.4 1.9 4.2 3.4 72 38 8.4 2.3 4.7 8.1 6 7 I 1 4 SS-101 mg/kg 0.5-1 0.18 h 1.3 1 I 1 1 1 I ۱ ł ł 1 I 1 ł 1 1 ł 1 ł I 1 1 I ł ł 1 I 1 1 Cyanide, Physiologically A ndeno(1-2:3-cd)pyrene Dibenzo(a,h)anthracene Detected Compound Benzo(b)fluoranthene Benzo(k)fluoranthene 1-Methylnaphthalene 2-Methylnaphthalene Benzo(g,h,i)perylene Benzo(a)anthracene C11-C22 Aromatics C19-C36 Aliphatics Naphthalene (VOC) C9-C18 Aliphatics Benzo(a)pyrene Benzo(e)pyrene Avenaprilityiene Cyanide, Total Phenanthrene luoranthene Naphthalene Anthracene o/m-Xylene Chromium Chrysene Cadmium fuorene: Perylene Mercury Arsenic Banum Pyrene ead

lotes:

Average Concentration represents arithmetic average of detects and one-half laboratory reporting limit for non-detects. Data represent soil samples collected in June 2002 (Haley and Aldrich, 2002)

"---" = Not Evaluated Fable4.xls

### Table 6a

Potential Hazard Quotient and Risk Following Exposure via Incidental Ingestion of Soil - 51 Commercial Street Amendment to Method 3 Risk Characterization Former Malden MGP, Malden, Massachusetts

۲

Scenario:	Current Cocupational Scenario (0 - 3 ft.)
Receptor:	Landscaper
Intake (mg/kg-day) =	CS X IF: X FI X RAFO X EF X ED X CF

Hazard Quotient (HQ) = Cancer Risk (ELCR) =

Intake (m;)/kg-day) / RfD (mg/kg-day) Intake (m;)/kg-day) \* CSF [1/(mg/kg-day)]

Parameter (	units)	Value
ADD =	Average Daily Dose Due to Ingestion (mg/kg-day)	See Below
CS =	Compound Concentration in Soil (mg/kg)	Chemical-Specifi
IR =	Soil Ingestion Rate (Ing/day)	50
Fla	Fraction of Soil Ingested from the Site (unitless)	1
RAFo =	Relative Absorption Factor (Oral-Soil) (unitless)	Chemical-Specifi
EF =	Exposure Frequency (days/year)	24
ED =	Exposure Duration () ears)	27
CF =	Conversion Factor (kg/mg)	1E-06
BW =	Body Weight (kg)	70
AT =	Averaging Time (days) (ED x 365 days/yr, noncance	9855
AT =	Averaging Time (days) (75 yr. x 365 days/yr, cancer	27375
RfD=	Reference Dose (mg/kg-day)	Chemical-Specifi
CSF=	Cancer Slope Factor [1/(mg/kg-day)]	Chemical-Specifi

			Noncancer H	lazard Quotient			Excess Lifetin	ne Cancer Risk	
Compound	Soil Concentration (0 - 3 fl.) (mg/kç)	Oral-Soil RAF (unitless)	Intake (Noncancer) (mg/kg-day)	Chronic Oral RfD (mg/kg-day)	Soil Ingestion HQ (unitless)	Oral-Soit RAF (unitless)	Intake (Cancer) (mg/kg-day)	Oral CSF	Soil Ingestion Risk (unitless)
1-Methylnachthalena	0.64	0.43	1 35.08	2 05 02	6 4E 07	NA			
2. Methylophthalene	0.85	0.43	1 75 09	2.00-02	0.4E-07	NA	NA	NA	NA
Acepaphthylene	23	0.43	4.75-08	2.0E-02	8.0E-07	NA	NA	NA	NA
Anthracene	21	0.43	4 35-08	305.01	1.92-07	NA	NA	Class D	NA
Arsenic	88	0.51	2 15-07	3.05.04	7.05.04	S IE OI	7.00	Class D	NA
Barium	63	1	3.05.06	7.0E-04	1.02-04	5.12-01	7.6E-08	1.5E+00	1.1E-0/
Benzo(a)anthracene	53	0.27	6.7E-08	A DE-02	4.20-05	0 75 of	NA IS OR	NA	NA
Benzo(a)ovrene	54	0.27	6 95-08	4.00-02	1.72-00	2.72-01	2.4E-08	7.3E-01	1.8E-08
Benzo(b)fluoranthene	45	0.27	5.7E-08	4.00-02	1.72-00	2.72-01	2.5E-08	7.3E+00	1.8E-07
Benzo(e)ovrene	3.8	0.43	7.75-08	3.05.02	1.46-00	2.7E-01	2.12-08	7.3E-01	1.5E-08
Benzo(a h i)perviene	41	0.27	5 35.08	4 0E 02	1.00-00	NA	INA	Class D	NA
Benzo(k)fluoranthene	44	0.27	5.65-08	4.02-02	1.32-00	2 75 04	NA NA	Class D	NA
C11-C22 Aromatics	810	0.43	1 65-05	3.05.02	1.4E-06	2.7E-01	2.0E-08	7.3E-02	1.55-09
C19-C36 Aliphatics	136	0.91	5 8E-06	205+00	3.56-04	NA	NA	NA	NA
C9-C18 Aliphatics	43	0.91	1.95-06	105.01	1.95-00	NA	NA	NA	NA
Cadmium	0.5	1	2 55.08	1.02-01	1.00-05	NA	NA	NA	NA
Chromium	23	03	2.35-00	1.02-03	2.36-05	NA	NA	NA	NA
Chrysene	55	0.27	7.05.08	3.0E-03	1.1E-04	NA	NA	NA	NA
Cyanide Physiologically Av	11	1	F.0E-00	4.0E-02	1.7E-06	2./E-01	2.5E-08	7.3E-03	1.8E-10
Cyanida Total	45	0.2	5.0E-00	2.0E-02	2.5E-06	NA	NA	Class D	NA
Dibenzo/a blanthracene	12	0.27	4.22-00	2.0E-02	2.1E-06	NA	NA	Class D	NA
Eluoranthene	77	0.27	1.5E-08	4.0E-02	3.8E-07	2.7E-01	5.5E-09	7.3E+00	4.0E-08
Elucrene	0.50	0.43	1.6E-07	4.0E-02	3.9E-06	NA	I NA	Class D	NA
Indexe(1.2.3. edlesmon	0.59	0.43	1.2E-08	4.0E-02	3.0E-07	NA	I NA	Class D	NA
Load	3.1	0.27	4.7E-08	4.0E-02	1.2E-06	2.7E-01	11.7E-08	7.3E-01	1.2E-08
Load	310	0.3	4.5E-06	7.5E-04	6.0E-03	NA	I NA	NA	NA
Nercury	0.35	2	3.2E-08	3.0E-04	1.1E-04	NA	NA	Class D	NA
(voc)	1.2	0.43	2.5E-08	2.0E-02	1.2E-06	NA	NA	NA	NA
p/m-Xylene	0.16	1	7.5E-09	2.0E+00	3.8E-09	NA	NA	Class D	NA
Pervene	1.5	0.43	3.0E-08	3.0E-02	9.9E-07	NA	NA	Class D	NA
Phenanthrene	4.8	0.43	9.8E-08	4.0E-02	2.4E-06	NA	NA	Class D	NA
ryrene	10	0.43	2.0E-07	3.0E-02	6.5E-06	NA	NA	Class D	NA
		<u>a' a</u>	12 2	Hazard Index:	0.008		То	tal Cancer Risk:	3.8.E-07

.

Notes:

4

Pyrene RAF, RfD, and CSF are used as a surrogate values for Benzo(e)pyrene and Perylene.

Table6\_xisTable6a

# AMENDMENT TO METHOD 3 RISK CHARACTERIZATION FORMER MALDEN MANUFACTURED GAS PLANT SUMMARY OF RISK ESTIMATES MALDEN, MASSACHUSETTS 51 COMMERCIAL STREET TABLE 7

Current Incidental Ingestion of Soil   Landscaper <sup>(1)</sup> Incidental Ingestion of Soil   Dermal Contact with Soil TOTAL RISK   Trespasser Incidental Ingestion of Soil   Trespasser Incidental Ingestion of Soil   Utility Scenario) Dermal Contact with Soil   Utility Worker Incidental Ingestion of Soil   Utility Worker Incidental Ingestion of Soil	of Soil 4E-07 81 Soli 3E-07 81 of Soil 7E-07 11 of Soil 1E-07 11 Soil 9E-08 31 Solust 7E-10 31	8E-03 3E-03 <b>1E-02</b> 3E-04 3E-05	°N N N N
Landscaper <sup>(1)</sup> Incidental Ingestion of Soil   Landscaper <sup>(1)</sup> Dermal Contact with Soil   ToTAL RISK Incidental Ingestion of Soil   Trespasser Incidental Ingestion of Soil   Utility Scenario) Dermal Contact with Soil   Inhalation of Fugitive Dust Inhalation of Vapors   TotAL RISK Incidental Ingestion of Soil   Utility Worker Incidental Ingestion of Soil   Utility Worker Incidental Ingestion of Soil	of Soil 4E-07 8 Soi 3E-07 3 50 Soi 7E-07 1 of Soi 1E-07 1 Soi 9E-08 3 E 0ust 7E-10 3 SE 00 3	8E-03 3E-03 <b>1E-02</b> 3E-05	No No No
Landscaper <sup>(1)</sup> Dermal Contact with Soli TOTAL RISK Incidental Ingestion of Soli Dermal Contact with Soli Inhalation of Fugitive Dust Inhalation of Vapors TOTAL RISK Incidental Ingestion of Soli Dermal Contact with Soli Dermal Contact with Soli Dermal Contact with Soli	Solit     3E-07     3       of Solit     1E-07     11       Solit     1E-07     11       Solit     1E-07     3       Solit     9E-08     3       Subst     7E-10     3	3E-03 1 <b>E-02</b> 1E-04 3E-05	oN No No
TOTAL RISK   Trespasser Incidental Ingestion of Soil   Trespasser Incidental Ingestion of Soil   Utility Scenario) Dermal Contact with Soil   Inhalation of Fugitive Dust Inhalation of Soil   Incidental Ingestion of Soil Inhalation of Fugitive Dust   Incidental Ingestion of Soil Incidental Ingestion of Soil   Utility Worker Incidental Ingestion of Soil	7E-07     1       of Soil     1E-07     1       i Soil     9E-08     3       e Dust     7E-10     3	<b>1E-02</b> 1E-04 3E-05	No
Trespasser     Incidental Ingestion of Soil       Trespasser     Dermal Contact with Soil       Utility Scenario)     Inhalation of Fugitive Dust       TOTAL RISK     Incidental Ingestion of Soil       Dermal Contact with Soil     Dermal Contact with Soil       Utility Worker     Inhalation of Fugitive Dust	of Soil 1E-07 1 Soil 9E-08 31 s Dust 7E-10 31	1E-04 3E-05	No
Trespasser (Utility Scenario) TOTAL RISK Inhalation of Fugitive Dust TOTAL RISK Incidental Ingestion of Soil Dermal Contact with Soil Inhalation of Fugitive Dust	Soil 9E-08 31 9 Dust 7E-10 31	3E-05	
Utility Scenario) TOTAL RISK TOTAL RISK Inhalation of Vapors TOTAL RISK Incidental Ingestion of Soil Dermal Contact with Soil Inhalation of Fugitive Dust	a Dust 7E-10 31		No
Utility scenaro) Inhalation of Vapors TOTAL RISK Incidental Ingestion of Soil Dermal Contact with Soil Utility Worker Inhalation of Fugitive Dust	CE DO	3E-06	No
TOTAL RISK Incidental Ingestion of Soil Dermal Contact with Soil Utility Worker Inhalation of Fugitive Dust	00-02	3E-03	No
Utility Worker Inhalation of Soil	2E-07 3I	3E-03	No
Utility Worker Inhalation of Fugitive Dust	of Soil 1E-06 41	4E-04	No
Utility Worker Inhalation of Fugitive Dust	Soil 8E-07 4I	4E-04	No
	3 Dust 4E-08 81	8E-05	No
Inhalation of Vapors	5E-08 71	7E-03	No
TOTAL RISK	2E-06 8I	8E-03	No
Site Worker Inhalation of Indoor Air	Air 2E-06 1	1E-01	No
TOTAL RISK	2E-06 11	1E-01	No

## Notes:

\* = MADEP establishes a condition of No Significant Risk if the hazard index is less than or equal to one and the estimated lifetime cancer risk is less than or equal to 1E-5.

(1) Represents revised risks using recent data. All other risk estimates from Phase II Risk Characterization (AMEC 2001).

Table7.xls

### Table 8 Potential Carcinogenic Risk Following Exposure via Inhalation of Indoor Air Substantial Hazard Evaluation Former Malden MGP, Nalden, Massachusetts

Receptor:

Site Worker (Adult)

LADD (mg/kg-day) =

CA x AAF x IR x ET x EF x ED BW x AP

ELCR =

LADD (mg/kg-day) \* CSF ( 1/mg/kg-day)

Paramete	r (units)	Value
LADD =	Lifetime Average Daily Dose Due to Inhalation (mg/kg-day)	See Below
CA =	Compound Concentration in Air (mg/m <sup>3</sup> )	Chemical-Specific
AAF =	Absorption Adjustment Factor (unitless)	Chemical-Specific
IR =	Inhalation Rate (m <sup>3</sup> /hr)	1.2
ET=	Exposure Time (hr/day)	8
EF =	Exposure Frequency (days/year)	250
ED =	Exposure Duration (years)	19
BW =	Body Weight (kg)	70
AP =	Averaging Period (days) (75 years x 365 days/yr, cancer)	27375
CSF=	Cancer Slope Factor (1/mg/kg-day)	Chemical-Specific
ELCR=	Estimated Lifetime Cancer Risk	Calculated

Compound	Compound (mg/m <sup>3</sup> )	AAF (unitless)	ADD (mg/kg-day)	Inhalation CSF (1/mg/kg-day)	Inhalation (unitless)	
Benzene	0.0132	1	3E-04	0.0273	9E-06	
Ethylbenzene	0.0025	1	6E-05	Class D	NA	
Naphthalene	0.0029	1	7E-05	NA	NA	
Styrene	0.0043	1	1E-04	NA	NA	
Toluene	0.0232	1	6E-04	Class D	NA	
m-&p-xylenes	0.0053	1	1E-04	Class D	NA	
o-xylenes	0.0023	1	6E-05	Class D	NA	
EL	CR #				9.E-06	

Table8.xlsTable8,pg2

### Table 10a

### Potential Exposure via Incidental Ingestion of Soll - 51 Commercial Street Substantial Hazard Evaluation

Former Malden MGP, Malden, Massachusetts

Scenario:	Current Occupational Scenario (0 - 3 ft.)
Receptor:	Landscaper

#### ADD/LADD (mg/kg-day) = CS x IR x FI x RAFo x EF x ED x CF BW x AT

Hazard Quotient (HQ) = ADD (mg/kg-day) / RfD (mg/kg-day) Cancer Risk (ELCR) = LADD (mg/kg-day) \* CSF [1/(mg/kg-day)]

Parameter (	units)	Value
ADD =	Average Daily Dose Dun to Ingestion (mg/kg-day)	See Below
LADD =	Lifetime Average Daily Dose (mg/kg-day)	See Below
CS =	Compound Concentration in Soil (mg/kg)	Chemical-Specific
IR =	Soil Ingestion Rate (mg/day)	50
FI=	Fraction of Soil Ingestec from the Site (unitless)	1
RAFo =	Relative Absorption Factor (Oral-Soil) (unitless)	Chemical-Specific
EF =	Exposure Frequency (days/year)	24
ED =	Exposure Duration (years)	19
CF =	Conversion Factor (kg/mg)	1E-06
BW =	Body Weight (kg)	70
AT =	Averaging Time (days) (ED x 365 days/yr, noncancer)	6935
AT =	Averaging Time (days) (75 yr. x 365 days/yr. cancer)	27375
RfD=	Reference Dose (mg/kg-day)	Chemical-Specific
CSF=	Cancer Slope Factor [1/img/kg-day)]	Chemical-Specific

			Noncancer	Hazard Quotient			Excess Lifeti	me Cancer Rist	
Compound	Soil Concentraticn (0 - 3 ft.) (mg/kg)	Oral-Soil RAF (unitless)	ADD (Noncancer) (mg/kg-day)	Chronic Oral RfD (mg/kg-day)	Soil Ingestion HQ (unitless)	Oral-Soil RAF (unitiess)	LADD (Cancer) (mg/kg-day)	Oral CSF (mg/kg-day) <sup>-1</sup>	Soil Ingestion Risk (unitless)
1-Methylnaphthalene	0.64	0.43	1.3E-08	0.02	6 4F-07	NAI	NA	NA	NA
2-Methylnaphthalene	0.85	0.43	1.7E-08	0.02	8.6E-07	NA	NA	NA	NA
Acanaphibylene	2 34	0.43	4 7E-08	0.06	7 95-07	NAL	NA	Class D	NA
Anthracene	2 14	0.43	4 3E-08	0.3	1 4E-07	NA	NA	Class D	NA
Arsenic	8.77	0.51	2 1E-07	0.0003	7 0E-04	0.51	5 3E-08	15	8 05-08
Barium	63	1	3 0E-06	0.07	4 2E-05	NA	NA.	NA	NA NA
Beozo(a)anthracene	5	0.27	6 7E-08	0.04	1 7E-06	0.27	175-08	0.73	1 25.08
Benzo(a)ovrene	5.42	0.27	6 9E-08	0.04	175-06	0.27	175-08	73	1 25.07
Benzo(b)fluoraothane	5	0.27	5 7E-08	0.04	145-06	0.27	1.55.08	0.73	1.32-07
Benzo(e)pyrepa	4	0.43	7 7E-08	0.03	2.65.06	NA	NA	Class D	1.12-00
Benzo(a h i)perviene	4	0.27	5 3E-08	0.04	1 35.06	NAL	NA	Class D	NA
Benzo(k)fuoranthene	4	0.27	5 6E-08	0.04	1 45.06	0.27	I AE OR	0.072	105.00
C11_C22 Aromatics	810 33	0.43	1.65-05	0.04	5.5E.04	0.27	1.46-06	0.073	1.0E-09
C10 C26 Aliabatics	426	0.45	E OF OF	0.03	0.00 00	NA	NA	NA	NA
C19-C30 Aliphatics	130	0.01	1.00-00		2.92-00	NA	NA	NA	NA
C9-C18 Alphaucs	43	0.91	1.82-00	0.1	1.85-05	NA	NA	NA	NA
Cadmium	22.02		2.52-08	0.001	2.5E-05	NA	NA	NA	NA
Chromium	23.33	0.3	3.3E-07	0.003	1.1E-04	NA	NA	NA	NA
Chrysene	6	0.27	7.0E-08	0.04	1.7E-06	0.27	1.8E-08	0.0073	1.3E-10
Cyanide, Physiologically Av	1	1	5.0E-08	0.02	2.5E-06	NA	NA	Class D	NA
Cyanide, Total	4	0.2	4.2E-08	0.02	2.1E-06	NA	NA	Class D	NA
Dibenzo(a,h)anthracene	1	0.27	1.5E-08	0.04	3.8E-07	0.27	3.9E-09	7.3	2.8E-08
Fluoranthene	7.70	0.43	1.6E-07	0.04	3.9E-06	NA	NA	Class D	NA
Fluorene	0.59	0.43	1.2E-08	0.04	3.0E-07	NA	NA	Class D	NA
Indeno(1,2,3-cd)pyrene	4	0.27	4.7E-08	0.04	1.2E-06	0.27	1.2E-08	0.73	8.6E-09
Lead	318	0.3	4.5E-06	0.00075	6.0E-03	NA	NA	NA	NA
Mercury	0	2	3.2E-08	0.0003	1.1E-04	NA	NA	Class D	NA
Naphthalene (VOC)	1	0.43	2.5E-08	0.02	1.2E-06	NA	NA	NA	NA
p/m-Xylene	0.16	1	7.5E-09	2	3.8E-09	NA	NA	Class D	NA
Perviene	1.47	0.43	3.0E-08	0.03	9.9E-07	NA	NA	Class D	NA
Phenanthrene	4.84	0.43	9.8E-08	0.04	2.4E-06	NA	NA	Class D	NA
Pyrene	10	0.43	2.0E-07	0.03	6.5E-06	NA	NA	Class D	NA
				Hazard Index	0.008		Tr	tal Cancer Risk	27E.07

~

4

Notes:

Toxicity information for Pyrene used as surrogate values for Benzo(e)pyrene and Perylene.

Table10.xlsTable10a

## TABLE 11 SUMMARY OF RISK ESTIMATES 51 COMMERCIAL STREET SUBSTANTIAL HAZARD EVALUATION FORMER MALDEN MANUFACTURED GAS PLANT MALDEN, MASSACHUSETTS

Receptor	Exposure Pathway	Estimated Lifetime Cancer Risk (ELCR)	Hazard Index (HI)	Exceeds MADEP Target Risk Leveis*
Current				
	Incidental Ingestion of Soil	3E-07	8E-03	No
Landscaper <sup>(1)</sup>	Dermal Contact with Soil	2E-07	3E-03	No
	TOTAL RISK	5E-07	1E-02	No
	Incidental Ingestion of Soil	1E-07	1E-04	No
	Dermal Contact with Soil	9E-08	3E-05	No
I respasser	Inhalation of Fugitive Dust	7E-10	3E-06	No
(Utility Scenario)	Inhalation of Vapors	6E-09	3E-03	No
	TOTAL RISK	2E-07	3E-03	No
	Incidental Ingestion of Soil	1E-06	4E-04	No
	Dermal Contact with Soil	8E-07	4E-04	No
Utility Worker	Inhalation of Fugitive Dust	4E-08	8E-05	No
	Inhalation of Vapors	5E-08	7E-03	No
	TOTAL RISK	2E-06	8E-03	No
Site Worker	Inhalation of Indoor Air	2E-06	1E-01	No
	TOTAL RISK	2E-06	1E-01	No

# Notes:

-

\* = MADEP establishes a condition of No Significant Risk if the hazard index is less than or equal to one and the estimated lifetime cancer risk is less than or equal to 1E-5.

Represents revised risks using recent data in Substantial Hazard Evaluation. All other risk estimates from Phase II Risk Characterization (AMEC 2001).



